

Audit Committee 14 July 2014

Time 2.00 pm Public Meeting? YES Type of meeting Regulatory

Venue Committee Room 3 - Civic Centre, St Peter's Square, Wolverhampton WV1 1SH

Membership

Chair Cllr Keith Inston (Lab)
Vice-chair Cllr Christine Mills (Con)

Labour Conservative Liberal Democrat

Cllr Harbans Bagri Cllr Philip Bateman Cllr Alan Bolshaw Cllr Dr Michael Hardacre Cllr Jasbir Jaspal Cllr Wendy Thompson

Quorum for this meeting is two Councillors.

Information for the Public

If you have any queries about this meeting, please contact the democratic support team:

Contact Dereck Francis

Tel: 01902 555835 or dereck.francis@wolverhampton.gov.uk

Address Democratic Support, Civic Centre, 2nd floor, St Peter's Square,

Wolverhampton WV1 1RL

Copies of other agendas and reports are available from:

Website http://wolverhampton.moderngov.co.uk/mgListCommittees.aspx?bcr=1

Email democratic.support@wolverhampton.gov.uk

Tel 01902 555043

Some items are discussed in private because of their confidential or commercial nature. These reports are not available to the public.

Agenda

Part 1 – items open to the press and public

Item No. Title

MEETING BUSINESS ITEMS

- 1 Apologies for absence
- 2 **Declaration of interests**
- 3 Minutes of previous meetings (Pages 1 14)
 - (a) Audit Committee 10 March 2014 [For approval]
 - (b) Audit (Monitoring of Audit Investigations) Sub-Committee 28 April 2014 [For information]
- 4 Matters arising

[To consider any matters arising from the minutes]

DECISION ITEMS

- 5 Internal Audit Report Performance Appraisal Scheme (Pages 15 28)
 [To receive the outcome of the internal audit review into the Council's performance appraisal scheme]
- Annual Review of the Effectiveness of Internal Audit (Pages 29 34)
 [To provide the Committee with sources of information and measures in place in order to assist it in being able to reach a conclusion on the adequacy and effectiveness of the Internal Audit Service]
- 7 Audit Committee Annual Report 2013/14 (Pages 35 38)
 [To endorse the annual report and refer it to Full Council for approval]
- 8 **Draft Statement of Accounts 2013/14** (Pages 39 256) [To note the draft statement of accounts]
- 9 **Annual Governance Statement 2013/14** (Pages 257 276) [To review and comment upon the Council's annual governance statement]
- 10 Corporate Risk Register (Pages 277 294)
 [To note and comment on the current list of corporate risks and the actions being taken to manage them]
- 11 **Annual Internal Audit Report 2013/14** (Pages 295 316) [To note the content of the internal audit report]

12 Chartered Institute of Public Finance and Accountancy (CIPFA) Audit Committee Update - Issue 13 (Pages 317 - 332)

[To note the contents of the latest CIPFA audit committee update – helping audit committees to be effective]

13 **Payment Transparency** (Pages 333 - 336)

[To note the Council's current position with regards to the publication of all payments over £500]





Audit Committee

Minutes - 10 March 2014

Attendance

Members of the Committee

Cllr Keith Inston (chair)
Cllr Harbans Bagri
Cllr Phil Bateman
Cllr Alan Bolshaw
Cllr Mike Hardacre
Cllr Jasbir Jaspal

Cllr Mrs Wendy Thompson

Independent members

Mike Ager Terry Day

Employees

Jane Alexander Client Lead Auditor Peter Farrow Head of Audit

Dereck Francis Democratic Support Officer Keith Ireland Strategic Director Delivery

Katy Morgan Client Lead Auditor
Richard Morgan Senior Audit Manager
Narinder Phagura Strategic Risk Manager

Martyn Sargeant Democratic Services Manager Mark Taylor Assistant Director Finance

Simon Warren Chief Executive

External Auditors

Richard Vialard PricewaterhouseCoopers
John Sawyer PricewaterhouseCoopers

Part 1 – items open to the press and public

Item Title No.

MEETING BUSINESS ITEMS

1. Apologies for absence

Apologies for absence were submitted on behalf of Cllr Matthew Holdcroft.

2. Declarations of interest

No interests were declared.

3. Minutes

(a) Audit Committee - 16 December 2013

Cllr Mrs Thompson reported that she had asked a number of questions to the external auditor from PricewaterhouseCoopers (PWC) on the Council's financial position, monitoring and reserves and these were not reflected in the minutes. She reported that she had no confidence in the accuracy of the minutes and had expressed her concerns to the Democratic Services Manager.

Resolved:

- 1. That the minutes of the meeting held on 16 December 2013 be approved as a correct record and signed by the Chair.
- 2. That the objections of Cllr Mrs Thompson to the accuracy of the minutes be noted.

(b) Audit (Monitoring of Audit Investigations) Sub-Committee – 9 December 2013

Resolved:

That the minutes of the meeting held on 9 December 2013 be noted.

(c) Audit (Monitoring of Audit Investigations) Sub-Committee – 20 January 2014

Resolved:

That the minutes of the meeting held on 20 January 2014 be noted.

4. Matters arising

There were no matters arising from the minutes.

5. Work programme 2013/14

The Committee received the updated work programme for the remainder of the municipal year 2013/14. Peter Farrow, Head of Audit reported that there was only one item of business, an external audit report on 'annual grants certification', scheduled for the meeting on 14 April. He advised that it made sense to cancel the April meeting and roll forward the external audit report to the meeting on 22 July.

Resolved:

- 1. That the updated Audit Committee work programme for 2013/14 be received and noted.
- 2. That the meeting scheduled for 14 April 2014 be cancelled.

EXTERNAL AUDIT REPORTS AND INSPECTION:

6. External Audit Plan 2013/14

Richard Vialard from the Council's external auditors PricewaterhouseCoopers' (PwC) presented their external audit plans for 2013/14. He advised that the audit would take place between July and August. In presenting the report he touched on the audit approach and the summary of the significant and elevated audit risks identified for 2013/14.

Referring to reserves, and specifically the movement of reserves from specific to general reserve. Cllr Mrs Thompson asked PwC how long that term should be. Against the background of the Audit Investigation reports referred to at items 15 and 16 on the agenda, she also asked whether PwC had confidence in the robustness of all the Council's budgets. PwC informed the Committee that the Council had the power to determine how it structured and used its reserves. The Council had a clear policy on the use of reserves which indicated that within two years reserves would have to be built back up to an acceptable level if they dipped too low. On the question of the robustness of the Council's budgets PwC explained that the responsibility of the external auditors was to look at the Council's financial statement once it had been audited and determine whether what was planned to be delivered had in fact been done. He added that it would not be right for him to comment on individual budgets, notwithstanding that the two audit investigation reports were important areas. He acknowledged that there may be lessons to be learned from the circumstances in the two reports but added that these were two specific issues. Peter Farrow added that the introduction of the Agresso system would provide real time budgetary information that would address a number of the issues referred to in the specific audit investigation reports.

Mike Ager asked whether the risks within the external audit plan were tailor-made for the Council. He was advised that the first two risks 'managing override of controls' and 'risk of fraud in revenue and expenditure recognition' were standard but the other four are specific to Wolverhampton.

Resolved:

That the contents of the external audit plan 2013/14 from the Council's external auditors, PricewaterhouseCoopers be noted.

RISK MANAGEMENT- ASSURANCE ON CORPORATE RISKS

7 Corporate Risk Register

The Committee received, for information, a report on the corporate risk register as at February 2014 which identified the key risks the Council faced in delivering its corporate themes of 'Encouraging new business'; 'Empowering people and communities';' Reinvigorating the city'; and 'Confident, capable Council'.

Additional risks were currently being considered by the Strategic Executive Board (SEB) and the Corporative Delivery Board, the outcomes of which would be reported to the Committee in due course. It was also reported that the Committee would receive information on the individual risk management action plans developed to address the risks and an accompanying corporate assurance map to identify how the risks were being managed and where the Committee could get the assurance on such activity.

Cllr Bateman reported that during the run-up to the consultations and approval of the Council's budget for 2014/15, the ability of the city to increase business activity and regeneration had featured heavily. He indicated that this issue was fundamental to the improvement of the city and he found it surprising that this issue did not feature in the risk register. The Chief Executive informed the Committee that it was addressed in risk reference 9. 'City centre regeneration'. Narinder

Phagura also advised the Committee that the report before the Committee was a high level risk register. Operational risk registers would capture issues like attracting businesses into the city centre and other regeneration programmes in other parts of the city. The SEB would be reviewing the register on an ongoing basis to ensure that it represented the risk profile of the Council. Simon Warren, Chief Executive added that the risk did not just encompass the city centre area but the description would be looked at to ensure it was clear.

Referring to risk reference 1 (Looked After Children), Cllr Bolshaw reported that this could be a sub-category of risk reference 7 (Safeguarding). He also suggested that improving safeguarding procedures might well lead to the number of looked after children increasing. Simon Warren reported that the two risk areas were associated but different. Risk reference 1 was about costs whilst risk 7 was about processes and procedures.

Regarding to risk references 2 (community skills) and 14 (school improvement), Cllr Mrs Thompson, commented that in terms of job creation and job opportunities, anyone with maths and literacy qualifications would find it easier to access job opportunities. She asked how many specialist maths colleges the city had and what was being done to extend good maths learning across all schools in the city. In response Simon Warren reported that the community skills was focussed on work readiness skills such as, being articulate, punctual, smart in appearance etc. The school improvement risk related to the Council's changed role in challenging its schools to raise standards. The specific focus on maths would be picked up in an operational risk register.

Responding to a question from Cllr Mrs Thompson about whether the Council was following other local authorities and providing Saturday classes to support children in passing their maths examinations earlier. Simon Warren reported that he would ask the appropriate Assistant Director to respond to her direct.

Mr Ager reported that for an organisation with eight high risk items on its risk register, it was challenging. He suggested that the Committee needed to receive assurances that appropriate arrangements are in place to address the risks. Narinder Phagura, Strategic Risk Manager confirmed that at its next meeting the Committee would receive more detail behind the items on the corporate risk register. A corporate assurance map would also be developed to identify where the Committee could gain assurance on such activity and this would be presented to the Committee in due course.

Resolved:

That the latest summary of the Council's corporate risk register be noted.

INTERNAL AUDIT REPORTS

8 Internal audit update-quarter three

The Committee received a report on progress made against the 2013/14 internal audit plan and on recent work which had been completed. Peter Farrow, Head of Audit drew the Committee's attention to the three limited opinion reports issued and

advised that they were not particularly high risk issues and action had been taken to address them. Referring to the appraisals review update that the Committee discussed at its last meeting, he advised that the findings to date had demonstrated low compliance with the corporate appraisals process. Audit Services were following up on questionnaires issued to heads of service and would present the results on the outcome of their review to the next meeting.

Cllr Keith Inston (Chair) reported that he was appalled at the level of response from head of service to the Audit Services questionnaires. He instructed that a letter be sent to the heads of service who had not responded instructing them to do so within seven days. The Chief Executive informed the Committee that he shared their concerns and undertook to arrange for a letter to be circulated in his name.

Referring to the financial implications of the report Cllr Ian Bolshaw asked whether all of the money from duplicate payments would be recovered. Peter Farrow reported that his team would be working with finance to hopefully recover all of the money. He also reported on the recovery of payments through the National Fraud Initiative.

Resolved:

- 1. That the content of the latest internal audit update at the end of quarter two be noted.
- 2. That the Chief Executive be requested to write to all heads of service who had not responded to the Audit Services questionnaires on appraisals, instructing them to respond within seven days otherwise the Committee would take appropriate action.

9 Internal audit plan 2014/15

Peter Farrow, Head of Audit presented the proposed internal audit plan for 2014/15

Cllr Wendy Thompson asked PwC for their view on the 'level of materiality' at the Council. In response she was advised that it was 2% of the Council's gross service expenditure. Further details were set out on page 12 of their external audit plan 2013/14 (Minute No. 6 above refers).

Cllr Dr Mike Hardacre expressed concern regarding self-started academies and single starter academies and the opportunities for things to go wrong in school finances. He reported that the Council had some control over local authority schools but he questioned what control the Council had over academies on their accounting and audit financial control. Kate Morgan advised that the School's Education Funding Agency was responsible for academies. The Chief Executive added that the Council was a co-sponsor the city's Academy trusts, and therefore had an elected member involved with the governance of the Trusts.

Terry Day, Independent Member reported that he was pleased to see ICT contract procurements and processes included in the audit plan but indicated that he could not see anything in the plan regarding technical audits proposals around ICT services. Peter Farrow advised that a lot of this aspect would be picked up with the introduction of the Agresso system under the FutureWorks programme. He

undertook to go through the plan and highlight for Mr Day where the technical aspects would be.

Resolved:

That the risk based internal audit plan for 2014/15 be approved.

10 Payments transparency

Peter Farrow, Head of Audit reported on the latest update position on the Council's payment transparency activity since the last meeting of the Committee.

Cllr Phil Bateman requested the details of the web link for the Spotlightonspend, that could be accessed via the Council's transparency and accountability internet pages.

Cllr Wendy Thompson asked about a payment of £12,500 to the Plough and Harrow. Mark Taylor, Assistant Director Finance reported that it was a sum of £10,800 relating to a grant for building works to the Plough and Harrow. Referring to recent media interest in items identified as a result of payment transparency Mark Taylor also reported that the Council provided the raw data on all its spending to. Spikes Cavell which then undertook the necessary review and analysis before publishing on Spotlightsonspend. Employees within his team were reviewing any foreign transactions to ensure that they are converted and in the right currency.

Resolved:

That the Council's continued compliance with the Code of Recommended Practice, regarding the publication of all payments over £500 be noted. Also, that from 1 April 2013 the Council had continued to publish all of its payment data, rather than just restrict it to that with a value of over £500

11. Review of fraud related policies and procedures

Peter Farrow presented four updated Council fraud related policies and procedures. The documents had been produced following a review of a number of policies, plans and guides and combining them into four new documents that had been updated to reflect best practice.

In response to questions from the Committee's two independent members, it was confirmed that the annual report summarising the activity undertaken under the whistleblowing policy would be submitted to the Committee and key details of concerns raised under the policy and their outcomes would be reported to the Monitoring of Audit Investigations Sub -committee. Awareness raising activities, using a variety of approaches would also be taking place to alert employees to the updated policies and procedures.

Resolved:

That the following updated Council's fraud related policies and procedures be noted:

- Whistleblowing policy and procedure
- Anti-fraud and corruption policy procedure
- Anti-money laundering policy and procedure
- · Raising fraud awareness guide.

12. The FRAUDitor: The Council's Anti-Fraud Corruption Newsletter

The Committee received, for information, issue number two - spring 2014 of the Council's anti-fraud and corruption newsletter produced by Audit Services

Resolved:

That the newsletter be received and noted

13 Audit Committee Terms of Reference

Peter Farrow, Head of Audit presented a report on updated terms of reference for the Committee which reflected the Chartered Institute of Public Finance and Accountancy's (CIPFA's) new position statement, "Audit Committee in Local Authorities. The report also included proposals to absorb the work of the Final Accounts Monitoring and Review Sub- committee within the main Audit Committee and to increase the membership of the Monitoring of Audit Investigations Sub-committee to reflect its expanding role and to enable additional councillors to play a more active part in this high profile area.

Resolved:

- 1. That it be recommended to the Special Advisory Group that the work of the Final Accounts Monitoring and Review Sub-committee be absorbed into the main Audit Committee terms of reference.
- That it be recommended to the Special Advisory Group that the membership of the Monitoring of Audit Investigations Sub-committee be increased from four to six members.
- 3. The updated terms of reference for the Committee that reflects the new Chartered Institute of Public Finance and Accountancy (CIPFA) Practical Guidance for Local Authorities Audit Committees be noted.

14. Exclusion of press and public

Resolved:

That in accordance with Section 100A(4) of the Local Government Act 1972, the press and public be excluded from the meeting for the following item(s) of business as they involve the likely disclosure of exempt information falling within the paragraphs of Schedule 12A of the Act set out below:

Item No.	Title	Applicable paragraph
15.	Audit investigation education and enterprise overspend	2 and 3
16.	Audit investigation - property services recharges	2 and 3

15. Audit investigation education and enterprise overspend

The Committee received a report on an investigation undertaken into the Education and Enterprise directorate overspend of £1.8 million. A number of finance management failures were identified. A detailed internal audit report had been issued for this investigation which included a series of recommendations that have been agreed and were in the process of being or had already been implemented.

Keith Ireland, Strategic Director Delivery reported on action in relation to employees referred to in the report. He also reported on the structural issues within the Council's budget which had been put right.

Cllr Wendy Thompson reported that it seemed that the areas of concern had been put right. She suggested that they should have been identified earlier and asked why this had not occurred. She added that the failure had caused an extra problem to the Council because of the considerable amount of money involved. In response Keith Ireland advised that historically when the overall budget was in a better position there may have been a tendency not to focus on these sorts of issues.

Mr Mike Ager asked how the Committee could be assured that there were no more similar issues waiting to be uncovered. In response Keith Ireland explained that a lot of work had been done to put things right. He explained how the Council's ageing mainframe computer had contributed to the situation occurring. He went on to inform the Committee that he was confident that the vast majority of issues in the budgets are now recharged correctly. The two audit reports were specific to Education and Enterprise and Property Services. He also reported that the Council's external auditors have been contacted regarding the two issues and concurred with the Council's approach to addressing them.

Resolved

That the outcome of an internal audit investigation on the 2012/13 overspend within the Education and Enterprise directorate be noted.

16. Audit investigation - property services recharges

The Committee received a report on an investigation undertaken into the recovery of charges by the Property Design and Commissioning Service. The investigation had highlighted a number of significant failures. A detailed internal audit report had been issued which included a series of recommendations that had been agreed and were in the process of being or already been implemented.

The Committee was informed that following a meeting with PwC it had been agreed that the Committee would receive follow up reports on both audit investigation to review the progress on actioning the recommendations from the audit investigations. This would include an update on the action taken to recover some of the charges.

[NOT PROTECTIVELY MARKED]

Cllr Wendy Thompson asked how has the culture within the service area had changed. Keith Ireland advised that the culture had not been changed yet. A report was being taken forward to create a corporate landlord function. Following this, work would be undertaken to review the corporate assets at which point changes to the culture would be addressed.

Resolved

That the outcome of the Internal Audit investigation on the 2012/13 under recovery of property services recharges be noted.





Audit (Monitoring of Audit **Investigations) Sub-committee**

Minutes - 28 April 2014

Attendance

Members of the Sub-committee

Cllr Wendy Thompson (chair) Cllr Mike Hardacre Cllr Matthew Holdcroft

Cllr Keith Inston

Employees

Peter Farrow Head of Audit Mark James Fraud Manager Katy Morgan Client Lead Auditor Richard Morgan Senior Audit Manager Mark Wilkes **Principal Auditor**

Dereck Francis Democratic Support Officer

Part 1 – items open to the press and public

Item Title

No.

MEETING BUSINESS ITEMS

1. Apologies for absence

An apology for absence was submitted on behalf of Mike Agar (Independent member).

2. **Declarations of interest**

Cllr Dr Mike Hardacre declared a personal interest in agenda items 7 and 8 in so far as they relate to the City of Wolverhampton College, the Central Learning Partnership Board and Woodthorne Primary School.

3. Minutes of the previous meeting (20 January 2014)

Resolved:

That the minutes of the previous meeting held on 20 January 2014 be approved as a correct record and signed by the Chair.

4. **Matters arising**

There were no matters arising from the minutes of the previous meeting.

5. Audit Services – Counter Fraud April 2014

Mark Wilkes presented a report which updated the Sub-committee on the current counter fraud activities undertaken by the Counter Fraud unit within Audit Services.

Mark James also outlined the salient points of the update report regarding benefit fraud investigations. In doing so he reported on two benefit fraud cases which had been reported in the local media. The investigations had resulted in successful prosecutions with one claimant receiving 18 months imprisonment. The other case had led to the false claimant receiving a three months suspended prison sentence, a three month curfew and a supervision order.

Cllr Mrs Thompson reported that it was good that benefits cheats were being caught and successfully prosecuted. She commented that as resources were finite fraudulent claims were taking resources away from people who are legitimately in need of that support. She also welcomed the increased joint working between the Council's Benefit Fraud team and the Department for Work and Pensions, and Wolverhampton Homes for tenancy and right to buy fraud.

Mark James also informed the Sub-committee that discussions were taking place regarding some of the Council's current benefit fraud cases being featured in a prime time evening tv documentary 'Briton of the Fiddle'.

Resolved:

That the contents of the latest Internal Audit counter fraud update be noted.

EXCLUSION OF PRESS AND PUBLIC

6. Exclusion of press and public

Resolved:

That in accordance with Section 100A(4) of the Local Government Act 1972, the press and public be excluded from the meeting for the following item(s) of business as they involve the likely disclosure of exempt information falling within the paragraphs of Schedule 12A of the Act set out below:

Item No. Title Applicable paragraph7 Audit update 1, 2, 3

Part 2 – exempt items, closed to the press and public

7. Audit update

The Sub-committee received updates on current audit investigations.

Resolved:

1. That the current position with regard to the audit investigations be noted.

[NOT PROTECTIVELY MARKED]

- 2. That in connection with case reference IL17W Building Schools for the Future Public Arts Works, a lessons learned report be submitted on the investigations that should include a clear mandate of what should happen in the future under similar circumstances.
- 3. That in connection with case reference IL61W Stolen laptop, periodic spot checks be undertaken to ensure that the controls introduced are being adhered to.
- 4. That in connection with case reference IL671 Potential mis-use of p-card, an update report be submitted to a future meeting on the ongoing investigation to include action taken by the Chair of the School Governing Body and the investigations by the Assistant Director.



Agenda Item No: 5



Audit Committee

14 July 2014

Report Title Internal Audit Report – Performance Appraisal

Scheme

Cabinet Member with Lead Responsibility Councillor Andrew Johnson

Resources

Accountable Strategic

Director

Keith Ireland, Delivery

Originating service Delivery/Audit

Accountable employee(s) Peter Farrow Head of Audit Tel 01902 554460

Email peter.farrow@wolverhampton.gov.uk

Report to be/has been

considered by

Not applicable

Recommendations for noting:

The Committee is asked to note:

The outcome of the recent internal audit review into the Council's new Performance Appraisal Scheme.

1.0 Purpose

1.1 The Audit Committee have been concerned for some time about the low take up of the Council's new Performance Appraisal Scheme, and asked to be made aware of the outcome of the planned internal audit review into this area.

2.0 Background

2.1 The Council's new performance appraisal scheme was introduced in April 2013. It has been understood for some time that the completion rate for the new appraisals was very low. The internal audit report confirms this, and has identified a number of issues that may have contributed to such a low take up.

3.0 Progress, options, discussion, etc.

3.1 Progress on the implementation of the actions in the internal audit report, will be reported back to the Audit Committee.

4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/27062014/M)

5.0 Legal implications

5.1 There are no direct legal implications arising from the recommendations in this report. (RB/26062014/G)

6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

9.0 Corporate landlord implications

There are no corporate landlord implications arising from the recommendations in this report.

10.0 Schedule of background papers – None



Draft Internal Audit Report – Performance Appraisal Scheme

Date Issued: June 2014

Contents:

- 1. Introduction
- 2. Executive summary
- 3. Issues arising

Report distribution:

Keith Ireland – Strategic Director, Delivery
Tim Johnson – Strategic Director, Education and Enterprise
Sarah Norman – Strategic Director, Community
Sue Davies – Chief Human Resources Officer



1 Introduction

An audit of the Corporate Performance Appraisal scheme has been undertaken as part of the Council's Internal Audit Plan. In April 2013, a new performance appraisal scheme was launched across the Council. This was communicated to all employees via City People and managers were emailed directly informing them of the scheme and offering support in completing the process. The completion of performance appraisals is being monitored as one of the Council's 51 performance measures which are used to monitor progress towards delivery of the Corporate Plan. Target completion levels have been set for each quarter. However, the table below illustrates that there is a considerable shortfall in meeting these targets.

	Target % (Cumulative)	Actual % (Cumulative)
Quarter 1	25	1
Quarter 2	50	3
Quarter 3	75	10
Quarter 4	100	25

The current RAG rating for this measure on the Corporate Measures report is **Red**.

Our review also aimed to establish if there were any underlying trends that could be identified which are contributing to the problems in embedding the performance appraisals.

It is envisaged that Agresso will improve the monitoring of performance appraisals once the first appraisals have been completed and recorded on the system. However, as the majority of problems identified relate to achieving completion in the first place, then problems may still exist despite the implementation of Agresso.

1.1 Scope and objectives of audit work

The objective of our audit was to deliver reasonable assurance on the adequacy and application of the risk management and associated control framework. The control system is put in place to ensure that risks to the achievement of the Council's objectives in this area are managed effectively. Our audit considered the Council's objectives for the area under review and the potential risks to the achievement of those objectives.

System Objective	Potential Risks
All employees should receive an annual appraisal from their manager in	 Failure to meet corporate performance measures established to demonstrate that Wolverhampton is a confident, capable Council.
accordance with Corporate Policy	 Low staff morale leading to poor performance and subsequent non delivery of services.



Page 18

As part of our review we also issues a series of questionnaires to a sample of managers in order to obtain their views on the new performance appraisal scheme, and their views have helped inform sections of this report.

2 Executive summary

2.1 Overall conclusion

Taking account of the issues identified in this report, in our opinion the controls within the system, as currently laid down and operating, provide **limited assurance** that risks material to the achievement of the Council's objectives for the system are adequately managed and controlled.

Definitions for the levels of assurance that can be given:

Substantial

 There is a robust framework of controls which ensures that objectives are likely to be achieved and controls are applied continuously or with only minor lapses

Satisfactory

 There is a sufficient framework of key controls for objectives to be achieved but the control framework could be stronger or the application of controls could be more consistent

Limited

 There is a risk of objectives not being achieved due to the absence of key internal controls or a significant breakdown in the application of controls

2.2 Key issues identified

This report has been prepared on an exception basis. Therefore, we have only reported on areas where action is required to address weaknesses in controls or to limit exposure to risk.

We have identified one **red** and four **amber** and two **green** issues, arising from the following:

- Low number of appraisals completed, including at a senior level
- Inaccurate reporting of performance information
- Low participation in appraisal training



Page 19

- Unrealistic profiling of performance measure targets
- Lack of clear lines of responsibility
- Insufficient promotion of the scheme
- Work plans not always fully completed

Suggested / agreed actions have been made for these areas and they are shown in the main body of the report. Consideration should be given to, where appropriate, feeding any red or amber issues from this report, into the relevant risk management process in order to help manage any associated risks. Also, the key issues arising from this report may be included in summary form to the Audit Committee.

2.3 Acknowledgement

A number of staff gave their time and co-operation during the course of this review. We would like to record our thanks to all of the individuals concerned.



Page 20

3 Issues arising and suggested/agreed actions

Priority rating for issues identified

Red - action is imperative to ensure that the objectives for the area under review are met.

Amber - requires action to avoid exposure to significant risks in achieving the objectives for the area under review.

Green - action is advised to enhance risk control or operational efficiency.

No	Priority	Issues arising				Action to address issues	Responsibility	Target date	
3.1 Page 2	Red	Low number of appraisals Within Council policy it is a rundertake a performance apmanage on an annual basis a very poor take up on the number the Council. The new appraisance appraisance and in the first 12 mor of employees had received appraisance and second s	mandatory requiperaisal with the Despite this resumber of appraisal scheme wanths, as the tab	irement of man e employees that equirement there hisals undertake s introduced on	agers to at they e has been en across n 1 April	All managers should be regularly reminded through a number of channels, of the need to carry out, and log regular performance appraisals in line with the requirements of the Council's policy. The number of appraisals			
_			Directorate	Approximate Appraisals completed*	Approximate Employee Count @ 31 March 2014*	Take up	undertaken against performance targets should be monitored at regular intervals by each directorate's management team,		
		Community	240	2,400	10%	and corporately by the Strategic Executive Board. Formal intervention should be taken in areas of continued noncompliance.			
		Delivery	1,200	2,400	50%				
		Education and Enterprise	150	1,500	10%				
		OCE	15	40	37%				
		Total	1,605	6,340	25%				
		* These figures are approximate a and in paragraph 3.2, the lack of a							



No	Priority	Issues arising	Action to address issues	Responsibility	Target date
Page 22		difficult to establish the exact figures. Within these figures, the number of more senior employees receiving an appraisal, and employees based away from the Civic Centre were particularly low. A number of reasons for such a low take up were offered from across the Council, including: • A lack of awareness of the new appraisal process/differing levels of success in communicating the new process (this matter is further commented on in paragraph 3.6); • Appraisals being undertaken but not being logged on the performance appraisal system; • Time constraints; • Awaiting confirmation of restructures and ongoing uncertainties around changes in line management; • Employees awaiting completion of their own appraisal before undertaking others; • Appraisals completed, but awaiting the return of completed forms/documentation from employees; • Potential restructures; • Appraisals delayed to align with the previous Employee Performance Review Scheme cycle. Implication: The Council's Performance Appraisal and Personal Development Policy is not being followed. Without an appraisal managers and employees do not have the opportunity to review performance or ensure that achievements, strengths and areas for improvement are recognised and acknowledged, personal development needs are not identified and managers are not providing the appropriate support.			



No	Priority	Issues arising	Action to address issues	Responsibility	Target date
Page 23	hampton	 Inaccurate reporting of performance information Inaccurate data on the number of appraisals being undertaken was being reported in the corporate performance report. There were a variety of reasons behind this, predominantly based around difficulties when attempting to reconcile the information held on the performance appraisal system to that on the corporate performance management system TENS, including: No clear definition for the "% of appraisals completed" figure - the performance appraisal system can be updated retrospectively (i.e. appraisals can be added at a much later date) whereas TENS uses data within specific time frames and makes use of cut-off points. Calculations using the total number of NI numbers appearing as live payroll records, but excluding school based and Pensions Fund employees. Employees who have been directly employed by the Council for less than six months are not required to have an appraisal. However, these employees are not excluded from the total number of employees. Whilst the policy does not specifically state that these employees. Whilst the policy does not specifically state that these employees are not required to be appraised, it would seem reasonable to exclude them from the process. Uncertainty over how to deal with employees who have more than one job at the Council and if they need a performance appraisal for each role. At the moment they are recorded in the figures as doing so. Implication: Inaccurate information on the number of appraisals completed/overall performance reported in corporate performance reports. 	The corporate performance measure "% of appraisals completed" should be clearly defined and a data quality review undertaken to ensure the validity of the data provided within the corporate performance report. Following such a review, any required clarification to the performance appraisal policy, particularly around the comments made in this report and the subsequent impact they have on determining and measuring performance, should be addressed.		page 6

No	Priority	Issues arising	Action to address issues	Responsibility	Target date
3.3 Po	Amber	Low participation in appraisal training A range of training opportunities to support managers in undertaking performance appraisals, and raise awareness of the Council's Policy, can be accessed through the Council's Learning Hub or a face to face Skills Gym. We reviewed these and found them to offer a good level and quality of training. However, an analysis of training undertaken in the first eight months of the scheme highlighted that only 20 employees had completed e-learning modules, although over 200 employees had attended the Skills Gym. Implication Low level of engagement with training opportunities may reduce its effectiveness, and contribute towards the low completion rates.	Managers who have failed to carry out appraisals, or feel they have a lack of awareness of what is required, should be encouraged to engage in the various training/policy raising opportunities available to them.		
Page 24	Amber	Unrealistic profiling of performance measure targets Given the unequal pattern in which performance appraisals have historically been undertaken, the assumption of an equal profile of target percentages across the year (i.e. 25% of performance appraisals to be completed in each quarter) and that 100% would be achieved within the first 12 months, would appear unrealistic. Implication: Adverse pressure and poor performance reporting through unrealistic expectations.	The profiling of the corporate performance measure target should be reviewed to ensure that it is both realistic and achievable.		



No	Priority	Issues arising	Action to address issues	Responsibility	Target date
3.5 Page 25	Amber	Lack of clear lines of responsibility During our review it was difficult to identify exactly who was responsible for undertaking performance appraisals due to the lack of readily available and current structural information and clear lines of accountability. It is envisaged that once fully operational, the Agresso system should improve this situation. Also, at the time of our review, there were approximately 240 agency/ interim/consultancy staff working at the Council. Given the issues identified above, it was not possible to identify which of these officers, if any, were responsible for completing performance appraisals, and if they were, what induction process they had received on this policy. Finally, we noted wide ranging differences in the number of appraisals being carried out by individuals, with many as low as performing just one, but in other cases, a much more significant number. This suggests that the levels of delegation for undertaking appraisals could be improved upon. Implication: Lack of accountability contributing towards the low number of appraisals being undertaken.	Accurate and up to date structural charts with clear lines of accountability should be maintained to identify responsibility for the completion of performance appraisals (including agency/interim/consultancy staff). All such accountability should be kept within reasonable and manageable limits.		



No	Priority	Issues arising	Action to address issues	Responsibility	Target date
3.6 Page 26	Green	Insufficient promotion of the scheme From a review of the Council's intranet, only two articles on the performance appraisal scheme could be found. One from March 2013, at the time of the launch and the other in June 2013, promoting the training available to support managers undertaking appraisals. Later in March 2014 the Chief Executive issued an email to all managers, raising the issue of non-completion of performance appraisals. Given the persistent low rates of completion and the overall red RAG rating in the corporate performance report, higher levels of promotion in order to raise employee awareness would have been expected. Implication: The high priority awarded by the Council to the performance appraisal scheme was not supported by the level of activity promoting the scheme, contributing to the low take up.	A more pro-active promotion of the performance appraisal scheme should be undertaken through mechanisms such as the intranet/Agresso/payslip messages. Further promotion activities should also be investigated at a local/directorate level		
3.7	Green	Work plans not always fully completed As part of each appraisal, an individual work plan should be prepared, setting objectives and key actions for the employee for the year ahead. From our review of a number of appraisals that had been undertaken, we noted that these work plans were not always being fully and correctly completed. Implication: Maximum benefit from each appraisal may not be obtained and objectives and key actions for employees may not be fully identified.	Managers carrying out appraisals should be reminded of the need to follow the official guidance, and fully complete the formal documentation (including the work plan).		



Limitations inherent to the internal auditor's work

This report has been prepared solely for Wolverhampton City Council in accordance with the terms and conditions set out in the terms of reference. Internal audit does not accept or assume any liability of duty of care for any other purpose or to any other party. This report should not be disclosed to any third party, quoted or referred to without prior consent. Internal audit has undertaken this review subject to the limitations outlined below.

Internal control

 Internal control systems, no matter how well designed and operated, are affected by inherent limitations. These include the possibility of poor judgement in decision making, human error, control processes being deliberately circumvented by employees and others, management overriding controls and the occurrence of unforeseeable circumstances.

Responsibilities of management and internal auditors

- It is management's responsibility to develop and maintain sound systems of risk management, internal control and governance for the prevention and detection of irregularities and fraud. Internal audit work should not be seen as a substitute for management's responsibilities for the design and operation of these systems.
- Internal audit endeavours to plan audit work so that it has a reasonable expectation of detecting significant control weakness and if detected, will carry out additional work directed towards identification of consequent fraud or other irregularities. However, internal audit procedures alone, even when carried out with due professional care, do not guarantee that fraud will be detected.
- Accordingly, these examinations by internal auditors should not be relied upon solely to disclose fraud or other irregularities which may exist.





Audit Committee

14 July 2014

Report title Annual Review of the Effectiveness of Internal

Audit

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s) Peter Farrow Head of Audit

Tel 01902 554460

Email peter.farrow@wolverhampton.gov.uk

Report to be/has been

considered by

Not applicable

Recommendation(s) for action or decision:

The Committee is recommended to:

1. Approve the annual review of the effectiveness of internal audit in order to discharge their responsibility under Regulation 6 of the Accounts and Audit Regulations 2011, where "the relevant body shall, at least once in each year, conduct a review of the effectiveness of internal audit".

1.0 Purpose

- 1.1 This report provides the Committee with sources of information and measures in place in order to assist in being able to reach a conclusion on the adequacy and effectiveness of the Internal Audit Service.
- 1.2 The definition of internal audit, as described in the Public Sector Internal Audit Standards, is set out below:
 - "Internal auditing is an independent, objective assurance and consulting activity designed to add value and improve an organisation's operations. It helps an organisation accomplish its objectives by bringing a systematic, disciplined approach to evaluate and improve the effectiveness of risk management, control and governance processes".
- 1.3 The Accounts and Audit Regulations 2011 state that the relevant body must "maintain an adequate and effective system of internal audit of its accounting records and of its system of internal control in accordance with the proper practices in relation to internal control". For local authorities the proper internal control practices for internal audit are those contained within the Public Sector Internal Audit Standards.
- 1.4 At paragraph 6 the regulations also state that "the relevant body shall, at least once in each year, conduct a review of the effectiveness of internal audit" The regulations go on to say that the findings of this review be considered by a committee of the relevant body, or by the members of the relevant body as whole, as part of the wider consideration of the system of internal control. At the Council, the Audit Committee perform this role.

2.0 Background

- 2.1 The Council's arrangements for Internal Audit are contained within the Constitution, which states that "The Section 151 Officer shall ensure that an adequate and effective internal audit of all Council activities is carried out in accordance with the most recent CIPFA Statements on Internal Audit Practice and relevant legislation."
- 2.2 Throughout 2013/14 Internal Audit followed the Public Sector Internal Audit Standards, which cover both attribute and performance standards.
- 2.3 In reviewing the effectiveness of internal audit, the following have been taken into consideration:
 - The process by which the control environment and key controls have been identified.
 - The process by which assurance has been gained over controls.
 - The adequacy and effectiveness of the remedial action taken where there are deficits in controls.
 - The operation of the Audit Committee and the internal audit function to current codes and standards.
 - An assessment against the five principles contained in the CIPFA document "The

Role of the Head of Internal Audit".

Further details on each of these are detailed below.

2.4 The process by which the control environment and key controls have been identified

The control environment and key controls of the Council are identified through the risk management system. Risks to the achievement of the Council's priorities are identified and mitigated within the control environment through risk management, with risk registers at various levels across the Council.

2.5 The process by which assurance has been gained over controls

Assurance is gained over controls through a number of sources including the work of Internal and External Audit (currently PwC). Internal Audit produce a risk based Internal Audit Plan, which is driven by the risks the Council faces. They also report progress on the delivery of this plan on a quarterly basis to the Audit Committee, and for 2013/14 produced an Annual Internal Audit Report, which gave the following opinion:

"Based on the work undertaken during the year, the implementation by management of the recommendations made and the assurance made available to the Council by other providers as well as directly by Internal Audit, Internal Audit can provide reasonable assurance that the Council has adequate and effective governance, risk management and internal control processes".

2.6 The adequacy and effectiveness of the remedial action taken where there are deficits in controls

Where deficits in internal control are identified, both Internal and External Audit will make recommendations that are entered onto an action plan. Management are then responsible for identifying what remedial action will be undertaken, who by and by when. Internal Audit will then follow up all key recommendations and report back on any significant non-compliance to the Audit Committee.

2.7 The operation of the Audit Committee and the internal audit function to current codes and standards

Internal Audit

In reviewing the internal audit function, the following have been taken into consideration:

- The Quality and Improvement Programme for internal audit during the year, the
 internal audit activity has followed this programme and there have been no
 significant areas of non-conformance or deviations from the standards as set out
 in the Public Sector Internal Audit Standards.
- The ongoing views of the Audit Committee.
- The external auditor comments from their latest Annual External Audit Reports.
- The internal audit self-assessment checklist included in the CIPFA Application Note that accompanies the new Public Sector Internal Audit Standards that came into effect from 1 April 2013.
- Ongoing dialogue held with key officers throughout the Council on the role internal audit plays.

Internal Audit also seeks feedback from each audit they undertake by providing a service quality questionnaire for the auditee to complete and return. Feedback from these has been positive throughout the year, and further details can be provided upon request.

Finally a series of Internal Audit effectiveness measures have been agreed by the Audit Committee, and good progress has been made against these and is reported in more detail in the Annual Internal Audit Report.

Audit Committee

The Audit Committee operate within a Terms of Reference based on the model provided by CIPFA in their Audit Committees – Practical Guidance for Local Authorities 2013 edition.





Assessment of the five principles contained in CIPFA's "Role of the Head of Internal Audit in public service organisations"

This document includes five principles for the Head of Internal Audit:

- Championing best practice in governance, objectively assessing the adequacy of governance and management of existing risks, commenting on responses to emerging risks and proposed developments.
- Giving an objective and evidence based opinion on all aspects of governance, risk management and internal control.
- Must be a senior manager with regular and open engagement across the organisation, particularly with the Leadership Team and the Audit Committee.
- Must lead and direct an internal audit service that is resourced to be fit for purpose.
- Must be professionally qualified and suitably experienced.

Underpinning each of the above five principles in the CIPFA document is a series of governance requirements, core responsibilities and personal skills and professional standards. As part of this review the current arrangements have been assessed against these requirements and found to be in compliance.

2.8 The Council's Head of Internal Audit's background and experience

The Head of Audit is a qualified accountant (FCCA) and also has the following experience:

- Executive Board member and Treasurer for the National Anti-Fraud Network (NAFN)
- Former Chair and a current member of the Organising Committee for the CIPFA in



the Midlands Audit Training Seminars (CATS)

- Chair of the Audit Committee and Governor at an FE college
- A regular speaker on internal audit matters for a number of organisations, including CIPFA and the Chartered Institute of Internal Auditors (CIIA)

The post is a shared service role as part of an arrangement with Sandwell MBC, and the current Head of Audit throughout 2013/14 acted in a similar role for a number of Midlands based public sector organisations, including:

- Wolverhampton City Council
- Sandwell Metropolitan Borough Council
- West Midlands Pension Fund
- West Midlands Fire Service

3.0 Progress, options, discussion

3.1 The Internal Audit Service will continue to report back to the Audit Committee on all relevant control issues throughout the year.

4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/23062014/B)

5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report. (AS/20062014/E)

6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations in this report

10.0 Schedule of background papers

10.1 Accounts and Audit Regulations (2011)

Public Sector Internal Audit Standards and accompanying CIPFA local government application note

CIPFA Role of the Head of Internal Audit in Public Service Organisations Audit Committees – Practical Guidance for Local Authorities 2013 Edition Wolverhampton City Council Constitution

Agenda Item No: 7



Audit Committee

14 July 2014

Report title Audit Committee Annual Report – 2013/14

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s) Peter

Peter Farrow Head of Audit

Tel 01902 554460

Email <u>peter.farrow@wolverhampton.gov.uk</u>

Report to be/has been

considered by

Not applicable

Recommendation(s) for action or decision:

The Committee is recommended to:

Endorse the Audit Committee Annual Report for 2013/14 and refer it to Full Council for approval.

1.0 Purpose

1.1 This report summarises the main areas of work undertaken by the Audit Committee during 2013/14.

2.0 Background

- 2.1 The Audit Committee is a key component of the Authority's corporate governance. It provides an independent and high-level focus on the audit, assurance and reporting arrangements that underpin good governance and financial standards.
- 2.2 The purpose of the Audit Committee is to provide independent assurance to the Members of the adequacy of the risk management framework and the internal control environment. It provides independent review of the governance, risk management and control frameworks and oversees the financial reporting and annual governance processers. It oversees internal audit and external audit, helping to ensure efficient and effective assurance arrangements are in place:
- 2.3 The key benefits of the Committee can be seen as:
 - Increasing public confidence in the objectivity and fairness of financial and other reporting.
 - Reinforcing the importance and independence of internal and external audit and similar review processes.
 - Providing additional assurance through a process of independent review.
 - Raising awareness of the need for internal control and the implementation of audit recommendations.2.4 The Audit Committee agrees a work programme for each year, which is refreshed on an on-going basis, and is based on (but not limited to) the following main sources of assurance:
 - Accounts/Annual Governance Statement These are the statutory reports which the Audit Committee approves in relation to the Council's Annual Statement of Accounts.
 - Internal Audit Reports These relate to the ongoing work of the internal audit and allow the Audit Committee to approve and monitor the audit plan and maintain an awareness of ongoing control issues.
 - Risk Management These reports relate to the Council's risk management arrangements.
 - External Audit and Inspection Reports These are the reports submitted to the Audit Committee by the Council's external auditors PricewaterhouseCoopers (PwC) and other inspection agencies.
 - Other Governance Issues These constitute other areas of governance which the Audit Committee needs to keep under review.
- 2.5 The Audit Committee also has a Monitoring of Audit Investigations Sub-Committee which receives reports on the activity the Council is taking in order to combat fraud, its participation in national anti-fraud activities, and on the nature and outcome of investigations taking place within the Council.

- 2.6 During the year the Terms of Reference for the Audit Committee were updated in order to respond to guidance provided in the Chartered Institute of Public Finance and Accountancy (CIPFA) Audit Committees Practical Guidance for Local Authorities 2013 Edition:
- 2.7 Key areas of business reviewed by the Committee during the year, were as follows:

Meeting	Activity
8 July 2013	Work Programme Annual Review of the Effectiveness of Internal Audit Audit Committee Annual Report Draft Statement of Accounts 2012/13 Annual Governance Statement 2012/13 External Audit Update Report 2012/13 Corporate Risk Register Update Annual Internal Audit Report 2012/13 CIPFA Audit Committee Update Payment Transparency
22 September 2013	Work Programme Audit Statement of Accounts 2012/13 External Audit Report to those charged with Governance 2012/13 Medium Term Financial Strategy Review – PwC Risk Management Benchmarking Internal Audit Update Q1 Audit Services Management Arrangements Update CIPFA Audit Committee Update Payment Transparency
16 December 2013	Work Programme Annual External Audit Letter Risk Management and Corporate Risk Register Internal Audit Update Q2 Audit Committee Self-Assessment Payment Transparency Benefit Fraud Sanctions 2012/13 Benefit Fraud Sanctions - Amendment to Arrangements CIPFA Audit Committee Update Annual Governance Statement - Action Plan Update
10 March 2014	Work Programme External Audit Plan 2013/14 Corporate Risk Register Internal Audit Update Q3 Internal Audit Plan 2014/15 Internal Audit Investigations Payment Transparency Review of Fraud Related Policies and Procedures Anti-Fraud and Corruption Newsletter Audit Committee Terms of Reference

- 2.8 The Committee believes its key achievements during the year were:-
 - Providing assurance through a process of independent review and challenge.
 - Raising the profile of internal control issues across the Council and of the need to ensure that audit recommendations are implemented.
 - Regular consideration and review of the risks that the Council faces.
 - Maintaining a good working relationship with the Council's internal and external auditors.
 - Maintaining an awareness of the likely changes to the appointment of external auditors through the Local Audit and Accountability Act.
 - Building the skills and knowledge of Committee members through regular technical updates and the consideration of related guidance issued by CIPFA.
 - The continued presence of two independent members in order to broaden the Committee's experience and independent view point.
 - Maintaining a dedicated Sub-Committee in order to focus on the actions being taken to combat fraud.

3.0 Progress, options, discussion

3.1 The Audit committee work programme for 2014/15 will continue to be developed and presented at each meeting of the Audit Committee.

4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/23062014/L)

5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report. (AS/20062014/T)

6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations in this report

10.0 Schedule of background papers - None

Agenda Item No: 8



Audit Committee

14 July 2014

Report title Draft statement of accounts 2013/14

Cabinet member with lead

responsibility

Councillor Andrew Johnson

Resources

Accountable director

Keith Ireland, Delivery

Originating service

Strategic Finance

Accountable employee(s)

Mark Taylor

Assistant Director Finance

Tel 01902 556609

Email

mark.taylor@wolverhampton.gov.uk

Report to be/has been

considered by

None

Recommendations for noting:

The Committee is asked to note:

- 1. That the Assistant Director Finance approved the Draft Statement of Accounts 2013/14 on 30 June 2014, as required by the Accounts and Audit (England) Regulations 2011.
- 2. That the 2013/14 Draft Statement of Accounts is to be audited by PricewaterhouseCoopers LLP from July through to September, and that any material changes required as a result of the audit will be reported to the Audit Committee.
- 3. That formal approval by the council and publication of the 2013/14 Statement of Accounts is required by 30 September 2014 (Accounts and Audit (England) Regulations 2011).
- 4. That the Statement of Accounts incorporates a copy of the Annual Governance Statement as required by the Accounts and Audit (England) Regulations 2011.

1.0 Purpose

1.1 The draft Statement of Accounts for 2013/14, which is subject to audit, has been approved by the Assistant Director Finance and is appended to this report. The report provides an overview of the document and a summary of the Council's financial performance for the year and its position at 31 March 2014.

2.0 Background

- 2.1 The draft Statement of Accounts is required by statute (Accounts and Audit (England) Regulations 2011) to be prepared and approved by the Section 151 Officer by 30 June 2014. Once again the Council has succeeded in meeting this very challenging deadline.
- 2.2 A copy of the draft Statement of Accounts is attached at Appendix A. This will now be audited by the Council's appointed external auditors, PricewaterhouseCoopers, during August, following which they will report their findings to the Audit Committee in September. At the same time the final audited Statement of Accounts to be published by the Council will be presented to the Committee for approval.
- 2.3 The statutory deadline for publication of the audited Statement of Accounts is 30 September 2014.
- 2.4 The format of the Statement of Accounts is governed by the Code of Practice on Local Authority Accounting (the Code), published by the Chartered Institute of Public Finance and Accountancy (CIPFA). They are prepared in accordance with International Financial Reporting Standards (IFRS). There were no significant changes to the Code for 2013/14.

3.0 Structure of the statement of accounts

- 3.1 The purpose of the Statement of Accounts is to give all interested parties clear information about the Council's finances. In order to make the Statement of Accounts as useful as possible to its intended audience, the Code requires:
 - (a) All local authority statements to follow a common pattern of presentation;
 - (b) Interpretation and explanation of the Statement of Accounts;
 - (c) The Statement of Accounts to be written in plain English as far as possible.
- 3.2 The Statement of Accounts comprises the following:
 - (a) **Introduction to the Statements**, which provides a brief overview of each of the main parts of the statement.
 - (b) **Financial Performance 2013/14:** this section provides a summary of the Council's financial performance for the year, as shown in Cabinet outturn reports.

- (c) **The Medium Term Financial Strategy**: this section provides a summary of the council's medium term financial strategy, for each of the General Fund, the Housing Revenue Account and the Capital Programme.
- (d) **Statement of Responsibilities**: this sets out the respective responsibilities of the Council and the Section 151 Officer with regard to the accounts.
- (e) **The Auditor's Report:** this will be provided by PricewaterhouseCoopers following the audit, and gives their opinion on whether the Council's financial statements present fairly its financial position and performance for the year, and their conclusion on the council's Use of Resources arrangements.
- (f) **The Financial Statements:** this section comprises the financial statements themselves, which include:
 - (i) Comprehensive Income and Expenditure Statement: this statement shows all the income, expenditure, gains and losses of the council during the year prepared on an accounting basis. It includes several items which are not charges or credits to the council's funds, as statutory provisions exist to neutralise their effect on Council Tax and Housing rents.
 - (ii) Balance Sheet: this statement reports the council's financial position at the year end. It shows the balances and reserves at the council's disposal, the fixed and current assets employed in its operations and the long-term indebtedness of the council. (iii) Movement in Reserves Statement: this statement analyses the movements across all of the council's reserves, including the General Fund and HRA balances.
 - (iv) Cash Flow Statement: this statement summarises the inflows and outflows of cash during the year.
 - (v) Notes to the Core Financial Statements: this is a series of notes that are required by statute or that give additional information on figures in the main statements. The last of these notes is the Statement of Accounting Policies, which sets out the policies and, where appropriate, estimation techniques used by the council in preparing its accounts.
- (g) The Housing Revenue Account (HRA) Statements: comprising the HRA Income and Expenditure Account, the Statement of Movement in the HRA Balance, and Notes to the HRA Statements.
- (h) **The Collection Fund Statements:** comprising the Collection Fund Income and Expenditure Account and the Notes to the Collection Fund Statements.
- (i) **West Midlands Pension Fund Statements**: although West Midlands Pension Fund is a separate entity and its accounts are not consolidated with the council's, the Code requires administering bodies such as Wolverhampton to include a summarised version of the Pension Fund accounts in their financial statements.
- (j) **Annual Governance Statement:** This is a statement required by the Accounts and Audit (England) Regulations 2011 that outlines the council's governance arrangements.
- (k) **Glossary:** explaining some of the terms used in the financial statements, particularly those that are not part of everyday English.
- 4.0 Highlights from the 2013/14 Statement of Accounts

- 4.1 The net cost of providing services, shown in the Comprehensive Income and Expenditure Statement, was a deficit of £96.7 million. This compares to a deficit of £98.2 million for 2012/13. It is important to remember that this is prepared on an accounting basis and includes many transactions which are subject to statutory adjustment and do not impact on the council's usable reserves and therefore on the setting of council tax and housing rents.
- 4.2 The Comprehensive Income and Expenditure Statement also includes a number of other gains and losses on council assets/liabilities. These relate to revaluations of fixed assets and notional gains and losses on the council's 'share' of West Midlands Pension Fund's assets and liabilities. These items are responsible for very significant year-on-year changes: in 2013/14 they amounted to a net gain of £116.4 million, while in 2012/13 they caused a net loss of £129.1 million. Despite the size of these numbers and their significant impact on the accounts, it must be remembered that they are only included in order to comply with accounting requirements, and are reversed for statutory purposes.
- 4.3 The net worth of the council as shown on the Balance Sheet has increased to £388.0 million from £368.4 million. The main contributors to this are set out below:

	Comments	Change £m
Increases in Net Worth		
Fixed Assets	Expenditure on capital programme	93.1
	Revaluation of fixed assets	6.0
Current Investments	Increase in amount held in Money Market Funds and Deposit Accounts at year-end	8.0
Debtors/Creditors	Net increase in debtors/decrease in creditors	12.7
Pension Liability	Due primarily to changes in Actuarial assumptions and calculations (*)	89.8
Decreases in Net Worth Fixed Assets	Depreciation and Impairment Charges for the Year (*)	(85.0)
	Disposal of fixed assets	(74.4)
Borrowing	Net increase in borrowing during the year to fund capital expenditure	(44.8)
		Donort Do

- (*) Statutory provisions exist to neutralise the impact of these items on the council's resources available for revenue or capital expenditure.
- 4.4 The net increase in the council's reserves for the year, as shown in the Movement in Reserves Statement, was £19.6 million. This is made up of an increase of £3.7 million on usable reserves, and an increase of £15.9 million on unusable reserves (unusable reserves are technical accounting reserves, usually reflecting statutory adjustments). This compares to a net decrease of £227.3 million in 2012/13 (£28.4 million decrease in usable reserves; £198.9 million decrease in unusable reserves).
- 4.5 The Movement in Reserves Statement shows the movement in the council's General Fund and HRA balances for 2013/14. The General Fund Balance increased by £11.1 million, standing at £27.0 million as at 31 March 2014. The HRA Balance did not change and remains at £5.0 million.
- 4.6 The HRA Income and Expenditure Account shows a surplus of £14.9 million for the year (2012/13 surplus of £9.3 million). This surplus was used to provide for the redemption of debt as part of the HRA's capital financing strategy to create headroom against the debt cap to fund capital expenditure in the future.
- 4.7 The HRA reserve stands at £5.0 million at 31 March 2014. The funds in this reserve can only be used to finance revenue expenditure against the HRA budget in accordance with statute.
- 4.8 The Collection Fund generated a surplus for the year of £0.7 million. This will be distributed between the preceptors. This includes a provision made this year against the rateable valuation set by the Valuation Office Agency (VOA) not settled as at 31 March 2014.
- 4.9 The Group Comprehensive Income and Expenditure Statement shows a net deficit on the provision of services for the year of £98.7 million, compared to a net surplus of £100.5 million in 2012/13. The net loss for 2013/14 is made up of the council's net loss of £96.7 million and a net loss for Wolverhampton Homes of £2.0 million (2012/13: net loss of £98.2 million and £2.3 million respectively).

5.0 Financial Implications

5.1 The statement of accounts is one of the most important financial documents that the council produces. The statement, and the forthcoming audit of those statements by the external auditors, is fundamental to the accountability and transparency of the council's finances.

[CF/01072014/K]

6.0 Legal implications

6.1 The Accounts and Audit (England) Regulations 2011 require the 2013/14 Statement of Accounts be produced in accordance with proper practice. This is exemplified by the

Code of Practice on Local Authority Accounting which is published by CIPFA. These regulations also require that the accounts are approved by 30 June 2014 and published by 30 September 2014. [RB/04072014/R]

7.0 Equalities implications

7.1 There are no equality implications arising from this report.

8.0 Environmental implications

8.1 There are no environmental implications arising from this report.

9.0 Human resources implications

9.1 There are no human resource implications arising from this report.

10.0 Corporate landlord implications

10.1 There are no implications for the council's property portfolio arising from this report.

11.0 Schedule of background papers

11.1 There are no relevant preceding reports.

Statement of Accounts

2013/14





CONTENTS

Section	Title	Page
1	Introduction to the Statements	2
2	Financial Performance 2013/14	6
3	The Medium Term Financial Strategy	11
4	Statement of Responsibilities	18
5	Independent Auditors' Report to the Councillors of Wolverhampton City Council	20
6	The Financial Statements	21
7	The Housing Revenue Account Statements	118
8	The Collection Fund Statement	126
9	West Midlands Pension Fund Statements	130
10	Annual Governance Statement	182
11	Glossary	199

Page 47

1. INTRODUCTION TO THE STATEMENTS

Important note for readers of the accounts

Local authority accounts, like those of any organisation, are prepared to comply with a series of rules and conventions set by the accounting profession. However, for local authorities there are many types of transaction where the law, which takes precedence, requires a different treatment from the accounting rules. This effectively means that local authorities are trying to simultaneously fulfil two conflicting sets of rules when preparing their accounts.

This conflict is addressed by having authorities present a set of financial statements which comply with the accounting rules, followed by a reconciliation of those statements to the accounts as prepared under the legal rules. This reconciliation essentially takes the form of a list of adjustments for things which have to be in the accounts according to the accounting rules but are not allowed in them under law, and vice versa.

It is the legal rules that must be used when calculating budget requirements, council tax and housing rents. As a result, all of the council's internal reporting and decision-making is based purely on accounts prepared under the legal rules, and the only time it prepares accounts that comply with the accounting rules is when it prepares this document. It is crucial to bear this in mind when reading the statements. In particular, it should be remembered that figures which have been prepared under the accounting rules may have no practical meaning or use in the context of how the council actually manages its finances.

The purpose and contents of this document

The purpose of this document is to show the council's financial performance over the course of the year, and its financial position at the end of the year. It also provides some information about things that may affect the council's financial performance in the future.

Section 2 provides a summary of the council's financial performance for 2013/14, and key items of interest in the accounts. Section 3 provides an outline of the council's medium term financial strategy, including its budget for 2014/15 and forecasts through to 2018/19.

Section 4 contains the statement of responsibilities, and sets out the roles and responsibilities of the authority and of the Assistant Director Finance in preparing the statement of accounts. The independent auditors' report is included at section 5. This report draws readers' attention to any important information they might need to take into account when reading the statements.

1. INTRODUCTION TO THE STATEMENTS

Section 6 contains the financial statements prepared in accordance with the CIPFA Code of Practice on Local Authority Accounting in the United Kingdom (the Code). These comprise four main statements, and a series of notes. The four main statements are:

The Comprehensive Income and Expenditure Statement – this summarises all expenditure, income, gains and losses for the council during the year. It is important to remember that this statement is prepared entirely in accordance with accounting rules, which differ in several ways from the legal rules used to calculate budgets and available balances.

The Balance Sheet – this shows all of the council's assets, liabilities and reserves at the end of the financial year. Assets are either things that the council owns and can use or sell in the future, or money that it is owed by other people. Liabilities are money owed by the council to other people. Reserves fall into two categories: usable reserves are funds that the council has available to spend in the future, while unusable reserves are amounts that have come about purely from accounting adjustments and are not therefore available to spend.

The Movement in Reserves Statement – this shows the amounts in the council's reserves, and how they have changed over the course of the year.

The Cash Flow Statement – this summarises all of the council's payments and receipts over the course of the year. The fundamental difference between this statement and the Comprehensive Income and Expenditure Statement is that it doesn't include adjustments to comply with the accounting concept of accruals.

The notes to the accounts provide additional information about the main statements, or items that the council is required by law or by the Code to include in the statement. The notes are:

Note 1 – Financial Performance for 2013/14 – this note provides more information on the council's financial performance for 2013/14. Crucially, it provides figures in the format that councillors and senior officers use when making decisions about the running of the council (none of the four main statements described above are ever used in decision-making because of the numerous discrepancies between them and the legal accounts).

Note 2 – Income and Expenditure – this note provides information about a number of specific areas of income and expenditure required by law or by the Code.

1. INTRODUCTION TO THE STATEMENTS

- Note 3 Current Receivables and Payables this note summarises how much money was owed to the council at the end of the year, and how much the council owed other people.
- Note 4 Provisions and Contingent Liabilities this note provides information about things for which the council knows it will or may have to pay money to other people, but there is uncertainty about one or more elements of that payment. This may be the amount of the payment, when it has to be paid, or even whether the council will actually have to make a payment.
- Note 5 Non-Current Assets this note provides information about the council's non-current assets, which are assets that it uses for more than one year.
- Note 6 Employee Pensions this note provides information about employee pensions, including the net pensions liability (the difference between current pension commitments and the assets available to fund those) at the end of the year.
- Note 7 Financial Instruments this note provides information about the council's financial instruments, which are assets or liabilities entered into under contracts.
- Note 8 Members of the Wolverhampton City Council Group and Other Related Parties the council has relationships with a number of other organisations that readers should be aware of when reading the accounts, and this note provides information about those relationships.
- Note 9 Trust Funds this note provides information about the trust funds that the council manages on behalf of other people.
- Note 10 Reconciliation of the Financial Statements to the Statutory Accounts as mentioned earlier, there are many differences between the financial statements and the legal accounts that the council actually uses to manage its finances. This detailed note analyses all of those differences for interested readers.
- Note 11 Accounting Policies this note describes the policies that have been used by the council to prepare these statements, changes in those since last year, and any significant judgements about applying the policies that had to be made when preparing the statements.

1. INTRODUCTION TO THE STATEMENTS

Section 7 provides a set of financial statements and associated notes relating to the Housing Revenue Account. By law, the council has to account for its council housing service separately from other services, to ensure that rents only pay for housing (and likewise, that council tax does not subsidise housing).

Section 8 contains statements for the Collection Fund. These show how much council tax was raised in Wolverhampton during the year, and how it was allocated between the council, fire and police authorities.

Section 9 provides the financial statements of West Midlands Pension Fund. These are completely separate from the council's accounts, but because the council is the administering body it has to include the Pension Fund's accounts alongside its own. They follow a similar format to the council's accounts, with two main statements followed by a series of notes.

Section 10 is the council's Annual Governance Statement. This provides important information about how the council is run, and in particular how it manages key risks. Finally, there is a glossary at Section 11, which describes many of the technical accounting terms and abbreviations used in these statements.

Note on Group Accounts

Because the council owns another organisation (Wolverhampton Homes Limited), it has to produce group accounts. These combine the accounts of the two organisations and show them as if they were one. Throughout the financial statements (Section 6), numbers in italics relate to the group, while non-italic numbers relate to the council only. Usually, these are combined in the same table, but occasionally, owing to space they are shown in completely separate tables. Where there is only one figure given, this means that the figure is the same for the group and the council.

Outturn 2013/14

General Fund

2013/14 continued to be another challenging year for the council's General Fund, with a savings target of £16.5 million included in the approved budget. Furthermore, the use of General Fund reserves totalling £3.7 million was planned for 2013/14. Taking this into account, the council's outturn position for the year was a £2.3 million budget deficit. £4.6 million was attributable to the cost of redundancy. The following table analyses the outturn for 2013/14, compared to budget, by directorate. It should be noted that the figures used in this table include certain accounting charges, such as impairment of non-current assets, which are reversed under statutory provisions and are therefore not budgeted for. These can give rise to significant variations for this reason alone.

Service	2013/14 Net Budget	2013/14 Net Outturn	Total Variation Over/(Under)
	£m	£m	£m
Community	157.6	159.5	1.9
Delivery	40.2	32.7	(7.5)
Education and Enterprise	25.8	33.4	7.6
Office of the Chief Executive	2.0	1.8	(0.2)
Corporate Budgets	30.0	31.1	1.1
Net Budget Requirement	255.6	258.5	2.9
Funding:			
Government Grant (General)	(178.4)	(179.0)	(0.6)
Council Tax	(73.3)	(73.3)	-
Collection Fund Surplus	(0.2)	(0.2)	1
Budgeted Use of Reserves	(3.7)	(3.7)	-
2013/14 Deficit - Use of Reserves	-	(2.3)	(2.3)
Total Funding	(255.6)	(258.5)	(2.9)
Balance	-	-	-

Housing Revenue Account (HRA)

The outturn position for the year was an operating surplus of £14.7 million, compared to a budgeted surplus of £11.6 million. Of this £11.6 million, £11.4 million had originally been earmarked for debt redemption. However based on the outturn the council was able to use £14.5 million on making a provision for the redemption of debt. This has the major advantage of creating additional 'headroom' under the HRA borrowing limit as set by law, which will enable the council to pay for additional investment in its houses in the future.

The operating surplus over the budgeted level of surplus was generated primarily by savings on interest payable and receivable. This was due to the council's treasury management policy of using internally-generated cash balances ahead of external borrowing wherever possible.

П		Budget 2013/14 £m	Outturn 2013/14 £m	Variance Over/(Under) £m
Page	Income	(95.8)	(95.6)	0.2
Эе	Expenditure	68.5	67.6	(0.9)
52	Net Cost of Services	(27.3)	(28.0)	(0.7)
	Net Cost of Borrowing and Investments Surplus for the Year	15.7 (11.6)	13.3 (14.7)	(2.4) (3.1)
				·
	Allocation of Surplus for the Year			
	Provision for Redemption of Debt	11.4	14.5	3.1
	Transfer to/(from) Reserves	0.2	0.2	-
	Total	11.6	14.7	(3.1)

Capital Programme

Capital expenditure by the council during 2013/14 totalled £160.6 million, as set out in the following table. This was £56.6 million under budget primarily due to slippage into future years and cost reductions.

Expenditure	Approved Budget	Outturn	Variation Over/(Under)
	£m	£m	£m
General Fund			
Community	7.1	4.5	(2.6)
Delivery	10.4	8.3	(2.1)
Education and Enterprise	117.7	102.5	(15.2)
Private Sector Housing	6.8	2.2	(4.6)
Total General Fund	142.0	117.5	(24.5)
ນັ້			
Housing Revenue Account	75.2	43.1	(32.1)
ာ ယ			
Total Expenditure	217.2	160.6	(56.6)

Items of Interest in the Accounts

This section discusses some of the key items of interest in this years' statement of accounts.

Provisions and Contingent Liabilities

The council's total level of provisions decreased by £6.8 million (net) over the course of the year. This was almost entirely due to the use of £7.8 million of the Capitalisation Risks provision. Total provisions at 31 March 2014 stood at £23.8 million: further details are provided at Note 4A to the Financial Statements.

Capital Expenditure

The council once again successfully managed a large capital expenditure programme in 2013/14, resulting in additions to non-current assets of £93.1 million, along with other capital expenditure of £67.5 million. The main additions were on council dwellings (£43.1 million), mostly due to spend under the council's Decent Homes programme, and other land and buildings (£31.2 million), which reflects a number of major investments including Building Schools for the Future and the i54 development. There is more information on capital expenditure for the year in Note 1 to the Financial Statements, while information about non-current assets held by the council can be found in Note 5.

Net Pensions Liability

In accordance with the Code, the council has adopted the amendments to IAS 19 Employee Benefits. The 2012/13 figures have been restated, however there is no impact on the net pension liability or the Pensions Reserve.

The amended requirements relate to the new classes of components of defined benefit cost:

- a) Interest cost and expected return on assets are replaced with the net interest cost
- b) Administration expenses are categorised separately
- c) Actuarial gains and losses on assets are replaced with re-measurements (assets)
- d) Actuarial gains and losses on liabilities are replaced with re-measurements (liabilities)

The council's net pensions liability shows the extent to which its existing pension commitments to employees and former employees exceed the assets currently available to meet those commitments. This liability decreased by £89.8 million during 2013/14, made up of a reduction of £74.1 million in liabilities, and a growth of £15.8 million in assets. The main reasons for the net movement were gains of £109.6 million resulting from changes in actuarial assumptions, net interest payable of £22.1 million, and other net expenditure of £2.3 million. Note 6 to the Financial Statements provides further information on employee pensions.

In practice, the value of the net pensions liability is not entirely meaningful, because pensions payments will generally not need to be made for many years, and the Pension Fund plans over long timescales as a result. Furthermore, the amount the council has to charge to its revenue accounts is the amount of employee contributions payable for the year, and not the costs calculated under the accounting rules. It is also important to note that the calculation of the net pensions liability relies on a number of complex judgements and assumptions, variations in which can lead to significant differences in the outcome: this is discussed in Note 11D to the Financial Statements.

Borrowing Facilities and Capital Borrowing

The council borrows to part-fund its capital expenditure programme. As a local authority, the council can borrow funds from the Public Works Loan Board (UK Debt Management Office), which allows the council to benefit from the relatively low cost of Government borrowing. At 31 March 2014, its total borrowing portfolio stood at £563.2 million, of which £399.4 million was owed to the Public Works Loan Board, £103.8 million to private sector lenders and £60.0 million to other local authorities. The council's borrowing activities are governed by the Prudential Code for Capital Finance in Local Authorities (CIPFA).

Revaluation of Property, Plant and Equipment

The council values its property, plant and equipment on a rolling five-year cycle. During 2013/14, assets with a combined value of £101.1 million (post 2013/14 depreciation) became due for revaluation. The impact of this exercise were an upwards revaluation of £18.7 million, a downwards revaluation of £15.2 million and a net impairment of £18.6 million. An additional £3.2 million of impairments arose during the course of the year and was not part of the revaluation exercise. Statutory provisions require the council to transfer an equivalent amount from its Capital Adjustment Account to its revenue accounts, meaning that there is no impact of this impairment on council tax or housing rents.

The Medium Term Financial Strategy 2014/15 to 2018/19

General Fund

The council's General Fund Medium Term Financial Strategy (MTFS) has been prepared in an environment of change and uncertainty that is unprecedented in recent years. A number of factors have combined to create a very challenging financial situation, which is expected to continue for the foreseeable future.

Economic Conditions

The UK economy, along with the US and other major EU economies, has generally been performing weakly since the 'credit crunch' crisis of 2007/08, following several years of consistently high economic performance since the mid-1990s. Price inflation in the UK has also generally been high during the last few years. The main impacts of these economic conditions on the council have included:

Page 5

- A reduction in spending power;
- Lower borrowing costs, as a result of UK Government debt becoming more attractive to investors, although this has to be considered against the significant reduction in return on investments that has resulted;
- A significant reduction in income;
- An increase in demand for services.

There continues to be uncertainty about future economic conditions which serves to make medium term financial planning even more challenging for the council.

Social and Demographic Factors

The city of Wolverhampton is amongst the most densely populated local authority areas in England with 249,470 people living in its 26.8 square miles. In addition the latest Indices of Deprivation (2010) indicate that Wolverhampton is more deprived than it was three years before (2007), a decline from the 28th most deprived to the 20th (out of 326 councils). Although it is important to note that deprivation in the city continues to be concentrated in a number of 'hot spots'.

In addition the city's demographic profile is changing, attracting new residents and increasing diversity, and as a result Wolverhampton's population is projected to increase, by about 10,300 (4.1%) between 2011 and 2021. This growth rate is below the national, regional and Black Country averages,

which therefore suggests that if population remains a dominant factor for the distribution of Government grant then Wolverhampton will continue to receive a declining share of those resources.

The projected increase in the population and in particular the number of younger and older people is likely to mean that services relating to supporting families and individuals will experience increased demand and therefore cost.

Other significant local factors include relatively high levels of unemployment and the depressed state of the local housing market. Both of which increase demand for council services and also the need for further investment in the city.

The Medium Term Financial Strategy

Whilst the council's financial planning process is driven by the annual statutory budget cycle, its horizons extend to the next five financial years. The Medium Term Financial Strategy is a critical part of the council's planning and performance framework, and is kept under continuous review. The Medium Term Financial Strategy, as approved by Full Council in March 2014, is summarised in the table below.

Page		2014/15 £m	2015/16 £m	2016/17 £m	2017/18 £m	2018/19 £m
57	Net Expenditure Budget	584.5	573.5	579.5	593.2	593.7
	Specific Grants	(335.1)	(332.3)	(332.2)	(332.1)	(332.1)
	Net Budget	249.4	241.2	247.3	261.1	261.6
	General Funding	(237.6)	(220.0)	(214.4)	(210.6)	(202.4)
	Projected Deficit	11.8	21.2	32.9	50.5	59.2

In order to balance the 2014/15 General Fund budget, the budgeted use of £11.8 million of general balances was approved by Full Council.

As is shown in the table above, the council forecasts that it will need to save a further £59.2 million over the next five years, in addition to £72.6 million of savings that are already planned and built into the Medium Term Financial Strategy. Further to this, the council has already identified savings in

excess of £100 million over the last four financial years. The extent of the financial challenge over the medium term is the most significant the council has ever faced.

Current projections indicate that the council's general balances could be exhausted by the beginning of 2015/16. Urgent action is therefore required and the following budget strategy to identify further savings has been approved by Cabinet:

- A minimum of £25 million of additional savings for 2015/16 should be identified and reported to Cabinet in June 2014, in order to demonstrate that a balanced budget can be achieved and that general reserves can be partially replenished.
- That additional savings for 2014/15 should be identified and reported to Cabinet in June 2014 and then Council in July 2014 in order to revise the 2014/15 budget and reduce the call on general reserves.
- That a further £35 million of additional savings should be identified, taking the total additional savings to be identified to £60 million, in order to address the projected budget deficit over the medium term to 2018/19.

Significant work has taken place during the early part of 2014/15 which has resulted in the identification of savings proposals amounting to £18.1 million for 2015/16; reported to Cabinet in June 2014. Of those savings, £1.7 million can be achieved during 2014/15. Further savings proposals, as yet unquantified will go some way towards achieving the remaining £7 million for 2015/16. The table below identifies the impact of identified savings proposals on the Medium Term Financial Strategy.

	2014/15 £m	2015/16 £m	2016/17 £m	2017/18 £m	2018/19 £m
Projected Remaining Budget Deficit – as reported February 2014	11.8	21.3	32.9	50.5	59.2
Savings Proposals Identified	(1.7)	(16.4)	0.7	0.4	0.1
Projected Remaining Budget Deficit – as reported June 2014	10.1	3.2	15.5	33.5	42.3

Housing Revenue Account

A number of significant developments are planned for the Housing Revenue Account (HRA) in 2014/15. The council is planning to utilise the new freedoms and resources resulting from the introduction of self-financing in April 2012 to develop new affordable housing in the city.

An updated HRA business plan was approved in January 2014, which reflected the council's plans to invest £1.8 billion in its houses over the next 30 years, and demonstrated that under current forecasts there is sufficient funding for this. In terms of 2014/15, the plan included an average rent increase of 6.23%, and a freeze in management and maintenance allowances for managing agents. The table below shows the approved budget for 2014/15, along with forecasts for the next two years.

		Budget 2014/15	Forecast 2015/16	Forecast 2016/17
		£m	£m	£m
	Income			
_	Gross Rents - Dwellings	(91.5)	(93.8)	(96.3)
Pa	Gross Rents - Non Dwellings	(1.6)	(1.7)	(1.7)
age	Charges to Tenants for Services and Facilities	(5.2)	(5.3)	(5.4)
59	Total Income	(98.3)	(100.8)	(103.4)
-	Expenditure			
	Management and Maintenance	47.6	47.8	48.0
	Depreciation of Long Term Assets	33.8	36.1	38.8
	Net Financing Costs	16.9	16.9	16.6
	Total Expenditure	98.3	100.8	103.4
ŀ	Balance	-	-	-

Capital Programme

Capital expenditure is investment in the council's property, plant, equipment and other long-life assets. It can also include investment in assets owned by other people, in certain circumstances. Capital funding has declined significantly at a national level, but nonetheless the council has been able to put together a capital programme that includes major projects such as Building Schools for the Future, Decent Homes, Physical Regeneration and the i54 business park. The table below shows the council's capital programme for the next five years, as approved by the council in March 2014.

	2014/15	2015/16	2016/17	2017/18	2018/19	TOTAL
	£m	£m	£m	£m	£m	£m
Forecast Expenditure	168.7	67.4	51.5	43.1	-	330.7

The following table lists some of the main projects in 2014/15:

Project Delivery Delivery	Forecast Expenditure 2014/15 £m
Delivery	
	10.7
ICT Developments	7.8
Fleet Services	3.4
Energy Efficiency Measures	2.7
FutureWorks	0.9
Markets Investment	0.3
	25.8
Community	
Sports Investment Strategy	2.5
Community Hubs	2.1
Parks Refurbishment Programme	0.8
Review of Children's Homes	0.8
Early Education for Two Year Olds	0.5
Social Inclusion	0.4
Electronic Social Care Records	0.3
	7.4

Project	Forecast Expenditure 2014/15 £m
Education and Enterprise	
Schools Modernisation, Suitability and Condition	18.7
Physical Regeneration	16.0
Building Schools for the Future	10.8
i54 Access & Infrastructure	9.3
Highways - Network Development Programme	6.0
Remediation of Contaminated Land	3.2
Highways - West Midlands Major Schemes	2.8
Highways - Structural Maintenance	2.8
Wolverhampton City Centre Interchange	1.6
Targeted Disposals Programme	1.3
Corporate Asset Management Initiatives	0.7
Property Management - Other Projects	0.4
Other Projects	0.1
	73.7
Housing Private Sector	3.9
Housing Revenue Account	
Decent Homes Stock Condition	39.3
Major Stock Condition Improvements	8.8
Other Stock Condition Improvements	7.0
Other Improvements to the Public Realm	1.3
Adaptations for People with Disabilities	1.1
Service Enhancements and Miscellaneous	0.4
	57.9
Grand Total	168.7
Orana rotar	100.7

Finally, the following table shows how the council is planning to fund the projects listed above:

Source of Fun	Forecast Expenditure ding 2014/15 £m
Borrowing	102.7
Grants and Contributions	40.3
Reserve Funds	22.6
Capital Receipts	3.1
Total	168.7

4. STATEMENT OF RESPONSIBILITIES

The Council's Responsibilities

The council is required to:

- (i) Make arrangements for the proper administration of its financial affairs and to secure that one of its officers has the responsibility for the administration of those affairs. In this council, that officer is the Assistant Director Finance.
- (ii) Manage its affairs to secure economic, efficient and effective use of resources and safeguard its assets.
- (iii) Approve the Statement of Accounts.

The Assistant Director Finance's Responsibilities

The Assistant Director Finance is responsible for the preparation of the council's Statement of Accounts in accordance with proper practices as set out in the CIPFA/LASAAC Code of Practice on Local Authority Accounting in the United Kingdom (the Code).

In preparing this Statement of Accounts, the Assistant Director Finance has:

- (i) Selected suitable accounting policies and then applied them consistently.
- (ii) Made judgements and estimates that were reasonable and prudent.
- (iii) Complied with the Code.

The Assistant Director Finance has also:

- (i) Kept proper accounting records which were up to date.
- (ii) Taken reasonable steps for the prevention and detection of fraud and other irregularities.

4. STATEMENT OF RESPONSIBILITIES

Certification of the Assistant Director Finance

I certify that the above responsibilities have been complied with and the Statement of Accounts herewith presents a true and fair view of the financial position of the council as at 31 March 2014 and its income and expenditure for the year ended the same date.

Mark Taylor

Assistant Director Finance

30 June 2014

Page 64

5. INDEPENDENT AUDITORS' REPORT TO THE COUNCILLORS OF WOLVERHAMPTON CITY COUNCIL

Independent auditors' report to the Councillors of Wolverhampton City Council

To follow.

6. THE FINANCIAL STATEMENTS

Comprehensive Income and Expenditure Statement (Council only)

		2012/13					2013/14	
	Gross Expenditure	Gross Income	Net Expenditure		Note	Gross Expenditure	Gross Income	Net Expenditure
	£m	£m	£m			£m	£m	£m
	126.5	(57.0)	69.5	Adult Social Care		119.1	(29.5)	89.6
	21.6	(4.0)	17.6	Central Services to the Public		26.3	(6.4)	19.9
	401.2	(227.7)	173.5	Education and Children's Services		304.7	(218.7)	86.0
	4.7	-	4.7	Corporate and Democratic Core		4.4	-	4.4
	30.8	(11.6)	19.2	Cultural and Related Services		42.0	(10.8)	31.2
	27.1	(8.2)	18.9	Environment and Regulatory Services		29.9	(9.4)	20.5
	9.6	(3.0)	6.6	Planning Services		11.5	(3.0)	8.5
U	31.7	(4.7)	27.0	Highways and Transport Services		33.0	(4.5)	28.5
	83.1	(94.0)	(10.9)	Housing Services		83.6	(99.9)	(16.3)
ODC O	166.0	(167.6)	(1.6)	Non-Distributed Costs		144.3	(143.2)	1.1
99	902.3	(577.8)	324.5	Total Continuing Operations Excluding Acquired Operations		798.8	(525.4)	273.4
				Services Acquired				
	-	-	-	Public Health		16.4	(19.0)	(2.6)
	902.3	(577.8)	324.5	Net Cost of Services		815.2	(544.4)	270.8
	13.4	-	13.4	Levies		13.3	-	13.3
	1.5	-	1.5	Payments to the Housing Capital Receipts Pool		1.8	-	1.8
	36.9	(9.3)	27.6	(Gains)/Losses on the Disposal of Non-current Assets		74.4	(12.5)	61.9

6. THE FINANCIAL STATEMENTS

Comprehensive Income and Expenditure Statement (Council only) (Continued)

		2012/13					2013/14	
	Gross Expenditure	Gross Income	Net Expenditure		Note	Gross Expenditure	Gross Income	Net Expenditure
	£m	£m	£m			£m	£m	£m
ļ	2.1	(0.8)		External Trading Organisations		1.2	(1.3)	(0.1)
	25.4	-		Interest Payable		31.3	-	31.3
	20.8	-		Net Interest Expense		22.1	-	22.1
	-	(0.4)	(0.4)	Interest Receivable		-	(0.4)	(0.4)
	-	-	-	Income and Expenditure in Relation to Investment Properties and Changes in their Fair Value		(0.2)	-	(0.2)
],	-	(0.6)	(0.6)	Other Investment Income		-	(4.1)	(4.1)
7	(0.1)	(93.7)	(93.8)	Council Tax		-	(74.0)	(74.0)
2	-	(134.2)	(134.2)	National Non-domestic Rates		-	(34.8)	(34.8)
5	-	(7.8)	(7.8)	Unringfenced Revenue Grants Receivable		-	(143.5)	(143.5)
ĺ	-	(79.5)	(79.5)	Capital Grants Receivable		-	(47.4)	(47.4)
	1,002.3	(904.1)	98.2	Deficit on the Provision of Services		959.1	(862.4)	96.7
ŀ	35.1	-	35.1	(Surplus) or Deficit on Revaluation of Non- current Assets		(6.8)	-	(6.8)
	1	-	-	(Surplus) or Deficit on Revaluation of Available- for-sale Financial Assets		-	-	-
	94.0	-	94.0	Re-measurements		-	(109.6)	(109.6)
	1,131.4	(904.1)	227.3	Total Comprehensive Income and Expenditure		952.3	(972.0)	(19.7)

6. THE FINANCIAL STATEMENTS

Comprehensive Income and Expenditure Statement (Group)

	2012/13					2013/14	
Gross Expenditure	Gross Income	Net Expenditure		Note	Gross Expenditure	Gross Income	Net Expenditure
£m	£m	£m			£m	£m	£m
126.5	(57.0)	69.5	Adult Social Care		119.1	(29.5)	89.6
21.6	(4.0)	17.6	Central Services to the Public		26.3	(6.4)	19.9
401.2	(227.7)	173.5	Education and Children's Services		304.7	(218.7)	86.0
4.7	-	4.7	Corporate and Democratic Core		4.4	-	4.4
30.8	(11.6)	19.2	Cultural and Related Services		42.0	(10.8)	31.2
27.1	(8.2)	18.9	Environment and Regulatory Services		29.9	(9.4)	20.5
9.6	(3.0)	6.6	Planning Services		11.5	(3.0)	8.5
31.7	(4.4)	27.3	Highways and Transport Services		33.0	(4.1)	28.9
81.0	(92.6)	(11.6)	Housing Services		81.5	(100.0)	(18.5)
166.2	(167.6)	(1.4)	Non-Distributed Costs		144.3	(143.2)	1.1
900.4	(576.1)	324.3	Total Continuing Operations excluding Acquired Operations		796.7	(525.1)	271.6
			Services Aquired				
-	-	-	Public Health		16.4	(19.0)	(2.6)
900.4	(576.1)	324.3	Net Cost of Services		813.1	(544.1)	269.0
13.4	-	13.4	Levies		13.3	-	13.3
1.5	-	1.5	Payments to the Housing Capital Receipts Pool		1.8	-	1.8
36.9	(9.3)	27.6	(Gains)/Losses on the Disposal of Non-Current Assets		74.4	(12.5)	61.9

23

Comprehensive Income and Expenditure Statement (Group) (Continued)

	2012/13					2013/14	
Gross Expenditure £m	Gross Income £m	Net Expenditure £m		Note	Gross Expenditure £m	Gross Income £m	Net Expenditure £m
2.1	0.9		External Trading Organisations		1.2	1.3	2.5
25.4	- 0.5	25.4	Interest Payable		31.3	7.5	31.3
21.7	_		Net Interest Expense		23.3	_	23.3
-	(0.5)		Interest Receivable		-	(0.4)	(0.4)
-	-	, ,	Income and Expenditure in Relation to Investment Properties and Changes in their Fair Value		(0.2)	, ,	(0.2)
-	(0.6)	(0.6)	Other Investment Income		-	(4.1)	(4.1)
(0.1)	(93.7)	(93.8)	Council Tax		-	(74.0)	(74.0)
-	(134.2)	(134.2)	National Non-domestic Rates		-	(34.8)	(34.8)
-	(7.8)	(7.8)	Unringfenced Revenue Grants Receivable		-	(143.5)	(143.5)
-	(79.5)	(79.5)	Capital Grants Receivable		-	(47.4)	(47.4)
1,001.3	(900.8)	100.5	Deficit on the Provision of Services		958.2	(859.5)	98.7
35.1	-	35.1	(Surplus) or Deficit on Revaluation of Non- current Assets		(6.8)	-	(6.8)
-	-	-	(Surplus) or Deficit on Revaluation of Available- for-sale Financial Assets		-	-	-
101.9	-	101.9	Re-measurements		-	(126.1)	(126.1)
1,138.3	(900.8)	237.5	Total Comprehensive Income and Expenditure		951.4	(985.6)	(34.2)

Balance Sheets

31 March	า 2013			31 Marcl	า 2014
Council £m	Group £m		Note	Council £m	Group £m
1,533.6	1,533.6	Property, Plant and Equipment	5	1,469.3	1,469.3
14.3	14.3	Investment Property	5	16.3	16.3
1.7	1.7	Intangible Assets	5	4.5	4.5
11.5	11.5	Heritage Assets	5	11.5	11.5
18.6	18.6	Non-current Investments		20.3	20.3
1.5	1.5	Non-current Debtors		1.4	1.4
1,581.2	1,581.2	Total Non-current Assets		1,523.3	1,523.3
4.5	4.5	Current Investments		12.5	12.5
0.6	0.6	Inventories		0.6	0.6
69.6	69.0	Current Debtors	3	77.1	75.9
3.5	12.1	Cash and Cash Equivalents		4.1	15.2
. 78.2		Total Current Assets		94.3	104.2
3					
(43.8)	, <u> </u>	Current Borrowing		(65.3)	(65.3)
(76.6)		Current Creditors	3	(71.4)	(75.0)
(30.6)		Current Provisions	4A	(25.5)	(25.5)
(151.0)	(153.5)	Total Current Liabilities		(162.2)	(165.8)
(492.4)	(492.4)	Non-current Borrowing		(515.7)	(515.7)
(551.7)	(580.0)	Net Pension Liability	6	(461.9)	(476.4)
(9.9)	(9.9)	Capital Grants Received in Advance		(6.6)	(6.6)
(86.0)	(86.0)	Other Non-current Liabilities		(83.2)	(83.2)
(1,140.0)	(1,168.3)	Total Non-current Liabilities		(1,067.4)	(1,081.9)
368.4	345 6	Net Assets		388.0	379.8

age /(

Balance Sheets (Continued)

Bail	31 Marc	ch 2013			31 Marc	h 2014
	Council	Group		Note	Council	Group
	£m	£m			£m	£m
	(15.9)	(15.9)	General Fund Balance	10B, 10C	(27.0)	(27.0)
	(61.1)	(61.1)	General Fund Earmarked Reserves	10B, 10C	(43.6)	(43.6)
	(5.0)	(5.0)	Housing Revenue Account Balance	10B, 10C	(5.0)	(5.0)
	(9.7)	(9.7)	Major Repairs Reserve	10B, 10C	(13.2)	(13.2)
	(5.5)	(5.5)	Capital Receipts Reserve	10B, 10C	(9.6)	(9.6)
	(31.6)	(31.6)	Capital Grants Unapplied Account	10B, 10C	(34.1)	(34.1)
	-	22.8	Reserves of Subsidiary	10B, 10C	-	8.2
<u>۔</u>	(128.8)	(106.0)	Total Usable Reserves		(132.5)	(124.3)
<u> </u>						
5. L	6.9	6.9	Short-term Accumulating Compensated Absences Account	10B, 10C	5.9	5.9
<u> </u>	(12.1)	(12.1)	Available-for-sale Financial Instruments Reserve	10B, 10C	(12.1)	(12.1)
	(575.4)	(575.4)	Capital Adjustment Account	10B, 10C	(537.3)	(537.3)
	0.5	0.5	Collection Fund Adjustment Account	10B, 10C	0.5	0.5
	3.2	3.2	Financial Instruments Adjustment Account	10B, 10C	3.5	3.5
	551.7	551.7	Pensions Reserve	10B, 10C	461.9	461.9
	(214.4)	(214.4)	Revaluation Reserve	10B, 10C	(177.9)	(177.9)
	(239.6)	(239.6)	Total Unusable Reserves		(255.5)	(255.5)
	(368.4)	(345.6)	Total Reserves		(388.0)	(379.8)

The notes on pages 31 to 117 form part of the financial statements.

Movement in Reserves Statement - 2013/14

(For a detailed breakdown of the figures in this Statement, see Note 10B)

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves	Unusable Reserves	TOTAL (Council)	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	(15.9)	(61.1)	(5.0)	(9.7)	(5.5)	(31.6)	(128.8)	(239.6)	(368.4)	22.8	(345.6)
Surplus or Deficit on Provision of Services	112.1	-	(15.4)	-	-	-	96.7	-	96.7	2.0	98.7
Other Comprehensive Income and Expenditure	-	-	-	-	-	-	-	(116.4)	(116.4)	(16.5)	(132.9)
Total Comprehensive Income and Expenditure	112.1	-	(15.4)	-	-	-	96.7	(116.4)	(19.7)	(14.5)	(34.2)
Adjustments between Accounting Basis & Funding Basis under Regulations	(105.9)	-	15.6	(3.5)	(4.1)	(2.5)	(100.4)	100.5	0.1	-	0.1
Net Decrease/(Increase) before Transfers & Other Movements	6.2	-	0.2	(3.5)	(4.1)	(2.5)	(3.7)	(15.9)	(19.6)	(14.5)	(34.1)
Transfers to/from other Reserves	(17.3)	17.5	(0.2)	-	-	-	-	-	-	-	-
Decrease/(Increase) for the Year	(11.1)	17.5	0.0	(3.5)	(4.1)	(2.5)	(3.7)	(15.9)	(19.6)	(14.5)	(34.1)
Balance Carried Forward	(27.0)	(43.6)	(5.0)	(13.2)	(9.6)	(34.1)	(132.5)	(255.5)	(388.0)	8.3	(379.7)

Movement in Reserves Statement – 2012/13

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves	Unusable Reserves	TOTAL (Council)	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	(19.5)	(83.6)	(10.5)	(5.2)	(2.3)	(36.1)	(157.2)	(438.5)	(595.7)	12.6	(583.1)
Surplus or Deficit on Provision of Services	99.6	-	(9.3)	-	-	-	90.3	-	90.3	1.4	91.7
Other Comprehensive Income and Expenditure	-	-	-	-	-	-	-	137.0	137.0	8.8	145.8
Total Comprehensive Income and Expenditure	99.6	-	(9.3)	-	-	-	90.3	137.0	227.3	10.2	237.5
Adjustments between Accounting Basis & Funding Basis under Regulations	(73.7)	-	15.0	(4.5)	(3.2)	4.5	(61.9)	61.9	-	-	•
Net Increase/Decrease before Transfers & Other Movements	25.9	-	5.7	(4.5)	(3.2)	4.5	28.4	198.9	227.3	10.2	237.5
Transfers to/from other Reserves	(22.3)	22.5	(0.2)	-	-	-	-	-	-	-	-
Decrease/(Increase) for the Year	3.6	22.5	5.5	(4.5)	(3.2)	4.5	28.4	198.9	227.3	10.2	237.5
Balance Carried Forward	(15.9)	(61.1)	(5.0)	(9.7)	(5.5)	(31.6)	(128.8)	(239.6)	(368.4)	22.8	(345.6)

rage /

Cash Flow Statement

2012	/13		2013	/14
Council	Group		Council	Group
£m	£m		£m	£m
98.2	100.5	Net (surplus) or deficit on the provision of services	96.7	98.7
(148.3)		Adjust for non-cash movements	(161.4)	(164.7)
9.3	9.3	Adjust for items that are investing and financing activities	12.5	12.5
(40.8)	(36.8)	Net cash flows from operating activities	(52.2)	(53.5)
		Comprising:		
25.4	25.2	Interest paid	31.3	30.2
(0.4)	(0.4)	Interest received	(0.4)	(0.4)
(0.6)	<u> </u>	Dividends received	(4.1)	(4.1)
(65.2)		Other operating activities	(79.0)	(80.3)
(40.8)	(37.0)	Net cash flows from operating activities	(52.2)	(54.6)
		Investing activities		
147.2	147.2	Purchase of property, plant and equipment, investment property and intangible assets	93.1	93.1
571.2	571.2	Purchase of short-term and long-term investments	584.6	584.6
(9.3)	(9.3)	Proceeds from the sale of property, plant and equipment, investment property and intangible assets	(12.5)	(12.5)
(576.2)	(3/0.2)	Other receipts from investing activities	(571.7)	(571.7)
132.9	132.9	Net cash flows from investing activities	93.5	93.5

Cash Flow Statement (continued)

2012	2/13		2013	3/14
Council	Group		Council	Group
£m	£m		£m	£m
		Financing activities		
(171.5)	(171.5)	Cash receipts of short- and long-term borrowing	(164.5)	(164.5)
1.0	1.0	Cash payments for the reduction of the outstanding liability relating to finance leases and on- balance sheet PFI contracts	1.7	1.7
85.4	85.4	Repayments of short-and long-term borrowing	120.8	120.8
(85.1)	(85.1)	Net cash flows from total financing activities	(42.0)	(42.0)
7.0	10.8	Net (increase) or decrease in cash and cash equivalents	(0.7)	(3.1)
J		Cash and cash equivalents at the start of the year:		
0.2	0.2	- Cash held by the council	0.2	0.2
2.4	13.9	- Bank current accounts	3.3	11.9
2.6	14.1		3.5	12.1
		Cash and cash equivalents at the end of the year:		
0.2	0.2	- Cash held by the council	0.2	0.2
3.3	11.9	- Bank current accounts	3.9	15.0
3.5	12.1		4.1	15.2

Note 1 – Financial Performance for 2013/14

The purpose of this note is to show how the council's financial performance for 2013/14 was reported to its management (senior officers and councillors).

General Fund

The following table compares the council's General Fund outturn for 2013/14 to its budget. It analyses spend by directorate, which is the format used for internal reporting to management. This table is calculated in line with the legal requirements. As the table shows, there was a £2.3 million deficit on the council's net General Fund expenditure for the year. £4.6 million was attributable to the cost of redundancy. After taking into account net transfers to/from earmarked reserves, there was a reduction in the General Fund Balance of £6 million, with remaining funds totalling £27 million at the end of the financial year.

Service	2013/14 Net Budget	2013/14 Net Outturn	Total Variation Over/(Under)
Community	£m 157.6	£m 159.5	£m 1.9
N Delivery	40.2	32.7	(7.5)
Education and Enterprise	25.8	33.4	7.6
Office of the Chief Executive	2.0	1.8	(0.2)
Corporate Budgets	30.0	31.1	1.1
Net Budget Requirement	255.6	258.5	2.9
Funding:			
Government Grant (General)	(178.4)	(179.0)	(0.6)
Council Tax	(73.3)	(73.3)	-
Collection Fund Surplus	(0.2)	(0.2)	-
Budgeted Use of Reserves	(3.7)	(3.7)	-
2013/14 Deficit - Use of Reserves	-	(2.3)	(2.3)
Total Funding	(255.6)	(258.5)	(2.9)
Budget (Surplus)/Deficit	-	-	-

Housing Revenue Account (HRA)

The outturn position for the year was an operating surplus of £14.7 million, compared to a budgeted surplus of £11.6 million. The following table shows the outturn position on the HRA during the year. The table excludes any transactions which are required by the Code but are not a charge or credit to the HRA under law.

	2013/14 Approved Budget	2013/14 Outturn	Variation
	£m	£m	£m
Income			
Gross Rents - Dwellings	(88.9)	(88.9)	-
Gross Rents - Non Dwellings	(1.6)	(1.6)	-
Charges to Tenants for Services and Facilities	(5.0)	(5.0)	-
Total Income	(95.5)	(95.5)	-
Expenditure			
Repairs and Maintenance	26.2	25.9	(0.3)
Supervision and Management	18.4	18.4	-
Rents, Rates and Taxes	0.2	0.4	0.2
Increase in Provision for Bad Debts	1.5	0.9	(0.6)
Depreciation of Long Term Assets	21.9	21.9	-
Total Expenditure	68.2	67.5	(0.7)
Net Cost of HRA Services	(27.3)	(28.0)	(0.7)
Interest Payable	15.8	13.4	(2.4)
Interest and Investment Income	-	(0.1)	(0.1)
Adjustment for Premiums & Discounts	-	-	-
Surplus before Transfers to/from Reserves and Provision for	(11.6)	(14.7)	(3.1)

	2013/14 Approved Budget	2013/14 Outturn	Variation
	£m	£m	£m
Redemption of Debt			
Allocation of Surplus			
Provision for Redemption of Debt	11.4	14.5	3.1
Transfer to/(from) Reserves	0.2	0.2	
Balance for the Year	-	-	-

Capital Programme

The following tables show capital expenditure for the year, and how that expenditure was financed. As with the General Fund revenue account, this is presented according to the breakdown used internally by councillors and senior managers when making decisions about capital budgets.

Expenditure	Approved Budget	2013/14 Outturn	Variation Over/(Under)
	£m	£m	£m
General Fund			
Community	7.1	4.5	(2.6)
Delivery	10.4	8.3	(2.1)
Education and Enterprise	117.7	102.5	(15.2)
Private Sector Housing	6.8	2.2	(4.6)
	142.0	117.5	(24.5)
Housing Revenue Account	75.2	43.1	(32.1)
Total Expenditure	217.2	160.6	(56.6)

Funding	Approved Funding	Outturn Funding	Variation Over/(Under)	
	£m	£m	£m	
General Fund				
External Funding				
Grants & Contributions	91.6	87.7	(3.9)	
	91.6	87.7	(3.9)	
	64.51%	74.64%		
Council Funding				
Capital Receipts	4.1	4.8	0.7	
Prudential Borrowing	46.4	25.1	(21.3)	
Revenue Contributions	(0.1)	(0.1)	-	
	50.4	29.8	(20.6)	
	35.49%	25.36%		
Total General Fund Funding	142.0	117.5	(24.5)	
Housing Revenue Account				
External Funding				
Grants & Contributions	12.0	12.0	-	
	12.0	12.0	-	
	15.90%	27.85%		
Council Funding				
Capital Receipts	2.5	1.8	(0.7)	
Prudential Borrowing	35.0	10.9	(24.1)	

age /

Page 80

6. THE FINANCIAL STATEMENTS

Funding	Approved Funding	Outturn Funding	Variation Over/(Under)	
	£m	£m	£m	
Major Repairs Reserve	25.8	18.4	(7.4)	
	63.3	31.1	(32.2)	
	84.10%	72.15%		
Total Housing Revenue Account Funding	75.2	43.1	(32.1)	

Reserves

The table below analyses the council's usable reserves, in the format reported to the Cabinet.

	Balance at 31 March 2013 £m	Approved Transfer to General Fund Balance 23 October 2013 £m	Other Net Movements 2013/14 £m	Balance at 31 March 2014 £m
Earmarked Reserves				
Community	2.6	(0.4)	1.6	3.8
Education and Enterprise (Non-Schools)	7.5	(1.1)	(1.7)	4.7
Office of the Chief Executive and Delivery	4.9	(0.1)	0.2	5.0
Corporate	28.5	(5.5)	(8.9)	14.1
Total Earmarked Reserves	43.5	(7.1)	(8.8)	27.6
Other Reserves				
Schools' Balances	17.6	-	(1.6)	16.0
General Fund Balance	15.9	7.1	*4.0	27.0
Housing Revenue Account Balance	5.0	-	-	5.0
Total Usable Revenue Reserves	82.0	-	(6.4)	75.6

^{*}This includes the transfer of £10 million generated from an adjustment to "Factor E" which had a positive impact on the General Fund Balance as approved by Cabinet on 23 October 2013, in addition to the drawdown of £6 million to meet General Fund expenditure in 2013/14.

Note 2 - Income and Expenditure

2A - Acquired Operations

With effect from 1 April 2013, responsibility for Public Health transferred from the former Primary Care Trust (PCT) to the council. The council received a ring-fenced grant of £18.8 million for 2013/14, as well as contributions from other agencies of £237,000, and its expenditure on Public Health activities in the year amounted to £16.3 million. The balance of £2.8 million has been identified for Public Health activities in 2014/15, and was transferred to earmarked reserves at the year end.

2B - Trading Operations

	2012	/13		2013/1	4
	(Turnover)	Deficit/ (Surplus)	Operation	Turnover	Deficit/ (Surplus)
ס	£m	£m		£m	£m
ag	(2.3)	2.3	Markets	(2.3)	(0.1)
Je	(4.0)	(0.3)	Cleaning of Buildings	(3.8)	0.3
∞	(3.2)	-	Ground Maintenance	(4.6)	-
_	(3.0)	-	Street Cleaning	(2.9)	(0.3)
	(7.4)	-	Schools and Welfare Catering	(7.5)	-
	(0.3)	-	Other Catering	(0.3)	-
	(9.4)	(0.6)	Transport Services	(5.7)	-
	(0.9)	-	Former DSO Depots	(0.9)	-
	(30.5)	1.4	Total	(28.0)	(0.1)

Trading operations are incorporated into the Comprehensive Income and Expenditure Statement. Some are an integral part of one of the council's services to the public (e.g. Street Cleaning), whilst others are support services to the council's services to the public (e.g. Schools and Welfare Catering). The expenditure of these operations is allocated or recharged to headings within the Net Cost of Services. Expenditure and income attributable to the external element of trading operations are disclosed on the face of the Comprehensive Income and Expenditure Statement.

The (surplus)/deficits shown in the table above are generated from the council's external trading operations and are therefore not apportioned at year end.

2C - Pooled Budgets

The council takes part in pooled budget schemes with Wolverhampton Clinical Commissioning Group (CCG). In 2012/13, these arrangements were with the former Wolverhampton Primary Care Trust (PCT). The table below provides a summary of these schemes, with the administering body's name in brackets.

	Council Contribution £m	2012/13 PCT Contribution £m	Total Expenditure £m	Scheme	Council Contribution £m	2013/14 CGG Contribution £m	Total Expenditure £m
Pa	2.3	1.5	3.8	Child Placements with External Agencies (Council) - An integrated service to provide placements for children with social care, education and health needs	2.5	1.7	4.2
age 82	25.4	1.0	26.4	Learning Disability Services (Council) Covers the provision and purchase of residential and nursing, domiciliary and day care placements	-	-	-
	4.6	16.4	21.0	Mental Health Service (Council and CGG) - Covers the provision and purchase of residential and nursing, domiciliary and day care placements	-	1	-

2D - Councillors' Allowances

The council paid £991,000 in councillors' allowances during 2013/14 (2012/13: £989,000).

2E - Senior Officers' Remuneration

The following table sets out the remuneration disclosures for Senior Officers (with reference to notes where applicable):

Post Title / Name		Salary, Fees and Allowances	Bonuses	Expenses Allowances	Termination Benefits	Employers' Pension Contribution	West Midlands Pension Fund Responsibilities ¹	TOTAL REMUNERATION
		£	£	£	£	£	£	£
Senior Officers with a salary of £150,000 or more per year								
Chief Executive, Simon Warren ²	2013/14	153,833	-	-	-	30,217	4,370	188,420
	2012/13	140,958	-	1,989	-	30,165	17,505	190,617
Senior Officers with a salary of less than £150,000 per year								
Strategic Director Delivery ²⁸³	2013/14	130,000	-	-	-	24,830	-	154,830
	2012/13	162,063	-	497	-	6,203	-	168,763
Strategic Director Delivery ⁴	2013/14	-	-	-	-	-	-	
	2012/13	9,420	-	-	-	-	-	9,420
Strategic Director Education and Enterprise ²⁸⁵	2013/14	112,527	-	-	-	21,493	-	134,020
Strategic Director Education and Enterprise, Charles Green ⁵	2012/13	187,223	-	-	-	-	-	187,223
Strategic Director Community ²	2013/14		-	_	-	25,177	-	156,995
	2012/13	130,818	-	1,989	-	24,968	-	157,775
Director of Pensions ¹	2013/14		-	-	-	23,464	-	146,314
	2012/13	120,831		6,627	-	23,090	-	150,548

²age 84

6. THE FINANCIAL STATEMENTS

Post Title / Name		Salary, Fees and Allowances	Bonuses	Expenses Allowances	Termination Benefits	Employers' Pension Contribution	West Midlands Pension Fund Responsibilities ¹	TOTAL REMUNERATION
	_	£	£	£	£	£	£	£
Director of Public Health ⁶	2013/14	87,747		963	-	12,285		100,995
	2012/13		-	-	-	-	-	-
Chief Human Resources Officer ⁷	2013/14	66,225	-	-	-	12,649	-	78,874
	2012/13		-	-	-	10,472	-	66,175
Chief Legal Officer ⁸	2013/14	131,467	-	-	-	-	-	131,467
	2012/13	-	-	-	-	-	-	-
Assistant Chief Executive 9	2013/14	-	-	-	-	-	-	-
	2012/13	73,012	-	722	48,984	12,628	-	135,346
Assistant Director Governance 8	2013/14		-	-	-	-	-	-
	2012/13	63,503	-	599	139,200	10,974	5,307	219,583
Assistant Director Business Change 10	2013/14	89,535	-	963	-	17,101	-	107,599
	2012/13	29,854	-	321	-	6,174	2,985	39,334
Assistant Director Regeneration	2013/14		-	963	-	16,562	-	104,237
	2012/13	85,287	-	963	-	16,034	-	102,284
Assistant Director Central Services	2013/14		-	963	-	16,562	-	104,237
	2012/13	85,287	-	963	-	16,034	-	102,284
Assistant Director Finance (Section 151 Officer) 11	2013/14		-	963	-	16,290	-	102,540
	2012/13	71,427	-	963	-	13,428	-	85,818
Assistant Director Corporate Services (Section 151 Officer) 11	2013/14		-	-	-	-	-	
	2012/13	59,708	-	642	-	12,347	5,971	78,668

	Post Title / Name		Salary, Fees and Allowances	Bonuses	Expenses Allowances	Termination Benefits	Employers' Pension Contribution	West Midlands Pension Fund Responsibilities ¹	TOTAL REMUNERATION
			£	£	£	£	£	£	£
	Assistant Director Safeguarding, Business Support and Community Services ¹²	2013/14	82,317		979		15,723	_	99,019
	Community dervices	2012/13	85,287		963	-	16,034	_	102,284
	Assistant Director Investments 1813	2013/14 2012/13	50,897 -	-	575 -	-	9,721	<u>-</u>	61,193 -
	Assistant Director Health, Wellbeing and Disabilities	2013/14 2012/13	86,712 85,287	-	963 963	-	16,562 16,034	<u>-</u>	104,237 102,284
	Assistant Director Older People and Personlisation	2013/14 2012/13	86,712 85,287	-	963 963	-	16,562 16,034	- -	104,237 102,284
U	Assistant Director Partnership, Economy and Culture	2013/14 2012/13	86,712 85,287	-	963 963	-	16,562 16,034	-	104,237 102,284
age	Assistant Director Schools, Skills and Learning - Interim	2013/14	96,052	-	-	-	-	-	96,052
85	Assistant Director Schools, Skills and Learning ¹⁴	2013/14 2012/13	26,502 85,287	-	254 963	63,415 -	4,363 16,034	-	94,534 102,284
O1	Assistant Director Children, Young People and Families	2013/14 2012/13		-	963 963	-	16,562 16,034		104,237 102,284

Note 1: The costs of West Midlands Pension Fund responsibilities are funded by the West Midlands Pension Fund and not by Wolverhampton City Council.

Note 2: Travel allowance for 2013/14 of £1,989 (2012/13 £1,989) has been voluntarily waived by the Chief Executive, Strategic Director Delivery, Strategic Director Education and Enterprise and Strategic Director Community with effect from 1 April 2013. This allowance has ceased permanently at the request of each of the postholders.

Note 3: The post of Strategic Director Delivery was held on an interim basis for part of 2012/13, with the post holder taking the post on a permanent basis from 1 January 2013.

	Note 4:	The post of Strategic Director Delivery was held on an interim basis for part of 2012/13, by a different interim to the individual referred to in note 3.
	Note 5:	The post of Strategic Director Education and Enterprise was held on an interim basis throughout 2012/13 until March 2013. The figures for 2012/13 include a later payment of £13,737.50 in relation to 2011/12 which has not been reported previously. The post was filled on a permanent basis on 20 May 2013 with an annual salary of £130,000.
	Note 6:	The post of Director of Public Health was created on 1 April 2013, in readiness for the transfer of the Public Health budget and responsibilities from Central Government.
	Note 7:	The post of Chief Human Resources Officer was created on 1 April 2013. The post of Head of Human Resources was deleted.
	Note 8:	The post of Assistant Director Governance became vacant on 14 November 2012 and had an annual salary of £85,287. This post was deleted. A new post of Chief Legal officer was created on 1 April 2013.
	Note 9:	The post of Assistant Chief Executive became vacant on 31 December 2012 and had an annual salary of £89,560. This post has since been deleted.
	Note 10:	The post of Assistant Director Business Change was created on 26 November 2012.
Pag	Note 11: Note 12:	The role of Section 151 Officer was held by two individuals during 2012/13. The amounts disclosed in this note relate to the full year's salary for both of these individuals, with transfer of responsibilities having taken place in November 2012.
уе 86	Note 12:	The post of Assistant Director Safeguarding, Business Support and Community Services became vacant on 6 August 2013 and had an annual salary of £86,712. This post was filled on 7 August 2013 on a secondment basis to 30 April 2014, at an annual salary of £78,500. Due to a restructure this post is to be deleted on 1 May 2014 with management responsibilities redistributed between the remaining posts at Assistant Director level within the Community Directorate.
	Note 13:	The post of Assistant Director Investments (within the West Midlands Pension Fund) was created on 27 August 2013 with an annual salary of £85,287.
	Note 14:	The post of Assistant Director Schools Skills and Learning became vacant on 5 July 2013 and had an annual salary of £86,712. This post has since been held on an interim basis from 22 July 2013.

The following table shows the number of other employees whose remuneration for the year (excluding employers' pension contributions) exceeded £50,000.

	230,000.					
				20	13/14	
		Nu	umber of Employees	Rest of	Total Number of Employees In	Employees receiving termination packages
	Remuneration Band	Schools	Pension Fund	Council	Band	(included in total)
	£50,000 - £54,999	56	2	31	89	5
	£55,000 - £59,999	32	1	28	61	5
	£60,000 - £64,999	31	1	14	46	10
	£65,000 - £69,999	23	-	5	28	4
	£70,000 - £74,999	7	-	4	11	4
	£75,000 - £79,999	3	-	4	7	4
τ	£80,000 - £84,999	2	-	1	3	-
'age	£85,000 - £89,999	6	-	2	8	2
ЭС	£90,000 - £94,999	3	-	1	4	1
α	£95,000 - £99,999	1	-	-	1	-
	£100,000 - £104,999	1	-	-	1	-
	£105,000 - £109,999	-	-	-	-	-
	£110,000 - £114,999	-	-	-	-	-
	£115,000 - £119,999	1	-	-	1	-
	£120,000 - £124,999	-	-	-	-	-
	£125,000 - £129,999	-	-	-	-	-
	£130,000 - £134,999	-	-	-	-	-
	£135,000 - £139,999	-	-	-	-	-
	£140,000 - £144,999	-	-	1	1	1
	£145,000 - £149,999	-	-	-	-	-
	Total	166	4	91	261	36

				201	2/13	
		Nι	umber of Employees	Rest of	Total Number of Employees In	Employees receiving termination packages
	Remuneration Band	Schools	Pension Fund	Council	Band	(included in total)
	£50,000 - £54,999	64	1	43	108	4
	£55,000 - £59,999	44	3	32	79	5
	£60,000 - £64,999	30	-	5	35	3
	£65,000 - £69,999	23	-	3	26	3
	£70,000 - £74,999	11	-	5	16	5
	£75,000 - £79,999	3	-	2	5	1
_	£80,000 - £84,999	2	1	2	5	2
В	£85,000 - £89,999	5	-	1	6	1
age	£90,000 - £94,999	1	-	-	1	-
	1 + 95 (IIII) - + 99 999	2	-	-	2	-
õ	£100,000 - £104,999	-	-	-	-	-
	£105,000 - £109,999	-	-	-	-	-
	£110,000 - £114,999	-	-	-	-	-
	£115,000 - £119,999	1	-	-	1	-
	£120,000 - £124,999	-	1	-	1	1
	£125,000 - £129,999	-	-	-	-	-
	£130,000 - £134,999	-	-	-	-	-
	£135,000 - £139,999	-	-	-	-	-
	£140,000 - £144,999	-	-	-	-	-
	£145,000 - £149,999	-	-	-	-	-
	Total	186	6	93	285	25

Page 89

6. THE FINANCIAL STATEMENTS

2F - Exit Packages

The following table provides information about exit packages payable by the council during the year. This includes both schools and the Pension Fund.

Compi	2012 Compulsory Volur			Tot	tal	Value of Individual Package Compulsory Voluntary				To	tal	
No.	£m	No.	£m	No.	£m	Fackage	No.	£m	No.	£m	No.	£m
1	0.2	-	-	1	0.2	£150,000 to £200,000	-		-	-	-	-
_	-	-	-	-	-	£100,000 to £150,000	-	-	-	-	-	-
1	0.1	1	0.1	2	0.2	£80,000 to £100,000	-	-	2	0.2	2	0.2
1	0.1	2	0.1	3	0.2	£60,000 to £80,000	3	0.2	1	0.1	4	0.3
2	0.1	4	0.1	6	0.2	£40,000 to £60,000	2	0.1	8	0.4	10	0.5
10	0.2	7	0.1	17	0.3	£20,000 to £40,000	11	0.3	45	1.1	56	1.4
50	0.3	40	0.3	90	0.6	Less than £20,000	83	0.5	391	3.2	474	3.7
65	1.0	54	0.7	119	1.7	Total	99	1.1	447	5.0	546	6.1

2G - Amounts Payable to the Auditor

The table below shows amounts payable to the council's external auditors during the year.

	2012/13 £m		2013/14 £m
		Audit Commission – statutory inspections:	
	-	- Audit inspection fee	-
	(0.022)	- Rebates for arrangements relating to the abolition of the Audit Commission (*)	(0.034)
	(0.022)	Sub Total Audit Commission	(0.034)
		PricewaterhouseCoopers LLP	
	0.251	- External audit (council)	0.251
O	0.049	- External audit (West Midlands Pension Fund)	0.049
Page	0.088	- Certification of grant claims and returns	0.049
	0.115	- Additional work (**)	0.216
90	0.503	Sub Total PricewaterhouseCoopers LLP	0.565
	0.481	TOTAL	0.531

^{*} The rebates of £22,000 in 2012/13 and £34,000 in 2013/14 were intended to smooth any financial impact of the abolition of the Audit Commission on local authorities.

^{**} The fees to PricewaterhouseCoopers LLP in 2012/13 for additional work relate to £70,000 risk based work, £36,000 estates review and £9,000 HRA delivery options review. In 2013/14 the additional work relates to £34,000 risk based work, £85,000 estates review, £94,000 FutureSpace Programme Advisory Service and Business Case Support and £3,000 multi tax helpline.

2H - Dedicated Schools Grant

The council's expenditure on schools is funded primarily by grant monies provided by the Department for Education: the Dedicated Schools Grant (DSG). DSG is ring-fenced and can only be applied to meet expenditure properly included in the Schools Budget, as defined in the School Finance (England) Regulations 2011. The Schools Budget includes elements for a restricted range of educational services provided on a council-wide basis and for the Individual Schools Budget (ISB), which is divided into a budget share for each maintained school. Over and underspends on the two elements are required to be accounted for separately. The table below shows how the council applied its DSG.

	Central Expenditure	2012/13 Individual Schools Budget	Total		Central Expenditure	2013/14 Individual Schools Budget	Total		
	£m	£m	£m (189.8)	Final DSG for the year before academy recoupment	£m	£m	£m (196.7)		
			22.1	Academy figure recouped			35.1		
Page			(167.7)	Total DSG after academy recoupment for the year	, ,				
				Brought forward from previous year		-			
91			-	Carry-forward to following year agreed in advance			-		
	(10.4)	(157.3)	(167.7)	Agreed initial budgeted distribution in the year	(14.0)	(147.6)	(161.6)		
=	(10.4)	(157.3)	(167.7)	Final budgeted distribution for the year					
=	10.8		10.8	Less actual central expenditure	14.2		14.2		
		157.3	157.3	Less actual ISB deployed to schools		148.2	148.2		
	(0.4)	-	(0.4)	Plus local authority contribution	-	-	-		
	-	-	-	Overspend carried forward to following year	0.2	0.6	8.0		

Note 2I – Exceptional Items

There were no new exceptional items of expense or income in 2013/14.

Note 2J – Events after the Reporting Period

There have been no significant events after the reporting period for 2013/14.

Note 3 – Current Receivables and Payables

The tables below show amounts owed to the council (receivables), and amounts owed by the council (payables) at the end of the year, split by type of organisation.

3A - Current Receivables

31 Marc	ch 2013		31 March	n 2014	
Council	Group	Type of organisation	Council	Group	
£m	£m		£m	£m	
32.4	32.4	Central government bodies	42.0	42.0	
37.2	36.6	Bodies external to general government	35.1	33.9	
69.6	69.0	Total	77.1	75.9	

שַ	69.6	69.0	Total	77.1	75.9
12	B – Current Pay	ables			
93	31 March	n 2013		31 March	n 2014
	Council	Group	Type of organisation	Council	Group
	£m	£m		£m	£m
	(4.3)	(5.4)	Central government bodies	(4.3)	(5.4)
	(0.2)	(0.2)	Other local authorities	(1.0)	(1.0)
	(0.7)	(0.7)	NHS bodies	(0.1)	(0.1)
	(71.4)	(72.8)	Bodies external to general government	(66.0)	(68.5)
	(76.6)	(79.1)	Total	(71.4)	(75.0)

Note 4 – Provisions and Contingent Liabilities

4A – Provisions

	Balance at 31 March 2013 £m	Provision Name	Provision Details	Amounts Used in 2013/14 £m	Provisions Made in 2013/14 £m	Balance at 31 March 2014 £m
	(26.5)	Capitalisation Risks	This provision is in respect of potential claims under equal pay legislation. The council had approval from the Government to capitalise payments it may need to make in respect of Equal Pay Back Pay claims. It is currently uncertain when payments might need to be made, and the value of any such payments.	8.2	(0.4)	(18.7)
Dago 0/	(2.6)	Insurance	The council self-insures risks to property and assets up to a total aggregate limit of £1.0 million and its liability exposures up to a limit of £250,000 on any one occurrence above which limits the external insurance cover operates. The insurance provision of £2.3 million is in respect of the outstanding claims under the self-insurance programme covering the current and past years.	0.3	-	(2.3)
	(0.5)	Termination Benefits	During 2012/13 and 2013/14, the council undertook a voluntary redundancy exercise. As a result of this initiative, there were a number of employees and former employees to whom termination benefits were due, but had not yet been made, at the end of the year.	0.5	(1.8)	(1.8)
	(0.2)	Midlands Housing Consortium (MHC)	MHC was previously a member of the West Midlands Pension Fund. It paid a lump sum to the council to support pension payments to fund members. This provision will reduce gradually over time as pension payments are made.	1	-	(0.2)
	(0.5)	Carbon Reduction Commitment	This provision is in respect of the council's liability under the national, compulsory Carbon Reduction Commitment scheme. It represents the council's estimate of the amount it will have to pay to purchase allowances for its use of carbon in 2013/14.	0.5	(0.6)	(0.6)
	(0.1)	Housing Revenue Account	There are three separate provisions: for legal disrepair claims, for tenant management organisation expenditure and for rent bonds.	-	-	(0.1)

Balance at 31 March 2013 £m	Provision Name	Provision Details	Amounts Used in 2013/14 £m	Provisions Made in 2013/14 £m	Balance at 31 March 2014 £m
-	Outstanding NNDR appeals	The Collection Fund account requires a provision for appeals against the rateable valuation set by the Valuation Office Agency (VOA) not settled as at 31 March 2014. This is the first year of this provision.	-	(1.7)	(1.7)
(0.2)	Other	These are small amounts relating to ex-members of the pension fund and refunds of aftercare payments made by residents subsequently falling within Section 117 of the Mental Health Act 1983 and from whom charges are not due.	0.1	-	(0.1)
(30.6)	Total		9.6	(4.5)	(25.5)

4B – Contingent Liabilities

At 31 March 2014, the council had the following contingent liabilities:

- The council entered into a waste disposal contract on 8 February 1996. Under the contract, the waste disposal contractor has constructed a waste energy plant at an estimated cost of £26.6 million and the plant became operational in February 1998. If the contract is terminated by the council for any reason, the council becomes liable to pay to the contractor a sum, (the termination sum), equal to 90% of £26.6 million, written down to zero on a straight line depreciation basis over 25 years commencing from the date the plant became operational. The unexpired value of the termination sum at 31 March 2014 is £9.1 million (31 March 2013 £10.1 million).
- There is a future payment to be paid under the council's well-being powers, which has arisen as a result of investigations into contamination of the site previously occupied by Courtaulds upon which there now stands a mix of private and social housing. The technical investigations into the land contamination affecting 84 properties is now complete. The council is in the process of assigning liabilities with a view to requiring those responsible to address the contamination. This however is a complex and lengthy legal process with a high likelihood of challenge from any of the parties considered by the council as to be responsible. The ability of the council to progress remediation is directly linked to the legal framework in place.
- A contingent liability exists for the costs of Equal Pay compensation. The council has established a provision for £18.7 million (31 March 2013: £26.5 million). The potential costs of any further cases not addressed by this provision cannot be reliably quantified at this stage.
- There are a number of instances where the council has acquired title to assets of land and buildings due to the use of compulsory purchase orders or an interest in land and buildings subject to blight notices. In these situations there has been no transfer of economic benefits due to difficulties in either establishing the original owner or in reaching an agreement to the level of compensation to be transferred. The existence of a recognisable liability can only be confirmed at the point that the owner comes forward to claim reimbursement or the formal acknowledgment of blight has been established. The total value of these cases as at 31 March 2014 is estimated at £629,000 (31 March 2013: £556,000).
- Under recent guidance issued by the Department for Environment Food and Rural Affairs in respect of the Environmental Information Regulations 2004, it has been suggested that local authorities do not have the power to charge fees for 'personal' local land register searches, and that they may be liable to repay fees they have received since 2005. For the council, this would equate to £200,000. If all property search fees are taken into account, the total liability is £2.1 million. However, the legal position remains unclear, meaning that the existence of a liability cannot be confirmed until the relevant legal issues are resolved.

- During 2013/14, the council undertook a voluntary redundancy exercise. There were a number of applications for voluntary redundancy that were approved prior to the balance sheet date for payment during 2014/15. A provision for £1.8 million was therefore established. However there were a number of applications for voluntary redundancy in progress which had not received sufficient approvals as at the Balance Sheet date to make it reasonably certain that a redundancy would ultimately result. It is not possible to place a reliable estimate on the number of such applications that will ultimately result in redundancy, and therefore the value of any resulting liability.
- Advantage West Midlands (AWM) arranged for the land at i54 to be remediated in 2004 and any contamination found at that time was removed from the site. At the time of purchasing part of the land at i54 from Seven Trent Water, AWM (now succeeded by the Housing and Community Agency (HCA)) agreed to indemnify Severn Trent against any future contamination claims from third parties. When Staffordshire County Council purchased the land at i54 from the HCA, Staffordshire County Council were required to agree to the same indemnity relating to the former Severn Trent land. Under the Joint Venture Agreement relating to i54, Wolverhampton City Council would be required to fund 50% of any liability arising from this indemnity, subject to a cap of £2.0 million.
- A further, separate contingent liability exists with respect to infrastructure works at the i54 development. Staffordshire County Council have provided a comprehensive warranty to Jaguar Cars Limited in respect of the design element for earthworks to plots. Under the Joint Venture Agreement relating to i54, Wolverhampton City Council would be required to fund 50% of any liability arising from this warranty, subject to a cap of £2.5 million.
- The council was previously insured by Municipal Mutual Insurance (MMI). MMI is subject to a contingent Scheme of Arrangement, the terms of which were triggered in November 2012. As a result, there is the possibility that the council may be subject to claw back of both previous and future paid claims. The value of any such claw back is estimated to be approximately £200,000.

Note 5 - Non-Current Assets

The following tables analyse movements in the carrying values of non-current assets during the year.

	2013/14									Carrying Value at 31 March 2014						
	Gross Value	Accumulated Depreciation/Impairment	Net Value	Additions	Disposals (Gross Value)	Disposals (Accumulated Depreciation/Impairment)	Revaluations/ Fair Value Gains/(Losses)	Depreciation Writeback on Revaluation	Impairments	Impairments Reversed	Depreciation/ Amortisation	Other Changes - Gross Value	Other Changes - Accumulated Depreciation/ Impairment	Gross Value	Accumulated Depreciation/Impairment	Net Value
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Council Dwellings	1,261.8	(574.0)	687.8	43.1	(7.0)	-	-	-	(0.2)	0.2	(21.9)	-	-	1,297.9	(595.9)	702.0
Other Land and Buildings	866.6	(216.7)	649.9	31.2	(75.6)	11.1	(33.4)	44.5	(24.2)	11.0	(24.1)	(9.4)	0.9	779.4	(197.5)	581.9
Vehicles, Plant, Furniture																
and Equipment	66.8	(51.9)	14.9	5.1	-	-	-	-	-	-	(5.6)	-	-	71.9	(57.5)	14.4
Infrastructure Assets	256.0	(125.7)	130.3	6.8	-	-	-	-	-	-	(10.3)	-	-	262.8	(136.0)	126.8
Community Assets	27.5	(4.9)	22.6	1.1	-	-	(7.8)	3.4	(5.5)	-	-	-	-	20.8	(7.0)	13.8
Surplus Assets	43.1	(15.0)	28.1	0.3	(2.1)	(8.0)	(1.5)	1.6	(3.2)	0.1	(0.9)	9.7	(0.9)	49.5	(19.1)	30.4
Assets Under Construction	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Investment Properties	16.9	(2.6)	14.3	2.3	-	-	1	-	-	-	-	(0.3)	-	18.9	(2.6)	16.3
Intangible Assets	2.5	(8.0)	1.7	3.2	-	-	-	-	-	-	(0.4)	-	-	5.7	(1.2)	4.5
Heritage Assets	11.5	-	11.5	-	-	-	-	-	-	-	-	-	-	11.5	-	11.5
Total	2,552.7	(991.6)	1,561.1	93.1	(84.7)	10.3	(42.7)	49.5	(33.1)	11.3	(63.2)	-	-	2,518.4	(1,016.8)	1,501.6

	Carrying Value at 1 April 2012						2012/13									Carrying Value at 31 March 2013		
		Gross Value	Accumulated Depreciation/Impairment	Net Value	Additions	Disposals (Gross Value)	Disposals (Accumulated Depreciation)	Revaluations/ Fair Value Gains/(Losses)	Depreciation Writeback on Revaluation	Impairments	Impairments Reversed	Depreciation/ Amortisation	Other Changes - Gross Value	Other Changes - Accumulated Depreciation/ Impairment	Gross Value	Accumulated Depreciation/Impairment	Net Value	
ñ		£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	
עַ																		
a De	Council Dwellings	1,224.2	(552.1)	672.1	41.4	(3.8)	-	-	-	-	-	(21.9)	-	-	1,261.8	(574.0)	687.8	
Page 9	Council Dwellings Other Land and Buildings						12.5	- (105.1)	68.8	(67.0)	3.5		(29.8)					
99 ans	•	1,224.2	(552.1)	672.1	41.4	(3.8)	-	-	-	(67.0)	-	(21.9)	(29.8)	-	1,261.8	(574.0)	687.8	
	Other Land and Buildings	1,224.2	(552.1)	672.1	41.4	(3.8)	-	-	-	- (67.0)	-	(21.9)	(29.8)	-	1,261.8	(574.0)	687.8	
	Other Land and Buildings Vehicles, Plant, Furniture	1,224.2 951.2	(552.1) (212.9)	672.1 738.3	41.4 92.9	(3.8)	-	-	-	- (67.0) - -	-	(21.9)	, , ,	-	1,261.8 866.6	(574.0) (216.7)	687.8 649.9	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment	1,224.2 951.2 62.5	(552.1) (212.9) (46.0)	672.1 738.3 16.5 135.0	41.4 92.9 4.1	(3.8)	-	-	-	- (67.0) - -	-	(21.9) (23.6) (5.9)	0.2	2.0	1,261.8 866.6 66.8	(574.0) (216.7) (51.9)	687.8 649.9 14.9	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment Infrastructure Assets	1,224.2 951.2 62.5 250.2	(552.1) (212.9) (46.0) (115.2)	672.1 738.3 16.5 135.0 22.5	41.4 92.9 4.1 6.8	(3.8) (42.6)	-	- (105.1) - -	-	- (67.0) - - - (11.2)	-	(21.9) (23.6) (5.9)	0.2	2.0	1,261.8 866.6 66.8 256.0	(574.0) (216.7) (51.9) (125.7)	687.8 649.9 14.9 130.3	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment Infrastructure Assets Community Assets	1,224.2 951.2 62.5 250.2 27.4	(552.1) (212.9) (46.0) (115.2) (4.9)	672.1 738.3 16.5 135.0 22.5	41.4 92.9 4.1 6.8 0.3	(3.8) (42.6) - - (0.1)	-	- (105.1) - - (0.1)	- 68.8 - -		3.5	(21.9) (23.6) (5.9) (10.5)	0.2 (1.0)	2.0	1,261.8 866.6 66.8 256.0 27.5	(574.0) (216.7) (51.9) (125.7) (4.9)	687.8 649.9 14.9 130.3 22.6	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment Infrastructure Assets Community Assets Surplus Assets	1,224.2 951.2 62.5 250.2 27.4 13.2	(552.1) (212.9) (46.0) (115.2) (4.9)	672.1 738.3 16.5 135.0 22.5 8.0	41.4 92.9 4.1 6.8 0.3	(3.8) (42.6) - - (0.1)	- 12.5 - - - -	- (105.1) - - (0.1)	- 68.8 - -		3.5	(21.9) (23.6) (5.9) (10.5)	0.2 (1.0)	- 2.0 - - - (7.4)	1,261.8 866.6 66.8 256.0 27.5	(574.0) (216.7) (51.9) (125.7) (4.9)	687.8 649.9 14.9 130.3 22.6	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment Infrastructure Assets Community Assets Surplus Assets Assets Under Construction	1,224.2 951.2 62.5 250.2 27.4 13.2 0.9 19.2 1.9	(552.1) (212.9) (46.0) (115.2) (4.9) (5.2)	672.1 738.3 16.5 135.0 22.5 8.0 0.9 16.3 1.4	41.4 92.9 4.1 6.8 0.3 1.0	(3.8) (42.6) - (0.1) (0.7)	- 12.5 - - - -	- (105.1) - - (0.1)	- 68.8 - -		3.5	(21.9) (23.6) (5.9) (10.5)	0.2 (1.0)	- 2.0 - - - (7.4)	1,261.8 866.6 66.8 256.0 27.5 43.1	(574.0) (216.7) (51.9) (125.7) (4.9) (15.0)	687.8 649.9 14.9 130.3 22.6 28.1 - 14.3 1.7	
	Other Land and Buildings Vehicles, Plant, Furniture and Equipment Infrastructure Assets Community Assets Surplus Assets Assets Under Construction Investment Properties	1,224.2 951.2 62.5 250.2 27.4 13.2 0.9 19.2	(552.1) (212.9) (46.0) (115.2) (4.9) (5.2) - (2.9)	672.1 738.3 16.5 135.0 22.5 8.0 0.9 16.3 1.4	41.4 92.9 4.1 6.8 0.3 1.0 -	(3.8) (42.6) - (0.1) (0.7)	- 12.5 - - - - 0.3	- (105.1) - - (0.1)	- 68.8 - - - 8.6		3.5	(21.9) (23.6) (5.9) (10.5) - (1.1)	0.2 (1.0)	- 2.0 - - - (7.4)	1,261.8 866.6 66.8 256.0 27.5 43.1 - 16.9	(574.0) (216.7) (51.9) (125.7) (4.9) (15.0) - (2.6)	687.8 649.9 14.9 130.3 22.6 28.1	

Depreciation/Amortisation

Property, plant and equipment assets are depreciated on a straight-line basis over the estimated useful economic life of the asset, with the exception of council dwellings, for which the Major Repairs Allowance is used as a proxy for depreciation. Intangible assets are amortised on a straight-line basis over the estimated useful economic life of the asset. No depreciation is charged on investment properties, heritage assets or land.

Revaluation

During 2013/14 the council re-valued a number of assets. The effective date of this revaluation was 1 April 2013. These valuations were carried out by the council's in-house valuation team using the guidance and methodology set out in the following paragraphs.

Legislation and guidance

Valuations are carried out as required by Sections 41 to 42 of The Local Government Housing Act 1989 and in accordance with CIPFA/IFRS guidance and the RICS Valuation Standards (Red Book). The valuations are subject to any limitations, caveats and assumptions as agreed between the Section 151 Officer and the Assistant Director Regeneration.

Basis of valuation

In accordance with the CIPFA Code of Practice on Local Authority Accounting 2013/14, infrastructure, community assets, and assets under construction are valued at depreciated historical cost. All other classes of asset are measured at fair value. For vehicles, plant, furniture and equipment fair value is determined by depreciated historical cost due to the short useful life of these assets (2-7 years depending on the asset). The fair value of council dwellings is measured using existing use value (social housing). For all other asset classes, where there is no market-based evidence of fair value the council estimates fair value using the depreciated replacement cost approach. The following table describes the measurement bases used to determine the gross carrying value of each of the council's classes of non-current assets.

Asset Class	Measurement Base
Council Dwellings	Fair value based on existing use value (social housing) (EUV-SH).
Other Land and Buildings	Fair value based on existing use value (EUV) or depreciated replacement cost (DRC) using the "instant build"
	approach if EUV cannot be determined.
Vehicles, Plant, Furniture and	Fair value based on depreciated historical cost due to the short useful life of the asset.
Equipment	
Infrastructure Assets	Depreciated historical cost.
Community Assets	Depreciated historical cost or valuation.
Surplus Assets	Fair value based on existing use value (EUV) applying the same assumptions relating to the level of usage, etc, as
	those of the most recent revaluation as an operational asset.
Assets Under Construction	Depreciated historical cost.
Investment Properties	Initially measured at cost (defined as the purchase price plus any directly attributable cost of preparing the asset for
	its intended use) and subsequently measured at fair value
Intangible Assets	Amortised Cost
Heritage Assets	Where the council has information on the cost or value of the asset, the asset is recognised at this amount.

D and section of the section of the

A desktop exercise is carried out using information from the land registry. It is assumed that any building is constructed in accordance with Building Regulations and does not contain any deleterious or hazardous substances. Detailed building surveys are not requisitioned unless there are obvious areas of concern which are likely to affect the valuation.

Key Assumptions

- Planning Planning consideration is, in the first instance, by reference to the Local Development Framework to ensure the use of the property to be valued is in accordance with the council's stated policies.
- Ground Conditions no reference is made to ground conditions unless it is evident upon inspection that a particular property has been affected by exceptional external influences greater than would be anticipated for its locality.
- Contamination no reference is made to contamination unless, upon inspection, it is evident that a particular property has been affected by external influences greater than would be anticipated for its locality.

Capital Commitments

At 31 March 2014, the council had entered into a number of contracts for the construction or enhancement of Property, Plant and Equipment in 2013/14 and future years with an estimated total value of £47.2 million (31 March 2013: £100.9 million). The major commitments are: Building Schools for the Future (£15.3 million), i54 (£10.2 million), Interchange Block 10 development (£7.6 million) and Decent Homes (£6.0 million).

Investment Properties

During the year the council had £1.3 million of income receivable from investment properties (2012/13: £1.0 million) and spent £226,000 on managing and maintaining those properties (2012/13: £140,000). There are no restrictions on the council's ability to realise the value of its investment property, the remittance of income or the proceeds of disposal. The council has no contractual obligations to purchase, construct or develop investment property, or for repairs, maintenance or enhancements.

Impairment

During 2013/14, the council identified a number of impairments as part of the regular review of its properties. The total value of those impairments was £21.8 million. There were no impairments that were individually material during 2013/14.

→ Capital Financing Requirement

The council's capital financing requirement, which represents the underlying need to borrow to pay for past capital expenditure, was £768.4 million at 31 March 2014 (31 March 2013: £750.5 million).

Note 6 – Employee Pensions

Participation in Pension Schemes

As part of the terms and conditions of employment of its officers, the council makes contributions towards the cost of post-employment benefits. Although these benefits will not actually be payable until employees retire, the council has a commitment to make the payments, which needs to be disclosed at the time that employees earn their future entitlement.

The council participates in two post-employment schemes and provides a further local discretionary scheme:

- The Local Government Pension Scheme, administered locally by The West Midlands Pension Fund. This is a funded defined benefit final salary scheme, meaning that the council and employees pay contributions into a fund, calculated at a level intended to balance the pensions liabilities with investment assets.
- Teachers employed by the council are members of the Teachers' Pension Scheme, administered by the Teachers' Pensions Agency (TPA). It provides teachers with defined benefits upon their retirement and the council contributes towards the costs by making contributions based on a percentage of members' pensionable salaries. During the year the council contributed £3.8 million which was a contribution rate of 14.1% (2012/13: £3.7 million; 14.1%).
- In addition, the council is responsible for all non-funded pension payments relating to added years enhancements it has awarded outside the terms of the teachers' scheme together with related increases.

From April 2014, pensions will be worked out in a different way as the scheme became a career average scheme. Benefits built up from April 2014 are worked out using actual pay each scheme year rather than final salary when an employee leaves. Protections are in place for all the benefits built up in the final salary scheme.

Transactions Relating to Post-employment Benefits

The council recognises the cost of retirement benefits in the cost of services when they are earned by employees, rather than when the benefits are eventually paid as pensions. However, the charge the council is required to make against council tax is based on the cash payable in the year, so the real cost of post-employment/retirement benefits is reversed out of the General Fund in the Movement in Reserves Statement. The following transactions have been made in the Comprehensive Income and Expenditure Statement and the Movement in Reserves Statement during the year:

	2012	/13		2013/	/14
	LGPS £m	Teachers £m		LGPS £m	Teachers £m
			Comprehensive Income And Expenditure Statement		
			Cost of Services:		
	(22.2)	-	- Current service cost	(28.9)	-
	(0.2)	-	- Past service costs	(0.1)	-
	2.1	(0.1)	- Settlements and curtailments	3.5	(0.3)
	(0.3)	-	- Administration Expenses	(0.4)	-
			Financing and Investment Income and Expenditure:		
	(18.4)	(2.4)	- Net Interest cost	(20.0)	(2.1)
	(39.0)	(2.5)	Total Post Employment Benefit Charged to the Surplus or Deficit on the Provision of Services	(45.9)	(2.4)
			Other Deat Francisco A Deag #4 Ober and the the Common bearing because		
2			Other Post Employment Benefit Charged to the Comprehensive Income and Expenditure Statement:		
-	(87.0)	(7.0)	- Re-measurements (liabilities and assets)	106.9	2.7
	(126.0)	(9.5)	Total Post Employment Benefit Charged to the Comprehensive Income and Expenditure Statement	61.0	0.3
			Movement In Reserves Statement		
	(31.1)	(2.5)	- Reversal of net charges made to the Surplus or Deficit for the Provision of Services for post-employment benefits in accordance with the Code	(45.9)	(2.4)
	_		Actual amount charged against the General Fund Balance for pensions in the year:		
	23.4		- Employer's contributions payable to scheme	24.8	
		3.7	- Retirement benefits payable to pensioners		3.8

'age 105

6. THE FINANCIAL STATEMENTS

Assets and Liabilities in Relation to Post-employment Benefits

The following tables show how the present values of the scheme assets and liabilities have changed over the course of the year:

2012/13 LGPS £m	Assets	2013/14 LGPS £m
675.0	Opening balance at 1 April	747.9
32.8	Interest Income	31.3
49.8		(7.4)
23.4	Employer contributions	24.8
7.3	Contributions by scheme participants	7.8
(38.3)	Benefits paid	(40.0)
(1.8)	Settlements	(0.3)
(0.3)	Admin expenses	(0.4)
747.9	Closing balance at 31 March	763.7

2012/13 Funded: LGPS £m	2012/13 Unfunded: LGPS £m	2012/13 Unfunded: Teachers £m	Liabilities	2013/14 Funded: LGPS £m	2013/14 Unfunded: LGPS £m	2013/14 Unfunded: Teachers £m
(1,043.0)	(21.4)	(54.0)	Opening balance at 1 April	(1,217.3)	(22.6)	(59.8)
(22.2)	-	-	Current service cost	(28.9)	-	-
(50.2)	(1.0)	(2.4)	Interest cost	(50.4)	(0.9)	(2.1)
(7.3)	-	-	Contributions - participants	(7.8)	-	-
(135.0)	(1.8)	(7.1)	Re-measurement Gain/(Loss)	114.8	(0.5)	2.7
36.7	1.6	3.7	Benefits paid	38.4	1.6	3.8
(0.2)	-	-	Past service costs	(0.1)	-	-
(1.2)	-	-	Curtailments	(1.6)	-	(0.3)
5.1	-	-	Settlements	5.4	-	-
(1,217.3)	(22.6)	(59.8)	Closing balance at 31 March	(1,147.5)	(22.4)	(55.7)

The expected return on scheme assets is determined by considering the expected returns available on the assets underlying the current investment policy. Expected yields on fixed interest investments are based on gross redemption yields as at the Balance Sheet date.

Expected returns on equity investments reflect long-term real rates of return experienced in the respective markets. The actual return on scheme assets in the year was £22.8 million (2012/13: (£82.3 million).

The liabilities show the underlying commitments that the council has in the long run to pay post-employment (retirement) benefits. The total liability of £461.9 million has a substantial impact on the net worth of the council as recorded in the Balance Sheet, reducing it from £ million to £ 0.10 million. However, statutory arrangements for funding the deficit mean that the financial position of the council remains healthy, because:

- The deficit on the local government scheme will be made good by increased contributions over the remaining working life of employees (i.e. before payments fall due), as assessed by the scheme actuary;
- Finance is only required to be raised to cover discretionary benefits when the pensions are actually paid.

The total contributions expected to be made to the Local Government Pension Scheme by the council in the year to 31 March 2015 is £25.9 million. Expected contributions for the Discretionary Benefits scheme in the year to 31 March 2015 are £3.9 million.

The Discretionary Benefits arrangements have no assets to cover their liabilities. The Local Government Pension Scheme's assets are set out in the following table.

All scheme assets have quoted prices in active markets, except UK unquoted and global unquoted.

	2012/13 LGPS £m	Asset Category	2013/14 LGPS £m
		Equities	
	73.9	- UK quoted	79.4
	1.4	- UK unquoted	14.6
	37.6	- Global quoted	40.2
L	92.5	- Global unquoted	79.1
	43.9	- Europe	51.2
	14.4	- Japan	14.1
L	32.4	- Pacific Basin	30.7
L	71.9	- North America	73.5
L	46.5	- Emerging Markets	61.8
L		Bonds	
ש	14.9	- UK Government fixed	14.7
Page		- UK Government indexed	48.1
g L	36.1	- UK other	36.7
_	24.8		19.9
07	24.5		25.4
`\		Property	
L	46.1	- UK	45.3
	2.5	- Overseas	1.7
L	15.3	- Property Funds	21.8
L		Alternatives	
L	15.8	- Commodities	14.4
L	24.9	- Infrastructure	21.8
-	53.2	- Absolute Return	48.7
-	18.2	- Cash instruments	15.2
f	6.5	- Cash accounts	5.4
		Total	763.7

Basis for Estimating Assets and Liabilities

Liabilities have been assessed on an actuarial basis using the projected unit method, an estimate of the pensions that will be payable in future years dependent on assumptions about things such as mortality rates and salary levels. Both the Teachers' Discretionary Pension Scheme and the West Midlands Pension Fund liabilities have been assessed by Mercer Limited, an independent firm of actuaries. Estimates for the West Midlands Pension Fund are based on the latest full valuation of the scheme as at 31 March 2010. The principal assumptions used by the actuary are set out in the following table.

2012/13		/13		2013/	14
	LGPS	Teachers		LGPS	Teachers
			Long-term expected rate of return on assets in the scheme:		
	7.0%	n/a	Equity investments	7.0%	n/a
	2.8%	n/a	Government Bonds	3.4%	n/a
	3.9%	n/a	Other Bonds	4.3%	n/a
<u> </u>	5.7%	n/a	Property	6.2%	n/a
	0.5%	n/a	Cash/current assets	0.5%	n/a
	7.0%	n/a	Other	7.0%	n/a
° _			Mortality assumptions:		
			Longevity at 65 for current pensioners:		
	22.1	22.1	- Men	22.9	22.9
	24.8	24.8	- Women	25.5	25.5
			Longevity at 65 for future pensioners:		
	23.9	n/a	- Men	25.1	n/a
	26.7	n/a	- Women	27.8	n/a
	2.4%	2.4%	Rate of inflation	2.4%	2.4%
	4.2%	-	Rate of increase in salaries	4.2%	-
	2.4%	2.4%	Rate of increase in pensions	2.4%	2.4%
	4.2%	3.7%	Rate for discounting scheme liabilities	4.4%	4.3%

Analysis of Group Net Pension Liability

The following table analyses the net liability for the group.

31 March 2013			31 Ma	rch 2014
Local Government Pension Scheme	Discretionary Pension Scheme		Local Government Pension Scheme	Discretionary Pension Scheme
£m	£m		£m	£m
		Estimated Liabilities in Scheme		
(1,239.9)	(59.8)	Wolverhampton City Council	(1,169.9)	(55.7)
(122.1)	-	Wolverhampton Homes	(114.8)	-
(1,362.0)	(59.8)	Total Liabilities	(1,284.7)	(55.7)
J				
		Estimated Assets in Scheme		
747.9	ı	Wolverhampton City Council	763.7	-
93.8	-	Wolverhampton Homes	100.3	-
841.7	-	Total Assets	864.0	-
(520.3)	(59.8)	Net Liabilities	(420.7)	(55.7)

Note 7 – Financial Instruments

Note 7A – Leases and Lease-Type Arrangements

The table below summarises the amounts payable and receivable by the council under lease agreements during the year, and the amounts that it expects to be payable or receivable under non-cancellable lease agreements in future years.

	2012/13					2013/14			
	Amounts Payable Amounts Receivable			Amounts Payable Amounts Rece			Receivable		
	Finance	Operating	Finance	Operating		Finance	Operating	Finance	Operating
	Leases	Leases	Leases	Leases		Leases	Leases	Leases	Leases
	£m	£m	£m	£m		£m	£m	£m	£m
ַס	0.5	0.8	-	3.4	Payable/receivable in the year	0.2	8.0	-	2.9
age									
е	0.1	0.6	-	2.3	Due within one year	0.1	0.5	-	2.3
_	0.2	1.5	-	6.1	Due in one to five years	0.1	1.6	-	7.3
10	-	0.6	6.0	26.7	Due after five years	-	0.5	6.0	27.1
	0.3	2.7	6.0	35.1	Total due in future years	0.2	2.6	6.0	36.7

The following table shows the net carrying value of assets held by the council under finance lease arrangements:

31 March 2013 £m		31 March 2014 £m
0.5	Vehicles, Plant, Furniture and Equipment	0.3
0.5	Total	0.3

Note 7B – Private Finance Initiative and Similar Contracts

The council has three contracts which are PFIs or similar in nature and which are accounted for as 'on balance sheet': the Bentley Bridge leisure centre, the waste disposal facility, and Highfields and Penn Fields School.

Bentley Bridge Leisure Centre: In 2006/07 the council signed a thirty year contract for a new leisure facility. The scheme was for the design, build, funding and operation of a major new regional leisure facility. The facility includes a leisure pool with a river run, wave pool and flumes; a 25 metre 6 lane traditional pool; a studio pool; a 140 station fitness suite; a dance/aerobics suite; children's play feature and a café. The facility cost £13.3 million and opened on 12 December 2006. The facility is operated by DC Leisure Management Limited on behalf of the council. The contract period is for 30 years, with an end date of 31 October 2036. The total amount payable by the council over the life of the contract is £52.5 million. Over the remaining life of the project the commitments are:

Page		Payment for Services £m	Interest £m	Capital Expenditure /Principal Redemption £m	Total £m
e	Payable in 2014/15	0.3	0.9	0.2	1.4
11	Payable within two to five years	1.5	3.4	0.9	5.8
1	Payable within six to ten years	2.1	3.7	1.4	7.2
	Payable within eleven to fifteen years	2.4	3.1	1.7	7.2
	Payable within sixteen to twenty years	1.9	2.1	3.2	7.2
	Payable within twenty-one to twenty-four years	1.0	0.4	2.4	3.8
	Total	9.2	13.6	9.8	32.6

The table below shows the balances that existed in respect of the contract at the end of the year, and how they have changed over the course of the year.

	Balance at 31 March 2013	Depreciation	Capital Expenditure /Principal Redemption	Balance at 31 March 2014
	£m	£m	£m	£m
Property, Plant and Equipment	10.9	(0.4)	-	10.5
Long-term Liability	(9.0)	-	0.2	(8.8)
Total	1.9	(0.4)	0.2	1.7

Waste Disposal Facility: In 1996 the council signed a contract for the construction and maintenance of a waste disposal facility. The facility cost £26.6 million. The contract period during which the council has the right to use the facility is 25 years from the date that the facility became operational (1998). At the end of the contract period the asset will revert to the ownership of the council, but there is an option to then retender, operate or operate with additional investment being targeted at the plant. During the contract period the council could terminate the contract if the clauses relating to termination in the contract are triggered. The facility is managed by Wolverhampton Waste Services Limited (WWS). The main income stream for WWS is the 'gate fee' paid by the council, which is based on the total tonnage of weight delivered to the facility for disposal. In addition to this, WWS generate a significant proportion of their turnover from the sale of electricity generated at the facility. The total amount payable by the council over the life of the contract is estimated to be £155.6 million. Over the remaining life of the contract the estimated payments are:

	Payment for Services £m	Interest	Capital Expenditure /Principal Redemption £m	Total £m
Payable in 2014/15	6.3	£m 0.8	1.0	8.1
Payable within two to five years	28.1	1.9	4.9	34.9
Payable within six to eight years	15.9	0.3	3.0	19.2
Total	50.3	3.0	8.9	62.2

The table below shows the balances that existed in respect of the contract at the end of the year, and how they have changed over the course of the year.

	Balance at 31 March 2013 £m	Depreciation/ Amortisation	Capital Expenditure /Principal Redemption £m	Balance at 31 March 2014 £m
Property, Plant and Equipment	9.9	£m (1.2)	Z.III -	8.7
Deferred Income	(2.8)	0.3	-	(2.5)
Long-term Liability	(9.9)	-	0.9	(9.0)
Total	(2.8)	(0.9)	0.9	(2.8)

Highfields & Penn Fields School: As part of the Building Schools for the Future Programme the council entered in to a PFI contract for the construction and management of a new building for the Highfields and Penn Fields Special School. The construction of the new building cost £47.8 million. The total amount payable by the council over the life of the contract is estimated to be £192.7 million. Over the remaining life of the contract the estimated payments are:

	Payment for Services	Interest	Capital Expenditure /Principal Redemption	Total
	£m	£m	£m	£m
Payable in 2014/15	1.8	4.8	0.6	7.2
Payable within two to five years	7.7	18.8	2.9	29.4
Payable within six to ten years	11.0	21.3	5.8	38.1
Payable within eleven to fifteen years	11.9	18.0	9.7	39.6
Payable within sixteen to twenty years	15.4	11.1	15.0	41.5
Payable within twenty-one to twenty-five years	12.1	3.2	14.4	29.7
Total	59.9	77.2	48.4	185.5

The table below shows the balances that existed in respect of the contract at the end of the year, and how they have changed over the course of the year.

	Balance at 31 March 2013 Depreciation		Capital Expenditure /Principal Redemption	Balance at 31 March 2014	
	£m	£m	£m	£m	
Property, Plant and Equipment	46.1	(1.8)	-	44.3	
Long-term Liability	(44.0)		0.5	(43.5)	
Total	2.1	(1.8)	0.5	8.0	

Note 7C – Financial Instruments

The table below shows the carrying values and fair values of financial instruments held by the council at the year end.

		31 Mar	ch 2013				31 Mar	ch 2014	
	Carrying	ng Value Fair Value		alue		Carrying	Carrying Value Fair		
Lo	ng-Term	Current	Long-Term	Current		Long-Term	Current	Long-Term	Current
	£m	£m	£m	£m		£m	£m	£m	£m
					Financial Assets				
	1.5	74.1	1.5	74.1	Loans and Receivables	1.4	89.6	1.4	89.6
	18.6	-	18.6	-	Unquoted Equity Investment at Cost	20.3	-	20.3	-
	-	3.5	-	3.5	Cash and Cash Equivalents	-	4.1	-	4.1
ַ	20.1	77.6	20.1	77.6	Total Financial Assets	21.7	93.7	21.7	93.7
D					Financial Liabilities				
72	(588.3)	(120.4)	(644.6)	(122.2)	Financial Liabilities at Amortised Cost	(605.5)	(136.7)	(636.5)	(136.8)
"	(588.3)	(120.4)	(644.6)	(122.2)	Total Financial Liabilities	(605.5)	(136.7)	(636.5)	(136.8)
	(568.2)	(42.8)	(624.5)	(44.6)	Net Financial Liabilities	(583.8)	(43.0)	(614.8)	(43.1)

The gains and losses recognised in the Comprehensive Income and Expenditure Statement in relation to financial instruments are shown in the following table.

		201 Financial	2/13				201 Financial	3/14	
	Financial Assets: Loans and Receivables	Assets: Unquoted Equity Investment at Cost	Financial Liabilities Measured at Amortised Cost	Total		Financial Assets: Loans and Receivables	Assets: Unquoted Equity Investment at Cost	Financial Liabilities Measured at Amortised Cost	Total
	£m	£m	£m	£m		£m	£m	£m	£m
	-	-	25.4	25.4	Interest Expense	-	-	- 31.3	31.3
, [(0.4)	-	-	(0.4)	Interest Income	(0.4)	-	-	(0.4)
<u> </u>	(0.4)	-	25.4	25.0	Net (Income)/Expense	(0.4)		31.3	30.9

The council holds a small amount of HM Treasury loans, which have been valued according to published quotations in an active market. The fair values of all other financial instruments have been derived from valuation techniques based on assumptions that are not supported by prices from observable current market transactions in the same instrument, and not based on available observable market data.

The fair values of the council's long-term financial liabilities at amortised cost have been calculated using the Net Present Value (NPV) approach, which provides an estimate of the value of payments in the future in today's terms. The discount rate used in the NPV calculation is equal to the current rate in relation to the same instrument from a comparable lender, being the rate applicable in the market on the date of valuation, for an instrument with the same duration: that is, equal to the outstanding period from valuation date to maturity.

Note 7D - Risks Arising from Financial Instruments

There are a number of risks associated with the council's financial instruments, which the council seeks to actively identify and manage. A key part of this is the preparation of the following documents on an annual basis, in accordance with the CIPFA Treasury Management Code and the Prudential Code:

- Treasury Management Strategy
- Annual Investment Strategy
- Prudential and Treasury Management Indicators

These strategies and indicators set out the council's approach to the key risks arising from financial instruments and how it will monitor and manage those risks. These are reflected in the following paragraphs.

Credit and Counterparty Risk Management

The security of principal sums invested is the council's top priority when making investment decisions: accordingly it only places sums with institutions for whom credit risk is assessed as very low. In order to form this assessment the council applies the creditworthiness model supplied by its external treasury advisors Capita Asset Services, which takes into account credit watches and credit outlooks from credit rating agencies, credit default swap → spreads, and sovereign ratings. The council also uses market data and market information, information on government support for banks and the redit ratings of the government in question. The council has combined these factors to develop a range of creditworthiness criteria which it applies strictly when making investment decisions.

The council's maximum exposure to credit risk at 31 March 2014 was £0.2 million (31 March 2013: £ million). This relates entirely to Loans and Receivables. The council does not hold any collateral in respect of these amounts.

Collateral and Other Credit Enhancements Obtained

The council did not obtain any collateral or other credit enhancements during 2013/14 or 2012/13.

Liquidity Risk Management

The council ensures it has adequate though not excessive cash resources, borrowing arrangements, overdraft or standby facilities to enable it at all times to have the level of funds available which are necessary for the achievement of its business/service objectives. The council actively manages its

cash balances on a daily basis, and forecasts cash requirements several months in advance. In its investment activities the council places strong emphasis on liquidity (second only to security, as discussed under credit risk).

Analysis of External Borrowing Financial Liabilities by Maturity Date

2012/13 £m	Time until Repayment	2013/14 £m
39.0	Payable next year	45.0
103.2	Payable within two to five years	118.2
10.2	Payable within six to ten years	17.3
22.6	Payable within eleven to fifteen years	20.0
13.3	Payable within sixteen to twenty years	8.9
31.0	Payable within twenty-one to twenty-five years	31.0
39.9	Payable within twenty-six to thirty years	44.3
ປ <u>39.9</u>	Payable within thirty-one to thirty-five years	27.5
53.2	Payable within thirty-six to forty years	97.6
44.3	Payable within forty-one to forty-five years	26.6
26.6	Payable within forty-six to fifty years	23.0
103.8	Payable within fifty-one to sixty years	103.8
519.0	Total	563.2

Interest Rate Risk Management

The council manages its exposure to fluctuations in interest rates with a view to containing its net interest costs, or securing its interest revenues. It achieves this by the prudent use of its approved financing and investment instruments, methods and techniques, primarily to create stability and certainty of costs and revenues, but at the same time retaining a sufficient degree of flexibility to take advantage of unexpected, potentially advantageous changes in the level or structure of interest rates. These are subject at all times to the consideration and, if required, approval of any policy or budgetary implications.

If interest rates had been 1% higher during 2013/14, the council's interest payable would have increased by £5.5 million, and its interest receivable would have increased by £353,000, resulting in an increase in net expenditure of £5.1 million. Had interest rates been 1% lower, equivalent decreases would have occurred, leading to a decrease in net expenditure of £5.1 million.

Inflation Risk Management

The effects of varying levels of inflation, insofar as they can be identified as impacting directly on the council's treasury management activities, are controlled as an integral part of the council's strategy for managing its overall exposure to inflation.

The council achieves this objective by the prudent use of its approved financing and investment instruments, methods and techniques, primarily to create stability and certainty of costs and revenues, but at the same time retaining a sufficient degree of flexibility to take advantage of unexpected, potentially advantageous changes in the level or structure of interest rates. These are subject at all times to the consideration and, if required, approval of any policy or budgetary implications.

Price Risk

The council does not generally invest in equity shares but does have shareholdings to the value of £18.6 million in Birmingham International Airport as well as an investment of £1.6 million in the SPV for Highfields PFI scheme. The council is consequently exposed to losses arising from movements in the prices of the shares. As the shareholdings have arisen in the acquisition of specific interests, the council is not in a position to limit its exposure to price movements by diversifying its portfolio. The shares are all classified as 'unquoted equity investment at cost', meaning that all movements in price will impact on gains and losses recognised in Other Comprehensive Income and Expenditure.

Refinancing Risk Management

Refinancing Risk Management

The council ensures that its borrowing, private financing and partnership arrangements are negotiated, structured and documented, and the maturity profile of the monies so raised is managed, with a view to obtaining offer terms for renewal or refinancing, if required, which are competitive and as favourable to the organisation as can reasonably be achieved in the light of market conditions prevailing at the time. It actively manages its relationships with its counterparties in these transactions in such a manner as to secure this objective, and avoids over-reliance on any one source of funding if this might jeopardise achievement of the above.

Legal and Regulatory Risk Management

The council ensures that all of its treasury management activities comply with its statutory powers and regulatory requirements. It is able to demonstrate such compliance, if required to do so, to all parties with whom it deals in such activities. The council ensures that there is evidence of counterparties' powers, authority and compliance in respect of the transactions they may carry out with the organisations, particularly with regard to duty of care and fees charged.

The council recognises that future legislative or regulatory changes may impact on its treasury management activities and, so far as it is reasonably able to do so, seeks to minimise the risk of these impacting adversely on the organisation.

Fraud, Error and Corruption Risk, and Contingency Management

The council ensures that it has identified the circumstances that may expose it to the risk of loss through fraud, error, corruption or other eventualities in its treasury management dealings. Accordingly, it employs suitable systems and procedures, and maintains effective contingency management arrangements to these ends.

Note 8 - Members of the Wolverhampton City Council Group and other Related Parties

Subsidiary

The council has one subsidiary entity: Wolverhampton Homes Limited. Wolverhampton Homes Limited is an arm's length management organisation (ALMO) which was established in 2005 to manage and maintain most of the council's housing stock, and is wholly owned by the council. The company's accounts have been wholly consolidated in the group elements of the financial statements.

Wolverhampton Homes Limited's main income stream comes from the council, in the form of a management fee for housing management and maintenance which amounted to £38.0 million in 2013/14 (2012/13: £37.0 million). There are a number of other transaction streams between the two entities, including capital works carried out by Wolverhampton Homes Limited for the council, support services provided by the council, and premises leases payable by Wolverhampton Homes Limited. Payments by the council to Wolverhampton Homes Limited amounted to £46.1 million in 2013/14 (2012/13: £45.2 million), whilst payments by Wolverhampton Homes Limited to the council totalled £4.3 million (2012/13: £5.0 million). At the year end, Wolverhampton Homes Limited owed the council £1.6 million (2012/13: £1.2 million), and the council owed Wolverhampton Homes Limited £0.8 million (2012/13: £1.8 million).

Central Government

┰

Central Government has effective control over the general operations of the council. It is responsible for providing the statutory framework within which the council operates, provides the majority of its funding in the form of grants and prescribes the terms of many of the transactions that the council has with other parties. The following table provides details of the grants received from Central Government.

	2012/13 £m	Grant	2013/14 £m
	(174.1)	Dedicated Schools Grant	(177.3)
	(2.6)	Revenue Support Grant	(105.2)
	(56.8)	Mandatory Rent Allowance	(59.0)
	(54.5)	Standards Fund Capital	(58.6)
	(56.0)	Mandatory Rent Rebates Subsidy	(56.4)
	-	Top Up Grant	(34.6)
	-	Public Health Grant	(18.8)
	(15.5)	•	(14.1)
	, ,	6th Form Funding	(7.7)
	(13.6)	Section 31 Grant	(6.6)
	(2.9)	Housing & Council Tax Benefit Administration	(2.8)
F	(1.1)		(2.2)
Pa	(1.5)	Adult Community Learning PCDL	(1.6)
age	(1.2)	Further Education 19+	(1.3)
		Social Fund Programme Grant	(1.3)
12		Weekly Collection Support Scheme	(1.3)
22	(1.3)	Disabled Facilities Grant	(1.1)
	-	Adoption Reform Grant	(1.1)
	(0.2)	Council Tax Freeze Grant	(1.0)
	(134.2)	NNDR	-
	(13.8)		-
	(11.0)	Learning Disability & Health Reform	-
	(27.6)	Council Tax Benefit	-
	(7.6)	Other grants (each less than £1.0m)	(6.4)
	(585.3)	Total	(558.4)

Councillors

Councillors have direct control over the council's financial and operating policies. The total of councillors' allowances paid during the year is shown in Note 2D. The register of councillors' interests is available on the council's website:

http://www2.wolverhampton.gov.uk/council/councillors/name

Other Related Parties

The table below shows total expenditure and income streams of £100,000 or more with other related parties of the council during the year.

I	2012	2/13		2013/ ⁻	14
	Expenditure	Income	Entity and Nature of Relationship	Expenditure	Income
	£m	£m	District Alexander (11-12-12-12-12-12-12-12-12-12-12-12-12-1	£m	£m
	-	(0.7)	Birmingham Airport Holdings Limited	-	(4.1)
			The Council holds 4.7% of the total ordinary shares and 9.58% of the preference		
			shares of the company. In 2013/14 the council received a preference dividend of £93,000, special dividend of £3.3 million, ordinary dividend of £641,000 and rental		
			income of £78,000.		
F	0.5	-	Wolverhampton Grand Theatre Limited	0.4	-
			The Grand Theatre is managed by Wolverhampton Grand Theatre Limited. The		
			council continues to own the building and retains the right to appoint or remove the		
			majority of the members of the board of directors.		
J			The council provides grant funding to support the net cost of operating the theatre.		
)	1.2	(1.2)	Wolverhampton Schools' Improvement Partnership	1.6	(1.6)
5			The Wolverhampton Schools' Improvement Partnership is a company limited by		
_			guarantee established to advance educational opportunities and outcomes for children		
í			and young people in Wolverhampton.		
١			The board of directors comprises of representatives from each school cluster and		
F	0.0		phase and the senior substantive council officer for schools.	0.4	
	0.2	-	Lighthouse Media Centre	0.1	-
			The Lighthouse Media Centre is an independent company that develops and supports the creative industries. The council provides grant funding to support the net cost of		
			operating the centre. The Lighthouse Media Centre leases premises within the		
			Chubb Building which is owned by the council.		
-	5.6	(1.7)	i54	10.0	(0.7)
	0.0	()	The council is party to a joint venture with Staffordshire County Council for the		(0)
			development of i54 at Wobaston Road, Wolverhampton.		
ľ			Entities in which Councillors have declared an interest:		
Į	13.6	(0.4)	West Midlands Integrated Transport Authority	14.8	(3.6)
	6.7	(0.1)	West Midlands Police Authority	6.2	-
Ī	3.4	_	West Midlands Fire Service	3.9	-

20	12/13		2013/	14
Expenditure £m	Income £m	Entity and Nature of Relationship	Expenditure £m	Income £m
-	-	Advanced Childcare Limited	3.3	-
0.2	-	City of Wolverhampton College	0.6	-
0.4	-	Wolverhampton Citizens Advice Bureau	0.6	-
0.2	-	Mencap	0.3	-
-	-	Royal Mail	0.3	-
-	-	FBC Manby Bowdler LLP	0.3	-
0.2	(0.2)	North East Wolverhampton Academy (prev. Heath Park)	0.2	(0.2)
0.3	-	All Saints Action Network	0.2	-
_	-	Central Learning Partnership Trust	0.2	(0.3)
-	-	Mid Counties Co-op	0.1	-
υ <u></u>	-	Perry Hall Academy	-	(0.1)
	-	Woden Academy	-	(0.1)
	-	Our Lady and St Chad Catholic Sports College	-	(0.1)
-	-	Smestow School	-	(0.1)
10	-	Wolverhampton Girls High School	-	(0.1)
0.2	-	South Wolverhampton and Bilston Academy	-	-
0.1	-	Black Country Housing Group	-	-
0.1	-	Heath Town Senior Citizens Welfare Project	-	-

Note 9 - Trust Funds

Wolverhampton City Council acts as a trustee for a number of trust funds. The funds are not assets of the council and therefore they have not been included in the council's balance sheet. The table below provides an overview of the funds and their finances over the last two years.

		2012/13				2013/14		
	Income	Expenditure	Fund Value at 31 March 2013	Fund Name and Purpose	Income	Expenditure	Fund Value at 31 March 2014	
	£000	£000	£000		£000	£000	£000	
				Springfield Reading Room				
	-	-	42	In its capacity as trustee, the council is authorised to apply income in various ways.	1	1	42	
ט				Greenway Benefaction				
2 D 1	-	-	29	Established for the convalescence, enjoyment, pleasure and amusement of poor children of Bradley.		1	29	
25	_	_	16	Butler Bequest Music in the Parks	1	_	17	
	_		10	To provide music in the parks.	'	_	17	
	_	- 10		Monica Lloyd	_	_	10	
	10		10	To provide assistance with further education.			10	
	-	-	25	Other smaller funds	1	ı	26	
	-	-	122	Total	2	-	124	

Note 10 - Reconciliation of the Financial Statements to the Statutory Accounts

10A - Subjective Analysis of the Surplus or Deficit on the Provision of Services, and Reconciliation to Outturn for the Year

The table below shows the surplus or deficit on the provision of services, as shown in the group comprehensive income and expenditure statement, analysed by the categories of income and expenditure specified by the Code.

	2012/13 £m	Category	2013/14 £m
	(166.0)	Fees, charges and other service income	(247.7)
	(1.1)	Interest and investment income	(4.5)
	(93.7)	Income from council tax	(74.0)
ָדֶר בי	(630.7)	Government grants and contributions	(520.8)
age	294.3	Employee expenses	290.6
	515.7	Other service expenses	483.7
1 26	114.1	Depreciation, amortisation and impairment	63.1
<u>ත</u> [25.4	Interest payments	31.3
	13.4	Precepts and levies	13.3
	1.5	Payments to the national housing capital receipts pool	1.8
	27.6	(Gain) or loss on the disposal of non-current assets	61.9
	100.5	(Surplus) or deficit on the provision of services	98.7

The following table shows the adjustments needed to get from the surplus or deficit on the provision of services as shown in the Group Comprehensive Income and Expenditure statement, to the surplus or deficit for the year calculated in line with the legal requirements, as shown in the segmental analysis in Note 1.

2012/13		2013/14
£m		£m
	(Surplus) or deficit on the provision of services	98.7
(81.2)	Less: items included in the segmental analysis but not included in the Comprehensive Income and Expenditure Statement	(107.8)
(1.4)	Less: Surplus or (deficit) attributable to subsidiaries	(2.0)
9.1	(Surplus) or deficit for the year	(11.1)

The table below shows the adjustments needed to get from the net cost of services as shown in the Group Comprehensive Income and Expenditure statement, to the surplus or deficit for the year calculated in line with the legal requirements, as shown in the segmental analysis in Note 1.

107			2013/2014 £m
•		Net Cost of Services	269.0
	(305.0	Less: items included in the segmental analysis but not included in the Net Cost of Services	(278.1)
	(1.4) Less: Surplus or (deficit) attributable to subsidiaries		(2.0)
	17.9	(Surplus) or deficit for the year	(11.1)

10B – Detailed Analysis of Movement in Reserves Statement: 2013/14 Part 1 – Usable Reserves

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	HRA Earmarked Reserves	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves
	£m	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	(15.9)	(61.1)	(5.0)	-	(9.7)	(5.5)	(31.6)	(128.8)
Surplus or Deficit on Provision of Services	112.1	-	(15.4)	-	-	-	•	96.7
Other Comprehensive Income and Expenditure								
Revaluations - Gains and losses	-	-	-	-	-	-	-	-
Gains on Available-for-Sale Financial Assets	-	-	-		-	-	-	-
Amounts recycled (AFS Reserve) to the I&E Account after impairment	-	-	-	-	-	-	-	-
or derecognition Re-measurements in the pensions reserve								
Total Comprehensive Income and Expenditure	112.1	-	(15.4)		-		-	96.7
Total Comprehensive income and Expenditure	112.1	<u> </u>	(13.4)	<u>-</u>	-		-	30.7
Adjustments between Accounting Basis & Funding Basis under Regulations								
Depreciation, amortisation & impairment of non-current assets	(62.6)	-	(0.5)	-	(21.9)	1	-	(85.0)
Difference between HRA depreciation and Major Repairs Allowance	-	-	-	-	-	-	-	-
HRA Share of Contribution to Pension Reserve	=	T	-	-	-	1	T	-
Movement in the market value of Investment Properties	-	-	0.2	-	-	-	-	0.2
Revenue Expenditure Funded from Capital under Statute	(66.1)	-	-	-	-	-	-	(66.1)
Net Gain/Loss on sale of non-current assets (net book value of assets)	(67.3)	ı	(7.0)	-	-	1	ı	(74.3)
Net Gain/Loss on sale of non-current assets (disposal proceeds)	4.1	-	8.4	-	-	(12.5)	-	-
Difference between statutory debit/credit and amount recognised as income and expenditure in respect of financial instruments	(0.3)	-	-	-	-	-	-	(0.3)
Difference between amounts credited to the CIES and amounts to be recognised under statutory provisions relating to Council Tax	-	-	-	-	-	-	-	-

age 129

6. THE FINANCIAL STATEMENTS

2013/14 Part 1 – Usable Reserves (Continued)

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	HRA Earmarked Reserves	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves
	£m	£m	£m	£m	£m	£m	£m	£m
Net Charges made for retirement benefits in accordance with IAS 19	(48.3)	-	-	-	-	-	-	(48.3)
Capital Expenditure charged in the year to the General Fund	0.1	-	-	-	-	-	-	0.1
Transfer from UCR to meet payments to Housing Capital Receipts	(1.8)	-	-	-	-	1.8	-	-
Employer's contributions payable to the Pension Fund and retirement benefits paid directly to pensioners Transfers of HRA Balance	28.6	-	-	-	-	-	-	28.6
Reversal of financing of unequal pay back provision								-
Application of Capital Grants and Contributions to capital financing transferred to the CAA	-	-	-	-	-	-	99.5	99.5
Capital grants and contributions unapplied credited to CIES	102.0	-	-	-	-	-	(102.0)	-
Movement in the Donated Assets Account	-	-	-	-	-	-	-	-
Amount by which officer remuneration charged to the CIES on an accruals basis is different from remuneration chargeable in the year in accordance with statutory requirements	0.9	-	-	-	-	-	-	0.9
Capital Expenditure Financed from UCR	-	-	-	-	-	6.6	_	6.6
Other income that cannot be credited to the General Fund	0.3	-	-	-	-	-	-	0.3
Revenue provision for the repayment of debt	4.5	-	14.5	-	-	-	-	19.0
Transfer of HRA Settlement Receipts to UCR	-	-	-	-	-	-	-	-
Use of Major Repairs Allowance to Finance Capital Spend	-	-	-	-	18.4	-	-	18.4
Net Increase/Decrease before Transfers & Other Movements	6.2	-	0.2	-	(3.5)	(4.1)	(2.5)	(3.7)
Group contributions to/from Reserves (Group a/cs only)								
Subsidiary Entities	-	-	-	-	-	-	-	-
Other Associates & Joint Ventures			-	-	-	-	-	-
Transfers to/from other Reserves	-	-	-	-	-	-	-	-
Transfers to/from other Earmarked Reserves	(17.3)	17.5	(0.2)	-	-	-	-	-
Balance Carried Forward	(27.0)	(43.6)	(5.0)	_	(13.2)	(9.6)	(34.1)	(132.5)

2013/14 Part 2 – Unusable Reserves and Reserves of Subsidiary

	Short-term Accumulating Compensated Absences Account	Reserve	Capital Adjustment Account	Collection Fund Adjustment Account	Financial Instruments Adjustment Account	Pensions Reserve	Revaluation Reserve	Total Unusable Reserves	TOTAL (Council)	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	6.9	(12.1)	(575.4)	0.5	3.2	551.7	(214.4)	(239.6)	(368.4)	22.8	(345.6)
Surplus or Deficit on Provision of Services	-	-	-	-	-	-	-	-	96.7	2.0	98.7
Other Comprehensive Income and Expenditure											
Revaluations - Gains and losses	-	-	-	-	-	-	(6.8)	(6.8)	(6.8)	-	(6.8)
Gains on Available-for-Sale Financial Assets	-	-	-	-	-	-	-	-	-	-	-
Amounts recycled (AFS Reserve) to the I&E Account	_	_	_	_	_	_	_	_	_	_	_
after impairment or derecognition					_			_			_
Re-measurements in the pensions reserve	-	-	-	-	-	(109.6)	-	(109.6)	(109.6)	(16.5)	(126.1)
Total Comprehensive Income and Expenditure	-	-	-	-	-	(109.6)	(6.8)	(116.4)	(19.7)	(14.5)	(34.2)
Adjustments between Accounting Basis & Funding Basis under Regulations											
Depreciation, amortisation & impairment of non-current assets	-	-	78.3	-	-	-	6.7	85.0	-	-	-
Difference between HRA depreciation and Major Repairs Allowance	-	-	-	-	-	-	-	-	-	-	-
HRA Share of Contribution to Pension Reserve	-	-	-	-	-	-	-	-	-	-	-
Movement in the market value of Investment Properties	=	-	(0.2)	-	-	-	-	(0.2)	-	-	-
Revenue Expenditure Funded from Capital under Statute	-	-	66.1	-	-	-	-	66.1	-	-	-
Net Gain/Loss on sale of non-current assets (net book value of assets)	-	-	37.7	-	-	-	36.6	74.3	-	-	-
Net Gain/Loss on sale of non-current assets (disposal proceeds)	-	-	-	-	-	-	-	-	-	-	-
Difference between statutory debit/credit and amount recognised as income and expenditure in respect of financial instruments	-	-	-	-	0.3	-	-	0.3	-	-	-
Difference between amounts credited to the CIES and amounts to be recognised under statutory provisions relating to Council Tax	-	-	-	-	-	-	-	-	-	-	-

³age 130

2013/14 Part 2 – Unusable Reserves and Reserves of Subsidiary (Continued)

	Short-term Accumulating Compensated Absences Account	Available-for- sale Financial Instruments Reserve	Capital Adjustment Account	Collection Fund Adjustment Account	Financial Instruments Adjustment Account	Pensions Reserve	Revaluation Reserve	Total Unusable Reserves	TOTAL (Council)	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Net Charges made for retirement benefits in accordance with IAS 19	-	-	-	-	-	48.3	1	48.3	•	-	-
Capital Expenditure charged in the year to the General Fund	-	-	(0.1)	-	-	-	-	(0.1)	-	-	-
Transfer from UCR to meet payments to Housing Capital Receipts Pool	-	-	-	-	-	-	-	-	-	-	-
Employer's contributions payable to the Pension Fund and retirement benefits paid directly to pensioners	-	-	-	-	-	(28.6)	-	(28.6)	-	-	-
Transfers of HRA Balance	-	-	-	-	-	-	-	-	-	-	-
Reversal of financing of unequal pay back provision	-	-	-	-	-	-	-	-	-	-	-
Application of Capital Grants and Contributions to capital financing transferred to the CAA	-	-	(99.5)	-	-	-	-	(99.5)	-	-	-
Capital grants and contributions unapplied credited to CIES	-	-	-	-	-	-	-	-	-	-	-
Movement in the Donated Assets Account	-	-	-	-	-	-	-	-	-	-	-
Amount by which officer remuneration charged to the CIES on an accruals basis is different from remuneration chargeable in the year in accordance with statutory requirements	(0.9)	-	-	-	-	-	-	(0.9)	-	-	-
Capital Expenditure Financed from UCR	-	-	(6.6)	-	-	-	-	(6.6)	-	-	-
Other income that cannot be credited to the General Fund	-	-	(0.3)	-	-	-	-	(0.3)	-	-	-
Revenue provision for the repayment of debt	-	-	(18.9)	-	-	-	-	(18.9)	0.1	-	0.1
Transfer of HRA Settlement Receipts to UCR	-	-	-	-	-	-	-	•	-	-	-
Use of Major Repairs Allowance to Finance Capital Spend	1	-	(18.4)	-	-	1	ı	(18.4)	•	-	-
Net Increase/Decrease before Transfers & Other Movements	(0.9)	-	38.1	-	0.3	(89.9)	36.5	(15.9)	(19.6)	(14.5)	(34.1)
Group contributions to/from Reserves (Group a/cs only)											
Subsidiary Entities	-	-	-	-	-	-	-	-	-	-	-
Other Associates & Joint Ventures	-	-	-	-	-	-	-	-	-	-	-
Transfers to/from other Reserves	-	-	-	-	-	-	-	-	-	-	-
Transfers to/from other Earmarked Reserves	-	-	-	-	-	-	-	-	-	-	-
Balance Carried Forward	6.0	(12.1)	(537.3)	0.5	3.5	461.8	(177.9)	(255.5)	(388.0)	8.3	(379.7)

6. THE FINANCIAL STATEMENTS

²age 131

age 132

6. THE FINANCIAL STATEMENTS

2012/13 Part 1 - Usable Reserves

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves
	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	(19.5)	(83.6)	(10.5)	(5.2)	(2.3)	(36.1)	(157.2)
Surplus or Deficit on Provision of Services	99.6	-	(9.3)	-	<u>-</u>	-	90.3
Other Comprehensive Income and Expenditure							
Revaluations - Gains and losses	-	-	-	-	-	-	-
Gains on Available-for-Sale Financial Assets	-	-	-	-	=	-	-
Re-measurements in the pensions reserve	=	-	-	-	-	-	-
Total Comprehensive Income and Expenditure	99.6	-	(9.3)	-	-		90.3
·			` '				
Adjustments between Accounting Basis & Funding Basis under							
Regulations							
Depreciation, amortisation & impairment of non-current assets	(114.1)	-		(22.0)	-	-	(136.1)
HRA Share of Contribution to Pension Reserve	-	-	-	-	-	-	-
Movement in the market value of Investment Properties	-	-	-	-	-	-	-
Revenue Expenditure Funded from Capital under Statute	(61.7)	-	-	-	-	-	(61.7)
Net Gain/Loss on sale of non-current assets (net book value of assets)	(32.1)	-	(4.8)	-	-	-	(36.9)
Net Gain/Loss on sale of non-current assets (disposal proceeds)	4.0	-	5.5	-	(9.0)	-	0.5
Difference between statutory debit/credit and amount recognised as income and expenditure in respect of financial instruments	(0.3)	-	-	-	-	-	(0.3)
Difference between amounts credited to the CIES and amounts to be recognised under statutory provisions relating to Council Tax	(0.4)	-	-	-	-	-	(0.4)

2012/13 Part 1 – Usable Reserves (Continued)

	General Fund Balance	General Fund Earmarked Reserves	HRA Balance	Major Repairs Reserve	Usable Capital Receipts Reserve	Capital Grants Unapplied Account	Total Usable Reserves
	£m	£m	£m	£m	£m	£m	£m
Net Charges made for retirement benefits in accordance with IAS 19	(33.6)	-	-	-	-	-	(33.6)
Capital Expenditure charged in the year to the General Fund	0.2	-	-	-	-	-	0.2
Transfer from UCR to meet payments to Housing Capital Receipts Pool	(1.5)	-	-	-	1.5	-	-
Employer's contributions payable to the Pension Fund and retirement benefits paid directly to pensioners	27.2	-		_		-	27.2
Transfers of HRA Balance							-
Reversal of financing of unequal pay back provision							ı
Application of Capital Grants and Contributions to capital financing transferred to the CAA	-	-	-	-	-	117.9	117.9
Capital grants and contributions unapplied credited to CIES	113.4	-	-	-	1	(113.4)	-
Movement in the Donated Assets Account							•
Amount by which officer remuneration charged to the CIES on an accruals basis is different from remuneration chargeable in the year in accordance with statutory requirements	(0.7)	-	-	-	1	1	(0.7)
Capital Expenditure Financed from UCR	-	-	-	-	4.3	-	4.3
Other income that cannot be credited to the General Fund	0.3	-	-	-	-	-	0.3
Revenue provision for the repayment of debt	25.6	-	14.3	-	-	-	39.9
Transfer of HRA Settlement Receipts to UCR	-	-	-	- 1	-	_	-
Use of Major Repairs Allowance to Finance Capital Spend	-	-	-	17.5	-	-	17.5
Net Increase/Decrease before Transfers & Other Movements	25.9	-	5.7	(4.5)	(3.2)	4.5	28.4
Transfers to/from other Earmarked Reserves	(22.3)	22.5	(0.2)	-	-	-	
			,				
Balance Carried Forward	(15.9)	(61.1)	(5.0)	(9.7)	(5.5)	(31.6)	(128.8)

Jage 133

2012/13 Part 2 – Unusable Reserves and Reserves of Subsidiary

	Short-term Accumulating Compensated Absences Account	Available-for- sale Financial Instruments Reserve	Capital Adjustment Account	Collection Fund Adjustment Account	Financial Instruments Adjustment Account	Pensions Reserve	Revaluation Reserve	Total Unusable Reserves	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Balance Brought Forward	6.2	(12.1)	(603.8)	0.1	2.9	443.4	(275.2)	(438.5)	12.6	(583.1)
Surplus or Deficit on Provision of Services	-	-	-	-	-	•	-	-	1.4	91.7
Other Comprehensive Income and Expenditure										
Revaluations - Gains and losses	-	-	-	-	-	-	35.1	35.1	-	35.1
Gains on Available-for-Sale Financial Assets	-	-	-	-	-	-	-	-	-	-
Actuarial Gain/Loss in the pensions reserve	-	-	-	-	-	101.9	-	101.9	8.8	110.7
Total Comprehensive Income and Expenditure	-	-	-	-	-	101.9	35.1	137.0	10.2	237.5
Adjustments between Accounting Basis & Funding Basis under Regulations										
Depreciation, amortisation & impairment of non-current assets	-	-	121.9	-	-	-	14.2	136.1	-	-
HRA Share of Contribution to Pension Reserve	-	-	-	1	-	-	-	-	-	-
Movement in the market value of Investment Properties	-	-	-	-	-	-	-	-	-	-
Revenue Expenditure Funded from Capital under Statute	-	-	61.7	-	-	-	-	61.7	-	-
Net Gain/Loss on sale of non-current assets (net book value of assets)	-	-	25.4	-	-	-	11.5	36.9	-	-
Net Gain/Loss on sale of non-current assets (disposal proceeds)	-	-	(0.5)	-	-	-	-	(0.5)	-	-
Difference between statutory debit/credit and amount recognised as income and expenditure in respect of financial instruments	-	-	-	-	0.3	-	-	0.3	-	-
Difference between amounts credited to the CIES and amounts to be recognised under statutory provisions relating to Council Tax	-	-	-	0.4	-	-	-	0.4	-	-

-(

6. THE FINANCIAL STATEMENTS

2012/13 Part 2 – Unusable Reserves and Reserves of Subsidiary (Continued)

			•							
	Short-term Accumulating Compensated Absences Account		Capital Adjustment Account	Collection Fund Adjustment Account	Financial Instruments Adjustment Account	Pensions Reserve	Revaluation Reserve	Total Unusable Reserves	Reserves of Subsidiary	TOTAL (Group)
	£m	£m	£m	£m	£m	£m	£m	£m	£m	£m
Net Charges made for retirement benefits in										
accordance with IAS 19	-	-	-	-	-	33.6	-	33.6	-	-
Capital Expenditure charged in the year to the General			(2.0)					(2.0)		
Fund	-	-	(0.2)	-	-	-	-	(0.2)	-	-
Transfer from UCR to meet payments to Housing										
Capital Receipts Pool	-	-	-	-	-	-	-	-	-	-
Employer's contributions payable to the Pension Fund						(07.0)		(27.0)		
and retirement benefits paid directly to pensioners	_	-	-	-	-	(27.2)	-	(27.2)	-	-
Transfers of HRA Balance								-		-
Reversal of financing of unequal pay back provision								-		-
Application of Capital Grants and Contributions to			(117.9)					(117.9)		
capital financing transferred to the CAA	-	-	(117.9)	-		ı	-	(117.9)	-	-
Capital grants and contributions unapplied credited to										
CIES		_			_		_	_	-	<u>-</u>
Movement in the Donated Assets Account								-		-
Amount by which officer remuneration charged to the										
CIES on an accruals basis is different from	0.7	_	_	_	_	_	_	0.7	_	_
remuneration chargeable in the year in accordance	0.7							0.7		
with statutory requirements										
Capital Expenditure Financed from UCR	-	-	(4.3)	-	-	-	-	(4.3)	-	-
Other income that cannot be credited to the General	_	_	(0.3)	_	_	_	_	(0.3)	_	_
Fund			` '					` '		
Revenue provision for the repayment of debt	-	-	(39.9)	-	-	-	-	(39.9)	-	-
Transfer of HRA Settlement Receipts to UCR	-	-	-	-	-	-	-	-	-	-
Use of Major Repairs Allowance to Finance Capital	_	_	(17.5)	_	_	_	_	(17.5)	_	-
Spend			(1710)					(,		
Net Increase/Decrease before Transfers & Other	0.7	_	28.4	0.4	0.3	108.3	60.8	198.9	10.2	237.5
Movements	+									
Transferre to the fire weather a Ferres and a different	1									
Transfers to/from other Earmarked Reserves								-	-	-
Polomos Comind Formand		(40.4)	(E7E 4)	^-	2.0	FF4 7	(24.4.4)	(222.0)	22.0	(245.0)
Balance Carried Forward	6.9	(12.1)	(575.4)	0.5	3.2	551.7	(214.4)	(239.6)	22.8	(345.6)

Name of Reserve	Description
Usable Reserves	
Revenue	
General Fund Balance	The General Fund is the statutory fund into which all the receipts of an authority are required to be paid, and out of which all liabilities of the authority are to be met, except to the extent that statutory rules might provide otherwise. These rules can also specify the financial year in which liabilities and payments should impact on the General Fund Balance, which is not necessarily in accordance with proper accounting practice. The General Fund Balance therefore summarises the resources that the council is statutorily empowered to spend on its services or on capital investment (or the deficit of resources that the council is required to recover) at the end of the financial year (however, the balance is not available to be applied to funding HRA services: see Housing Revenue Account Balance below).
Housing Revenue Account Balance	The Housing Revenue Account Balance reflects the statutory obligation to maintain a revenue account for local authority council housing provision in accordance with Part VI of the Local Government and Housing Act 1989. It contains the balance of income and expenditure as defined by the 1989 Act that is available to fund future expenditure in connection with the council's landlord function or (where in deficit) that is required to be recovered from tenants in future years.
Earmarked Reserves	Earmarked Reserves represent amounts that the council has chosen to set aside to fund specific items of expenditure in the future. The most significant earmarked reserves are the Efficiency Reserve (£5.4 million), the Insurance Reserve (£3.3 million), the Job Evaluation Reserve (£2.6 million) and the FutureWorks Reserve (£2.4 million).
Capital	
Major Repairs Reserve	The council is required to maintain the Major Repairs Reserve, which controls the application of the Major Repairs Allowance (MRA). The MRA is restricted to being applied to new capital investment in HRA assets or the financing of historical capital expenditure by the HRA. The balance shows the MRA that has yet to be applied at the year end.
Capital Receipts Reserve	The Capital Receipts Reserve holds the proceeds from the disposal of land or other assets, which are restricted by statute from being used other than to fund new capital expenditure or to be set aside to finance historical capital expenditure. The balance on the reserve shows the resources that have yet to be applied for these purposes at the year end.
Capital Grants Unapplied Account	The Capital Grants Unapplied Account (Reserve) holds the grants and contributions received towards capital projects for which the council has met the conditions that would otherwise require repayment of the monies but

	Name of Reserve	Description
		which have yet to be applied to meet expenditure. The balance is restricted by grant terms as to the capital
		expenditure against which it can be applied and/or the financial year in which this can take place.
	Unusable Reserves	
	Revaluation Reserve	The Revaluation Reserve contains the gains made by the council arising from increases in the value of its Property, Plant and Equipment. The balance is reduced when assets with accumulated gains are revalued downwards or impaired and the gains are lost, used in the provision of services and the gains are consumed through depreciation, or disposed of and the gains are realised. The reserve only contains revaluation gains accumulated since 1 April 2007, the date that the reserve was created. Accumulated gains arising before that date are consolidated into the balance on the Capital Adjustment Account.
	Available-for-Sale Financial Instruments Reserve	The Available-for-Sale Financial Instruments Reserve contains the gains made by the council arising from increases in the value of its investments that have quoted market prices or otherwise do not have fixed or determinable payments. The balance is reduced when investments with accumulated gains are revalued
Dogo 137	Capital Adjustment Account	downwards or impaired and the gains are lost, or disposed of and the gains are realised. The Capital Adjustment Account absorbs the timing differences arising from the different arrangements for accounting for the consumption of non-current assets and for financing the acquisition, construction or enhancement of those assets under statutory provisions. The account is debited with the cost of acquisition, construction or enhancement as depreciation, impairment losses and amortisations are charged to the Comprehensive Income and Expenditure Statement (with reconciling postings from the Revaluation Reserve to convert fair value figures to a historical cost basis). The account is credited with the amounts set aside by the council as finance for the costs of acquisition, construction and enhancement. The account contains accumulated gains and losses on Investment Properties and gains recognised on donated assets that have yet to be consumed by the council. It also contains revaluation gains accumulated on Property, Plant and Equipment before 1 April 2007, the date that the Revaluation Reserve was created to hold such gains.
	Financial Instruments Adjustment Account	The Financial Instruments Adjustment Account absorbs the timing differences arising from the different arrangements for accounting for income and expenses relating to certain financial instruments and for bearing losses or benefiting from gains per statutory provisions.
	Pensions Reserve	The Pensions Reserve absorbs the timing differences arising from the different arrangements for accounting for post-employment benefits and for funding benefits in accordance with statutory provisions. The council accounts for post-employment benefits in the Comprehensive Income and Expenditure Statement as the benefits are earned by employees accruing years of service, updating the liabilities recognised to reflect inflation, changing assumptions and investment returns on any resources set aside to meet the costs. However, statutory arrangements require benefits earned to be financed as the council makes employer's contributions to pension funds or eventually pays any pensions for which it is directly responsible. The debit balance on the Pensions Reserve therefore shows a substantial shortfall in the benefits earned by past and current employees

Name of Reserve	Description
	and the resources the council has set aside to meet them. The statutory arrangements will ensure that funding
	will have been set aside by the time the benefits come to be paid.
Collection Fund Adjustment	The Collection Fund Adjustment Account manages the differences arising from the recognition of council tax
Account	and business rates income in the Comprehensive Income and Expenditure Statement as it falls due from
	council tax and business rates payers compared with the statutory arrangements for paying across amounts to
	the General Fund from the Collection Fund.
Accumulated Absences Account	The Accumulated Absences Account absorbs the differences that would otherwise arise on the General Fund
	Balance from accruing for compensated absences earned but not taken in the year, for example annual leave
	entitlement not yet used at 31 March. Statutory arrangements require that the impact on the General Fund
	Balance is neutralised by transfers to or from the account.

Note 11 - Accounting Policies

Note 11A – Accounting Policies Applying to these Statements

1. General Principles

The Statement of Accounts summarises the council's transactions for the 2013/14 financial year and its position at 31 March 2014. The council is required to prepare an annual Statement of Accounts by the Accounts and Audit (England) Regulations 2011, which those regulations require to be prepared in accordance with proper accounting practices. These practices primarily comprise the Code of Practice on Local Authority Accounting in the United Kingdom 2013/14 and the Service Reporting Code of Practice 2013/14, supported by International Financial Reporting Standards (IFRS).

The accounting convention adopted in the Statement of Accounts is principally historical cost, modified by the revaluation of certain categories of non-current assets and financial instruments.

2. Accruals of Income and Expenditure

Activity is accounted for in the year that it takes place, not simply when cash payments are made or received. In particular:

- Revenue from the sale of goods is recognised when the council transfers the significant risks and rewards of ownership to the purchaser and it is probable that economic benefits or service potential associated with the transaction will flow to the council.
- Revenue from the provision of services is recognised when the council can measure reliably the percentage of completion of the transaction and it is probable that economic benefits or service potential associated with the transaction will flow to the council.
- Supplies are recorded as expenditure when they are consumed. Where there is a gap between the date supplies are received and their consumption they are carried as inventories on the Balance Sheet.
- Expenses in relation to services received (including services provided by employees) are recorded as expenditure when the services are received rather than when payments are made.

- Interest receivable on investments and payable on borrowings is accounted for respectively as income and expenditure on the basis of the effective interest rate for the relevant financial instrument rather than necessarily the cash flows fixed or determined by the contract.
- Where revenue and expenditure has been recognised but cash has not been received or paid, a debtor or creditor for the relevant amount is recorded in the Balance Sheet. Where debts may not be settled, the balance of debtors is written down and a charge made to revenue for the income that might not be collected.
- Manual Accruals are only processed for amounts of £10,000 or more. Except where the expenditure is by schools or funded directly from external grants.

3. Cash and Cash Equivalents

Cash is represented by cash in hand and deposits with financial institutions, repayable without penalty on notice of not more than 24 hours. Cash equivalents are highly liquid investments that mature in three months or less from the date of acquisition and that are readily convertible to known amounts of cash with insignificant risk of change in value.

→ 4. Exceptional Items

When items of income and expense are material, their nature and amount is disclosed separately, either on the face of the Comprehensive Income and Expenditure Statement or in the notes to the accounts, depending on how significant the items are to an understanding of the council's financial performance.

5. Prior Period Adjustments, Changes in Accounting Policies and Estimates and Errors

Prior period adjustments may arise as a result of a change in accounting policies or to correct a material error. Changes in accounting estimates are accounted for prospectively, i.e. in the current and future years affected by the change and do not give rise to a prior period adjustment.

Changes in accounting policies are only made when required by proper accounting practices or the change provides more reliable or relevant information about the effect of transactions, other events and conditions on the council's financial position or financial performance. Where a change is made, it is applied retrospectively (unless stated otherwise) by adjusting opening balances and comparative amounts for the prior period as if the new policy had always been applied.

Material errors discovered in prior period figures are corrected retrospectively by amending opening balances and comparative amounts for the prior period.

6. Charges to Revenue for Non-Current Assets

Services, support services and trading accounts are charged with the following amounts to record the cost of holding non-current assets during the year:

- Depreciation attributable to the assets used by the relevant service;
- Revaluation and impairment losses on assets used by the service where there are no accumulated gains in the Revaluation Reserve against which the losses can be written off;
- Amortisation of intangible assets attributable to the service.

The council is not required to raise council tax to fund depreciation, revaluation and impairment losses or amortisations. However, it is required to make an annual contribution from revenue towards the reduction in its overall borrowing requirement (calculated on a prudent basis determined by the council in accordance with statutory guidance). Depreciation, revaluation and impairment losses and amortisations are therefore replaced by the contribution in the General Fund Balance (known as Minimum Revenue Provision), by way of an adjusting transaction with the Capital Adjustment Account in the Movement in Reserves Statement.

₽ 7. Employee Benefits

Benefits Payable During Employment - Short-term employee benefits are those due to be settled in their entirety within 12 months of the year end. They include such benefits as wages and salaries, paid annual leave and paid sick leave, bonuses and non-monetary benefits for current employees and are recognised as an expense for services in the year in which employees render service to the council. An accrual is made for the cost of holiday entitlements (and any other form of leave) earned by employees but not taken before the year end, and which employees can carry forward into the next financial year. The accrual is made at the salary rates applicable in the following accounting year, being the period in which the employee takes the benefit. The accrual is charged to Surplus or Deficit on the Provision of Services, but then reversed out through the Movement in Reserves Statement in the Accumulated Absences Account so that holiday benefits are charged to revenue in the financial year in which the holiday absence occurs.

Termination Benefits - Termination benefits are amounts payable as a result of a decision by the council to terminate an officer's employment before the normal retirement date or an officer's decision to accept voluntary redundancy in exchange for those benefits, and are charged on an accruals basis to the Non-Distributed Costs line in the Comprehensive Income and Expenditure Statement at the earlier of the point at which the authority can no longer withdraw the offer of those benefits or when the authority recognises the costs of the restructure.

Where termination benefits involve the enhancement of pensions, statutory provisions require the General Fund Balance to be charged with the amount payable by the council to the pension fund or pensioner in the year, not the amount calculated according to the relevant accounting standards. In the Movement in Reserves Statement, appropriations are required to and from the Pensions Reserve to remove the notional debits and credits for pension enhancement termination benefits and replace them with debits for the cash paid to the pension fund and pensioners and any such amounts payable but unpaid at the year end.

Post-Employment Benefits - Employees of the council are members of two separate pension schemes:

- The Teachers' Pension Scheme, administered by Capita Teachers' Pensions on behalf of the Department for Education;
- The Local Government Pensions Scheme, administered by West Midlands Pension Fund.

Both schemes provided defined benefits to members (retirement lump sums and pensions), earned as employees worked for the council. However, the arrangements for the teachers' scheme mean that liabilities for these benefits cannot ordinarily be identified specifically to the council. The scheme is therefore accounted for as if it were a defined contribution scheme and no liability for future payments of benefits is recognised in the Balance Sheet. The Children's and Education Services line in the Comprehensive Income and Expenditure Statement is charged with the employer's contributions payable to Teachers' Pension Scheme in the year.

The Local Government Pension Scheme

The Local Government Pension Scheme is accounted for as a defined benefits scheme. The liabilities of the West Midlands Pension Fund attributable to the council are included in the Balance Sheet on an actuarial basis using the projected unit method i.e. an assessment of the future payments that will be made in relation to the retirement benefits earned to date by employees, based on assumptions about mortality rates, employee turnover rates, etc. and projections of earnings for current employees.

Liabilities are discounted to their value at current prices, using a discount rate based on the redemption yields on AA-rated corporate bonds with a term corresponding to the term of the liabilities. The assets of West Midlands Pension Fund attributable to the council are included in the Balance Sheet at their fair value, which varies depending on the type of asset:

- Quoted securities current bid price;
- Unquoted securities professional estimate;
- Unitised securities current bid price;
- Property market value.

The change in the net pensions liability is analysed into the following components:

Service cost comprising:

- Current service cost the increase in liabilities as a result of years of service earned this year allocated in the Comprehensive Income and Expenditure Statement to the services for which the employees worked;
- Past service cost the increase in liabilities arising from current year decisions whose effect relates to years of service earned in earlier years –
 debited to the Surplus or Deficit on the Provisions of Services in the Comprehensive Income and Expenditure Statement as part of NonDistributed Costs;
- Net interest on the net defined benefit liability / asset, i.e. net interest expense for the authority the change during the period in the net defined benefit liability / asset that arises from the passage of time charged to the Financing and Investment Income and Expenditure line of the Comprehensive Income and Expenditure Statement this is calculated by applying the discount rate used to measure the defined benefit obligation at the beginning of the period to the net defined benefit liability / asset at the beginning of the period taking into account any changes in the net defined benefit liability / asset during the period as a result of contribution and benefit payments.

Changes in valuations comprising:

• The return on plan assets – excluding amounts included in net interest on the net defined benefit liability / asset – charged to the Pensions Reserve as Other Comprehensive Income and Expenditure;

Contributions paid to the West Midlands Pension Fund – cash paid as the employer's contributions to the pension fund in settlement of liabilities, not accounted for as an expense.

In relation to retirement benefits, statutory provisions require the General Fund balance to be charged with the amount payable by the council to the pension fund or directly to pensioners in the year, not the amount calculated according to the relevant accounting standards. In the Movement in Reserves Statement, this means that there are appropriations to and from the Pensions Reserve to remove the notional debits and credits for retirement benefits and replace them with debits for the cash paid to the pension fund and pensioners, and any such amounts payable but unpaid at the year end. The negative balance that arises on the Pensions Reserve thereby measures the beneficial impact to the General Fund of being required to account for retirement benefits on the basis of cash flows rather than as benefits are earned by employees.

→ Discretionary Benefits - The council has restricted powers to make discretionary awards of retirement benefits in the event of early retirements. Any liabilities estimated to arise as a result of an award to any member of staff (including teachers) are accrued in the year of the decision to make the award and accounted for using the same policies as are applied to the Local Government Pension Scheme.

8. Events after the Reporting Period

Events after the Balance Sheet date are those events, both favourable and unfavourable, that occur between the end of the reporting period and the date when the Statement of Accounts is authorised for issue. Two types of event can be identified:

- Those that provide evidence of conditions that existed at the end of the reporting period the Statement of Accounts is adjusted to reflect such events;
- Those that are indicative of conditions that arose after the reporting period the Statement of Accounts is not adjusted to reflect such events, but where a category of events would have a material effect, disclosure is made in the notes of the nature of the events and their estimated financial effect.

Events taking place after the date of authorisation for issue are not reflected in the Statement of Accounts.

9. Financial Instruments

Financial Liabilities - Financial liabilities are recognised on the Balance Sheet when the council becomes a party to the contractual provisions of a financial instrument and are initially measured at fair value and are carried at their amortised cost. Annual charges to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement for interest payable are based on the carrying amount of the liability, multiplied by the effective rate of interest for the instrument. The effective interest rate is the rate that exactly discounts estimated future cash payments over the life of the instrument to the amount at which it was originally recognised.

For most of the borrowing that the council has, this means that the amount presented in the Balance Sheet is the outstanding principal repayable (plus accrued interest), and interest charged to the Comprehensive Income and Expenditure Statement is the amount payable for the year according to the loan agreement. Gains and losses on the repurchase or early settlement of borrowing are credited and debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement in the year of repurchase / settlement.

Where premiums and discounts have been charged to the Comprehensive Income and Expenditure Statement, regulations allow the impact on the General Fund Balance to be spread over future years. The council has a policy of spreading the gain or loss over the term that was remaining on the Doan against which the premium was payable or discount receivable when it was repaid. The reconciliation of amounts charged to the Comprehensive Income and Expenditure Statement to the net charge required against the General Fund Balance is managed by a transfer to or from the Financial Instruments Adjustment Account in the Movement in Reserves Statement.

Financial Assets - Financial assets are classified into two types:

- Loans and receivables assets that have fixed or determinable payments but are not quoted in an active market;
- Available-for-sale assets assets that have a quoted market price and / or do not have fixed or determinable payments.

Loans and Receivables - Loans and receivables are recognised on the Balance Sheet when the council becomes a party to the contractual provisions of a financial instrument and are initially measured at fair value. They are subsequently measured at their amortised cost. Annual credits to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement for interest receivable are based on the carrying amount of the asset multiplied by the effective rate of interest for the instrument. For loans the council has made the amount presented in the Balance Sheet is the outstanding principal receivable (plus accrued interest) and interest credited to the Comprehensive Income and Expenditure Statement is the amount receivable for the year in the loan agreement.

Page 146

6. THE FINANCIAL STATEMENTS

Where assets are identified as impaired because of a likelihood arising from a past event that payments due under the contract will not be made, the asset is written down and a charge made to the relevant service (for receivables specific to that service) or the Financing and Investments Income and Expenditure line in the Comprehensive Income and Expenditure Statement. The impairment loss is measured as the difference between the carrying amount and the present value of the revised future cash flows discounted at the asset's original effective interest rate. Any gains and losses that arise on the derecognition of the asset are credited or debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement.

Available-for-Sale Assets - Available-for-sale assets are recognised on the Balance Sheet when the council becomes a party to the contractual provisions of a financial instrument and are initially measured and carried at fair value. Where the asset has fixed or determinable payments, annual credits to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement for interest receivable are based on the amortised cost of the asset multiplied by the effective rate of interest for the instrument. Where there are no fixed or determinable payments, income (e.g. dividends) is credited to the Comprehensive Income and Expenditure Statement when it becomes receivable by the council.

Assets are maintained in the Balance Sheet at fair value. Values are based on the following principles:

- Instruments with quoted market prices the market price;
- Other instruments with fixed and determinable payments discounted cash flow analysis;
- Equity shares with no quoted market prices independent appraisal of company valuations.

Changes in fair value are balanced by an entry in the Available-for-Sale Reserve and the gain/loss is recognised in the Surplus or Deficit on Revaluation of Available-for-Sale Financial Assets. The exception is where impairment losses have been incurred - these are debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement, along with any net gain or loss for the asset accumulated in the Available-for-Sale Reserve.

Where assets are identified as impaired because of a likelihood arising from a past event that payments due under the contract will not be made (fixed or determinable payments) or fair value falls below cost, the asset is written down and a charge made to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement. If the asset has fixed or determinable payments, the impairment loss is measured as the difference between the carrying amount and the present value of the revised future cash flows discounted at the asset's original

³age 147

6. THE FINANCIAL STATEMENTS

effective interest rate. Otherwise, the impairment loss is measured as any shortfall of fair value against the acquisition cost of the instrument (net of any principal repayment and amortisation).

Any gains and losses that arise on the derecognition of the asset are credited or debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement, along with any accumulated gains or losses previously recognised in the Available-for-Sale Reserve. Where fair value cannot be measured reliably, the instrument is carried at cost (less any impairment losses).

Offsetting Financial Assets and Liabilities - A financial asset and a financial liability shall be offset and the net amount presented in the Balance Sheet when the council has a legally enforceable right to set off the recognised amounts, and intends either to settle on a net basis, or to realise the asset and settle the liability simultaneously.

10. Foreign Currency Translation

Where the council entered into a transaction denominated in a foreign currency, the transaction was converted into sterling at the exchange rate applicable on the date the transaction was effective.

11. Government Grants and Contributions

Whether paid on account, by instalments or in arrears, government grants and third party contributions and donations are recognised as due to the council when there is reasonable assurance that the council will comply with the conditions attached to the payments, and the grants or contributions will be received.

Amounts recognised as due to the council are not credited to the Comprehensive Income and Expenditure Statement until conditions attached to the grant or contribution have been satisfied. Conditions are stipulations that specify that the future economic benefits or service potential embodied in the asset received in the form of the grant or contribution are required to be consumed by the recipient as specified, or future economic benefits or service potential must be returned to the grant issuing body.

Monies advanced as grants and contributions for which conditions have not been satisfied are carried in the Balance Sheet as creditors. When conditions are satisfied, the grant or contribution is credited to the relevant service line (attributable revenue grants and contributions) or Taxation and Non-Specific Grant Income (non-ring-fenced revenue grants and all capital grants) in the Comprehensive Income and Expenditure Statement.

Where capital grants are credited to the Comprehensive Income and Expenditure Statement, they are reversed out of the General Fund Balance in the Movement in Reserves Statement. Where the grant has yet to be used to finance capital expenditure, it is posted to the Capital Grants Unapplied

Reserve. Where it has been applied, it is posted to the Capital Adjustment Account. Amounts in the Capital Grants Unapplied Reserve are transferred to the Capital Adjustment Account once they have been applied to fund capital expenditure.

12. Intangible Assets

Expenditure on non-monetary assets that do not have physical substance but are controlled by the council as a result of past events (e.g. software licences) is capitalised when it is expected that future economic benefits or service potential will flow from the intangible asset to the council. Expenditure on the development of websites would not be capitalised if the website were solely or primarily intended to promote or advertise the council's goods or services.

Intangible assets are measured initially at cost. Amounts are only revalued where the fair value of the assets held by the council can be determined by reference to an active market. The depreciable amount of an intangible asset is amortised over its useful life to the relevant service line(s) in the Comprehensive Income and Expenditure Statement. An asset is tested for impairment whenever there is an indication that the asset might be impaired - any losses recognised are posted to the relevant service line(s) in the Comprehensive Income and Expenditure Statement. Any gain or loss arising on the disposal or abandonment of an intangible asset is posted to the Other Operating Expenditure line in the Comprehensive Income and Expenditure Statement.

Where expenditure on intangible assets qualifies as capital expenditure for statutory purposes, amortisation, impairment losses and disposal gains and losses are not permitted to have an impact on the General Fund Balance. The gains and losses are therefore reversed out of the General Fund Balance in the Movement in Reverses Statement and posted to the Capital Adjustment Account and, for any sale proceeds greater than £10,000, the Capital Receipts Reserve.

13. Interests in Companies and Other Entities

The council is the sole owner of a company where its interest has the nature of a subsidiary (Wolverhampton Homes), which requires it to prepare group accounts. It has no other material interests in companies or other entities that have the nature of a subsidiary, associate or jointly-controlled entity.

14. Inventories

Inventories are included on the Balance Sheet at the lower of cost and net realisable value. The cost of inventories is assigned using the weighted average costing formula.

15. Investment Property

Investment properties are those that are used solely to earn rentals and/or for capital appreciation. The definition is not met if the property is used in any way to facilitate the delivery of services or production of goods, or is held for sale.

Investment properties are measured initially at cost and subsequently at fair value, based on the amount at which the asset could be exchanged between knowledgeable parties at arm's-length. Properties are not depreciated but are revalued according to market conditions at the year-end. Gains and losses on revaluation are posted to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement. The same treatment is applied to gains and losses on disposal.

Rentals received in relation to investment properties are credited to the Financing and Investment Income and Expenditure line and result in a gain for the General Fund Balance. However, revaluation and disposal gains and losses are not permitted by statutory arrangements to have an impact on the General Fund Balance. The gains and losses are therefore reversed out of the General Fund Balance in the Movement in Reserves Statement and posted to the Capital Adjustment Account and (for any sale proceeds greater than £10,000) the Capital Receipts Reserve.

🛡 16. Jointly Controlled Operations and Jointly Controlled Assets

Jointly controlled operations are activities undertaken by the council in conjunction with other ventures that involve the use of the assets and resources of the ventures rather than the establishment of a separate entity. The council recognises on its Balance Sheet the assets that it controls and the liabilities that it incurs, and debits and credits the Comprehensive Income and Expenditure Statement with the expenditure it incurs and the share of income it earns from the activity of the operation.

Jointly controlled assets are items of property, plant or equipment that are jointly controlled by the council and other ventures, with the assets being used to obtain benefits for the ventures. The joint venture does not involve the establishment of a separate entity. The council accounts for only its share of the jointly controlled assets, the liabilities and expenses that it incurs on its own behalf or jointly with others in respect of its interest in the joint venture and income that it earns from the venture.

17. Leases

Leases are classified as finance leases where the terms of the lease transfer substantially all the risks and rewards incidental to ownership of the property, plant or equipment from the lessor to the lessee. All other leases are classified as operating leases. Where a lease covers both land and buildings, the land and buildings elements are considered separately for classification.

50

6. THE FINANCIAL STATEMENTS

Arrangements that do not have the legal status of a lease but convey a right to use an asset in return for payment are accounted for under this policy where fulfilment of the arrangement is dependent on the use of specific assets.

The Council as Lessee: Finance Leases - Property, plant and equipment held under finance leases is recognised on the Balance Sheet at the commencement of the lease at its fair value measured at the lease's inception (or the present value of the minimum lease payments, if lower). The asset recognised is matched by a liability for the obligation to pay the lessor. Initial direct costs of the council are added to the carrying amount of the asset. Premiums paid on entry into a lease are applied to writing down the lease liability. Contingent rents are charged as expenses in the periods in which they are incurred.

Lease payments are apportioned between a charge for the acquisition of the interest in the property, plant or equipment, applied to write down the lease liability, and a finance charge (debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement).

Property. Plant and Equipment recognised under finance leases is accounted for using the policies applied generally to such assets, subject to depreciation being charged over the lease term if this is shorter than the asset's estimated useful life (where ownership of the asset does not transfer → to the council at the end of the lease period).

The council is not required to raise council tax to cover depreciation or revaluation and impairment losses arising on leased assets. Instead, a prudent annual contribution is made from revenue funds towards the deemed capital investment in accordance with statutory requirements. Depreciation and revaluation and impairment losses are therefore substituted by a revenue contribution in the General Fund Balance, by way of an adjusting transaction with the Capital Adjustment Account in the Movement in Reserves Statement for the difference between the two.

The Council as Lessee: Operating leases - Rentals paid under operating leases are charged to the Comprehensive Income and Expenditure Statement as an expense of the services benefitting from use of the leased property, plant or equipment. Charges are made on a straight-line basis over the life of the lease, even if this does not match the pattern of payments (e.g. there is a rent-free period at the commencement of the lease).

The Council as Lessor: Finance Leases - Where the council grants a finance lease over a property or an item of plant or equipment the relevant asset is written out of the Balance Sheet as a disposal. At the commencement of the lease, the carrying amount of the asset in the Balance Sheet (whether Property, Plant and Equipment or Assets Held for Sale) is written off to the Other Operating Expenditure line in the Comprehensive Income and Expenditure Statement as part of the gain or loss on disposal. A gain representing the council's net investment in the lease is credited to the same line in the Comprehensive Income and Expenditure Statement as part of the gain or loss on disposal (i.e. netted off against the carrying value of

the asset at the time of disposal), matched by a lease (long-term debtor) asset in the Balance Sheet.

Lease rentals receivable are apportioned between a charge for the acquisition of the interest in the property - applied to write down the lease debtor (together with any premiums received) – and finance income (credited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement).

The gain credited to the Comprehensive Income and Expenditure Statement on disposal is not permitted by statute to increase the General Fund Balance and is required to be treated as a capital receipt. Where a premium has been received, this is transferred out of the General Fund Balance to the Capital Receipts Reserve in the Movement in Reserves Statement. Where the amount due in relation to the lease asset is to be settled by the payment of rentals in future financial years, this is transferred out of the General Fund Balance to the Deferred Capital Receipts Reserve in the Movement in Reserves Statement. When the future rentals are received, the element for the capital receipt for the disposal of the asset is used to write down the lease debtor. At this point, the deferred capital receipts are transferred to the Capital Receipts Reserve.

The write-off value of disposals is not a charge against council tax, as the cost of property, plant and equipment is fully provided for under separate arrangements for capital financing. Amounts are therefore appropriated to the Capital Adjustment Account from the General Fund Balance in the Movement in Reserves Statement.

The Council as Lessor: Operating Leases - Where the council grants an operating lease over a property or an item of plant or equipment, the asset

The Council as Lessor: Operating Leases - Where the council grants an operating lease over a property or an item of plant or equipment, the asset is retained in the Balance Sheet. Rental income is credited to the Other Operating Expenditure line in the Comprehensive Income and Expenditure Statement. Credits are made on a straight-line basis over the life of the lease, even if this does not match the pattern of payments (e.g. there is a premium paid at the commencement of the lease).

18. Overheads and Support Services

The costs of overheads and support services are charged to those that benefit from the supply or service in accordance with the costing principles of the CIPFA Service Reporting Code of Practice (SERCOP). The total absorption costing principle is used - the full cost of overheads and support services are shared between users in proportion to the benefits received, with the exception of:

- Corporate and Democratic Core costs relating to the council's status as a multi-functional, democratic organisation.
- Non Distributed Costs the cost of discretionary benefits awarded to employees retiring early and impairment losses chargeable on Assets Held for Sale.

These two cost categories are defined in the SERCOP and accounted for as separate headings in the Comprehensive Income and Expenditure Statement, as part of Net Expenditure on Continuing Services.

19. Property, Plant and Equipment

Assets that have physical substance and are held for use in the production or supply of goods or services, for rental to others, or for administrative purposes, and that are expected to be used during more than one financial year are classified as Property, Plant and Equipment.

Recognition - Expenditure on the acquisition, creation or enhancement of Property, Plant and Equipment is capitalised on an accruals basis, provided that it is probable that the future economic benefits or service potential associated with the item will flow to the council and the cost of the item can be measured reliably. Expenditure that maintains but does not add to an asset's potential to deliver future economic benefits or service potential (i.e. repairs and maintenance) is charged as an expense when it is incurred.

Measurement - Assets are initially measured at cost, comprising the purchase price and any costs attributable to bringing the asset to the location and condition necessary for it to be capable of operating in the manner intended by management, subject to a de-minimis value of £100,000 for land and property and heritage assets and £10,000 for new vehicles, plant and equipment. The council does not capitalise borrowing costs incurred whilst assets are under construction.

The cost of an asset acquired other than by purchase is deemed to be its fair value, unless the acquisition does not have commercial substance (i.e. it will not lead to a variation in the cash flows of the council). In the latter case, where an asset is acquired via an exchange, the cost of the acquisition is the carrying amount of the asset given up by the council.

Donated assets are measured initially at fair value. The difference between fair value and any consideration paid is credited to the Taxation and Non-Specific Grant Income line of the Comprehensive Income and Expenditure Statement, unless the donation has been made conditionally. Until conditions are satisfied, the gain is held in the Donated Assets Account. Where gains are credited to the Comprehensive Income and Expenditure Statement, they are reversed out of the General Fund Balance to the Capital Adjustment Account in the Movement in Reserves Statement.

Assets are then carried in the Balance Sheet using the following measurement bases:

Infrastructure, community assets and assets under construction - depreciated historical cost;

Page 153

6. THE FINANCIAL STATEMENTS

- Dwellings fair value, determined using the basis of existing use value for social housing (EUV-SH);
- All other assets fair value, determined as the amount that would be paid for the asset in its existing use (existing use value EUV).

Where there is no market-based evidence of fair value because of the specialist nature of an asset, depreciated replacement cost (DRC) is used as an estimate of fair value. Where non-property assets have short useful lives or low values (or both), the depreciated historical cost basis is used as a proxy for fair value.

Assets included in the Balance Sheet at fair value are revalued sufficiently regularly to ensure that their carrying amount is not materially different from their fair value at the year end, but as a minimum every five years. Increases in valuations are matched by credits to the Revaluation Reserve to recognise unrealised gains. Gains are credited to the Surplus or Deficit on the Provision of Services where they arise from the reversal of a loss previously charged to a service.

Where decreases in value are identified, they are accounted for by:

- Where there is a balance of revaluation gains for the asset in the Revaluation Reserve, the carrying amount of the asset is written down against that balance (up to the amount of the accumulated gains);
- Where there is no balance in the Revaluation Reserve or an insufficient balance, the carrying amount of the asset is written down against the relevant service line(s) in the Comprehensive Income and Expenditure Statement.

The Revaluation Reserve contains revaluation gains recognised since 1 April 2007 only, the date of its formal implementation. Gains arising before that date have been consolidated into the Capital Adjustment Account.

Impairment - Assets are assessed at each year end as to whether there is any indication that an asset may be impaired. Where indications exist and any possible differences are estimated to be material, the recoverable amount of the asset is estimated and, where this is less than the carrying amount of the asset, an impairment loss is recognised for the shortfall.

Where impairment losses are identified, they are accounted for by:

- Where there is a balance of revaluation gains for the asset in the Revaluation Reserve, the carrying amount of the asset is written down against that balance (up to the amount of the accumulated gains);
- Where there is no balance in the Revaluation Reserve or an insufficient balance, the carrying amount of the asset is written down against the relevant service line(s) in the Comprehensive Income and Expenditure Statement.

Where an impairment loss is subsequently reversed, the reversal is credited to the relevant service line(s) in the Comprehensive Income and Expenditure Statement, up to the amount of the original loss, adjusted for depreciation that would have been charged if the loss had not been recognised.

Depreciation - Depreciation is provided for on all Property, Plant and Equipment assets by the systematic allocation of their depreciable amounts over their useful lives. An exception is made for assets without a determinable finite useful life (i.e. freehold land and certain Community Assets) and assets that are not yet available for use (i.e. assets under construction). Depreciation is charged on surplus assets.

Depreciation is calculated on the following bases:

- Dwellings and other buildings straight-line allocation over the useful life of the property as estimated by the valuer;
- Vehicles, plant, furniture and equipment straight-line allocation over the useful life of the asset;
- Infrastructure straight-line allocation over 50 years.

Where an item of Property, Plant and Equipment has major components whose cost is significant in relation to the total cost of the item, the components are depreciated separately. Revaluation gains are also depreciated, with an amount equal to the difference between current value depreciation charged on assets and the depreciation that would have been chargeable based on their historical cost being transferred each year from the Revaluation Reserve to the Capital Adjustment Account.

Disposals

When an asset is disposed of or decommissioned, the carrying amount of the asset in the Balance Sheet (whether Property, Plant and Equipment or Assets Held for Sale) is written off to the Other Operating Expenditure line on the Comprehensive Income and Expenditure Statement as part of the gain or loss on disposal. Receipts from disposals (if any) are credited to the same line in the Comprehensive Income and Expenditure Statement as part of the gain or loss on disposal (i.e. netted off against the carrying value of the asset at the time of disposal). Any revaluation gains accumulated for the asset in the Revaluation Reserve are transferred to the Capital Adjustment Account.

Amounts received for a disposal in excess of £10,000 are categorised as capital receipts. A proportion of receipts relating to housing disposals (75% for dwellings, 50% for land and other assets, net of statutory deductions and allowances) is payable to the Government. The balance of receipts is required to be credited to the Capital Receipts Reserve, and can then only be used for new capital investment (or set aside to reduce the council's underlying need to borrow (the capital financing requirement)). Receipts are appropriated to the Capital Receipts Reserve from the General Fund Balance in the Movement in Reserves Statement.

The written-off value of disposals is not a charge against council tax, as the cost of property, plant and equipment is fully provided for under separate arrangements for capital financing. Amounts are appropriated to the Capital Adjustment Account from the General Fund Balance in the Movement in Reserves Statement.

⁽¹⁾ 20. Private Finance Initiative (PFI) and Similar Contracts

PFI and similar contracts are agreements to receive services, where the responsibility for making available the property, plant and equipment needed to provide the services passes to the PFI contractor. As the council is deemed to control the services that are provided under PFI schemes, and as ownership of the property, plant and equipment will pass to the council at the end of the contracts for no additional charge, the council carries the assets used under the contracts on its Balance Sheet as part of Property, Plant and Equipment.

The original recognition of these assets at fair value (based on the cost to purchase the property, plant and equipment) was balanced by the recognition of a liability for amounts due to the scheme operator, from the council and third parties where relevant, to pay for the capital investment. Non-current assets recognised on the Balance Sheet are revalued and depreciated in the same way as property, plant and equipment owned by the council.

The amounts payable to the PFI operators each year are analysed into five elements:

- Fair value of the services received during the year debited to the relevant service in the Comprehensive Income and Expenditure Statement;
- Finance cost an interest charge on the outstanding Balance Sheet liability, debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement;
- Contingent rent increases in the amount to be paid for the property arising during the contract, debited to the Financing and Investment Income and Expenditure line in the Comprehensive Income and Expenditure Statement;
- Payment towards liability applied to write down the Balance Sheet liability to the PFI operator (the profile of write-downs is calculated using the same principles as for a finance lease);
- Lifecycle replacement costs proportion of the amounts payable is posted to the Balance Sheet as a prepayment and then recognised as additions to Property, Plant and Equipment when the relevant works are eventually carried out.
- Third party income credited to the Comprehensive Income and Expenditure Statement, reflecting the extent to which the asset and the service are financed by third party income.

රා රා 21. Provisions, Contingent Liabilities and Contingent Assets

Provisions - Provisions are made where an event has taken place that gives the council a legal or constructive obligation that probably requires settlement by a transfer of economic benefits or service potential, and a reliable estimate can be made of the amount of the obligation. For instance, the council may be involved in a court case that could eventually result in the making of a settlement or the payment of compensation.

Provisions are charged as an expense to the appropriate service line in the Comprehensive Income and Expenditure Statement in the year that the council becomes aware of the obligation, and are measured at the best estimate at the balance sheet date of the expenditure required to settle the obligation, taking into account relevant risks and uncertainties. When payments are eventually made, they are charged to the provision carried in the Balance Sheet. Estimated settlements are reviewed at the end of each financial year - where it becomes less than probable that a transfer of economic benefits will now be required (or a lower settlement than anticipated is made), the provision is reversed and credited back to the relevant service.

Where some or all of the payment required to settle a provision is expected to be recovered from another party (e.g. from an insurance claim), this is only recognised as income for the relevant service if it is virtually certain that reimbursement will be received if the council settles the obligation.

Contingent Liabilities - A contingent liability arises where an event has taken place that gives the council a possible obligation whose existence will only be confirmed by the occurrence or otherwise of uncertain future events not wholly within the control of the council. Contingent liabilities also arise in circumstances where a provision would otherwise be made but either it is not probable that an outflow of resources will be required or the amount of the obligation cannot be measured reliably. Contingent liabilities are not recognised in the Balance Sheet but disclosed in a note to the accounts.

22. Reserves

The council sets aside specific amounts as reserves for future policy purposes or to cover contingencies. Reserves are created by appropriating amounts out of the General Fund Balance in the Movement in Reserves Statement. When expenditure to be financed from a reserve is incurred, it is charged to the appropriate service in that year to score against the Surplus or Deficit on the Provision for Services in the Comprehensive Income and Expenditure Statement. The reserve is then appropriated back into the General Fund Balance in the Movement in Reserves Statement so that there is no net charge against council tax for the expenditure. Certain reserves are kept to manage the accounting processes for non-current assets, financial instruments, retirement and employee benefits and do not represent usable resources for the council – these reserves are explained in the relevant policies.

23. Revenue Expenditure Funded from Capital under Statute

Expenditure incurred during the year that may be capitalised under statutory provisions but does not result in the creation or enhancement of a non-current asset is charged as expenditure to the relevant service in the Comprehensive Income and Expenditure Statement in the year. Where the council determines to meet the cost of this expenditure from capital resources, a transfer in the Movement in Reserves Statement from the General Fund Balance to the Capital Adjustment Account then reverses out the amounts charged so that there is no impact on the level of Council Tax.

24. Value Added Tax

VAT payable is included as an expense only to the extent that it is not recoverable from Her Majesty's Revenue and Customs. VAT receivable is excluded from income.

25. Pension Fund Accounts

As a result of Local Government Reorganisation on 1 April 1986, the Council assumed responsibility for administering the West Midlands Pension Fund. The fund's accounts are separately prepared and are included within these accounts. The accounting policies for the pension fund can be found at note P3 in the accounts.

Copies of the fund's Accounts and Annual Report are available on request from the Assistant Director Finance, Civic Centre, Wolverhampton, WV1 1RL.

26. Heritage Assets

Heritage assets are recognised and measured (including the treatment of revaluation gains and losses) in accordance with the council's accounting policies on property, plant and equipment, with the following exceptions:

- Where there is no market-based evidence of fair value, insurance valuation is used as an estimate of fair value;
- There is no cyclical revaluation of heritage assets, which instead are kept under review for impairment on an annual basis;
- The groupings used to classify property, plant and equipment assets are not used for heritage assets.

27. Accounting for the Carbon Reduction Commitment Scheme

The council is required to participate in the Carbon Reduction Commitment (CRC) Energy Efficiency Scheme. This scheme is currently in its introductory phase which will last until 31 March 2014. The council is required to purchase and surrender allowances, currently retrospectively, on the basis of emissions i.e. carbon dioxide produced as energy is used. As carbon dioxide is emitted (i.e. as energy is used), a liability and an expense are recognised. The liability will be discharged by surrendering allowances. The liability is measured at the best estimate of the expenditure required to meet the obligation, normally at the current market price of the number of allowances required to meet the liability at the reporting date. The cost to the authority is recognised and reported in the costs of the council's services and is apportioned to services on the basis of energy consumption. CRC scheme assets are classified as either current intangible assets or if held for the purposes of trading, as current assets. The asset is initially measured at cost. Allowances that are issued for less than their fair value are initially measured at their fair value, with the difference between fair value and the purchase price recognised as income.

Note 11B - Changes in Accounting Policies from Last Year

There have been no changes in accounting policies from 2012/13.

Note 11C - Critical Judgements Made When Applying the Accounting Policies

In applying the accounting policies set out in this note, the council has made judgements about complex transactions or those involving uncertainty about future events. The critical judgements made in the Statement of Accounts are as follows:

Group Accounts

The council is the sole shareholder of Wolverhampton Homes Limited, an arm's length management organisation that provides housing management services to the council in respect of its HRA dwellings. It has been determined that the council is able to control Wolverhampton Homes Limited, and it has therefore been consolidated within the Group Accounts.

The council is the main funder of the Grand Theatre, and in practice bears the risk of the Theatre going into overall deficit. However, it has been determined that the transactions and balances of the theatre company are not material to the council's accounts, and it has therefore not been consolidated in the Group Accounts.

During 2012/13, the Wolverhampton Schools' Improvement Partnership was established as a company limited by guarantee. The directors of the Tompany are representatives of local schools and the senior substantive council officer for schools. Two councillors are non-voting directors. Whilst in this way the council exerts significant influence over the activities of the company, it has been determined that the transactions and balances of the company are not material to the council's accounts, and it has therefore not been consolidated in the Group Accounts.

The council, along with the other six West Midlands district councils, holds shares in Birmingham International Airport. However, it has been determined that the council does not have the power to influence or control the Airport, and it has therefore not been consolidated in the Group Accounts.

Private Finance Initiative (PFI) Contracts

The council provides services via private sector partners under a PFI or PFI-type contract in three areas: Waste Management, Bentley Bridge Leisure Centre, and Highfields and Penn Fields School. In all three cases, it has been determined that the council controls the use of the relevant non-current assets such that they are recognised as assets of the council, and a corresponding liability has been recognised in the council's accounts.

Equal Pay Back Pay

Under the Equal Pay Act 1970, as amended by the Equal Pay Act (Amendments) Regulations 2003, employees are entitled to equal pay for work of equal value. Where this has historically not been the case, the council may be liable to make compensatory payments to employees who were disadvantaged by the prevailing rates of pay.

The timing and amount of any such compensation payments are not certain, however a provision has been established based on high-level estimates of the total potential liability.

Property, Plant and Equipment belonging to Voluntary Aided Schools

The council owns land on which a number of voluntary aided schools have, with its consent, placed buildings. The buildings belong to the foundation/controlling interest and therefore the council cannot exercise control over those buildings. Until the tenant decides to remove or vacate and demolish those buildings there is no alternative use for this site and therefore no market value in the land. Whilst the schools provide a service to the city of Wolverhampton by delivering education from those sites, the cost of maintaining the schools falls upon the Dedicated Schools Grant or other entities. The land on which those buildings are sited is therefore not recognised as an asset of the council.

Property, Plant and Equipment belonging to Academy Schools

When an agreement in principle has been made for a school to convert to Academy status on a long term lease the lease is classified as a finance lease, since substantially all associated risks and rewards of ownership of the asset have been transferred. As the Council would no longer controls or maintains the asset for the majority of its economic life the buildings are removed from the balance sheet as a disposal.

→ Accounting for the Voluntary Redundancy Programme

The council undertook a voluntary redundancy exercise during 2013/14, and there were a number of employees who were part-way through the approval process at the year end. The council has taken a prudent approach to the treatment of the anticipated costs of such employees, recognising those costs in 2013/14 where it seemed likely that approval would be given and a redundancy would result.

Business Rates

As part of the changes to Business rates retention, commencing 1 April 2014, councils have assumed the liability for refunding rate payers who successfully appeal against the rateable value of their properties, including amounts that were paid to the Government in 2012/13 and earlier. The council has set aside a provision for these refunds, calculated using the Valuation Office list of ratings appeals data.

Note 11D – Major Assumptions about the Future

The Statement of Accounts contains estimated figures that are based on assumptions made by the council about the future or that are otherwise uncertain. Estimates are made taking into account historical experience, current trends and other relevant factors. However, because balances cannot be determined with certainty, actual results could be materially different from the assumptions and estimates.

The items in the council's Balance Sheet at 31 March 2014 for which there is a significant risk of material adjustment in the forthcoming financial year are as follows:

Pensions Net Liability

Estimation of the net liability to pay pensions depends on a number of complex judgements relating to the discount rate used, the rate at which salaries are projected to increase, changes in retirement ages, mortality rates and expected returns on pension fund assets. A firm of consulting actuaries is engaged to provide the council with expert advice about the assumptions to be applied. The table below sets out the impact on the net pensions liability if different assumptions had been made in certain key areas:

	Variation to Assumptions	Impact on Net Liability		
		Council	Group	
lge 161	Discount Rate 0.1% higher	Decrease of £21.0m	Decrease of £23.4m	
	Salary Inflation 0.1% p.a. higher	Increase of £21.3m	Increase of £23.8m	
	Pay Growth 0.1% p.a. higher	Increase of £4.8m	Increase of £5.7m	
	Life expectancy of scheme members 1 year higher	Increase of £23.8m	Increase of £25.9m	

Property, Plant and Equipment

In accordance with the requirements of the Code, the council re-values its property, plant and equipment assets on a five-yearly cyclical basis. As a result, it always carries a number of such assets at values which are not recent, but which are nonetheless assumed to be materially correct.

Note 11E - Accounting Standards Issued but Not Yet Adopted

The Code of Practice on Local Council Accounting in the United Kingdom 2014/15 (the Code) has introduced several changes in accounting policies which will be required from 1 April 2014. If these had been adopted for the financial year 2013/14 there would be no material changes as detailed below.

- In accordance with the Code, the council has not adopted IFRS 13 Fair Value Measurement, which has been deferred until 2014/15. IFRS 13 aims to provide a precise definition of fair value and a single source of fair value measurement and disclosure requirements for use across IFRSs. The application of this standard to local government assets and liabilities will be considered in detail in the development of the 2014/15 Code.
- IFRS 10 Consolidated Financial Statements This standard introduces a new definition of control, which is used to determine which entities are consolidated for the purposes of group accounts. The largest subsidiary of the council is Wolverhampton Homes, which is already consolidated into the group accounts. From 2014/15 Yoo recruit, the council's new recruitment agency, will be consolidated into the accounts, assuming the transactions are material.
- IFRS 11 Joint Arrangements This standard addresses the accounting for a 'joint arrangement', which is defined as a contractual arrangement over which two or more parties have joint control. These are classified as either a joint venture or a joint operation. In addition proportionate consolidation can no longer be used for jointly controlled entities. The council has no material joint venture arrangements.
- IFRS 12 Disclosures of Involvement with Other Entities This is a consolidated disclosure standard requiring a range of disclosures about an entity's interests in subsidiaries, joint arrangements, associates and unconsolidated 'structured entities'. The council has a number of arrangements with other entities under IFRS12.
- IAS 27 Separate Financial Statements and IAS 28 Investments in Associates and Joint Ventures These statements have been amended to conform with the changes in IFRS 10, IFRS 11 and IFRS 12. Given that there would be no changes in the financial statements, except for disclosure, due to the changes to IFRS 10, IFRS 11 and IFRS 12, there is therefore also no impact as a result of changes in IAS 27 and IAS 28.
- IAS 32 Financial Instruments Presentation The Code refers to amended application guidance when offsetting a financial asset and a financial liability. The gains and losses are separately identified in the Comprehensive Income and Expenditure Statement and therefore no further disclosure is required.
- IAS 1 Presentation of the Financial Statements The changes clarify the disclosure requirements in respect of comparative information for the preceding period. The Statement of Accounts fully discloses comparative information for the preceding period, therefore these changes will not have a material impact on the Statement of Accounts.

Housing Revenue Account Income and Expenditure Statement

2012/13 £m		Notes	2013/14 £m
(83.0)	Gross Rents - Dwellings		(89.0)
(1.6)	-		(1.6)
(5.0)	<u> </u>		(5.0)
-	Contributions		(0.1)
(89.6)	Total Income		(95.7)
26.1	Repairs and Maintenance		25.9
17.3	 		18.4
0.2	Rents, Rates and Taxes		0.4
0.6	Increase in Allowance for Bad Debts		0.9
22.6	Depreciation of Property, Plant and Equipment	H1	21.9
66.8	Total Expenditure		67.5
(22.8)	Net Cost of HRA Services as Included in Council Comprehensive Income and Expenditure Statement		(28.2)
0.2	HRA Share of Corporate and Democratic Core		0.2
(22.6)	Net Cost of HRA Services		(28.0)
(0.7)	(Gain) on Sale of Property, Plant and Equipment		-
-	(Gain) on the Fair Value of Investment Assets		(0.2)
14.2	Interest Payable		13.4
-	Premiums and Discounts		-
(0.2)	Interest and Investment Income		(0.1)
(9.3)	Surplus for the Year		(14.9)

age 16

Page 164

7. THE HOUSING REVENUE ACCOUNT STATEMENTS

Movement on the Housing Revenue Account Balance Statement

2012/13 £m		Notes	2013/14 £m
(10.5)	Opening HRA Balance		(5.0)
	(Increase)/Decrease in the HRA balance for the year analysed between:		
(9.3)	- (Surplus) for the year on the Income and Expenditure Account		(14.9)
14.8	- Net additional amount required by statute and non-statutory proper practices to be debited or credited to the HRA balance for the year	Н3	14.9
5.5	(Increase)/Decrease in the HRA balance for the year		-
(5.0)	Closing HRA Balance		(5.0)

Notes to the Housing Revenue Account Statements

Note H1 - Depreciation

2012/13 £m		2013/14 £m
21.9	Council Dwellings	21.9
0.5	Other Land and Buildings	0.2
0.2	Vehicles, Plant, Furniture and Equipment	0.2
22.6	Total Depreciation Charge for the Year	22.3

Under the Housing Revenue Account regulations, depreciation and impairment charges are reversed out and replaced with a provision for the repayment of debt.

Note H2 – Impairment

There were no impairments in 2013/14 or 2012/13.

Note H3 - Analysis of the Movement on the HRA Balance Statement

	2012/13 £m		Note	2013/14 £m
	14.8	Net additional amount required to be debited or credited to the HRA Balance		14.9
	£m			
		Comprising:		
		Amounts included in the Income and Expenditure Account but not in the HRA Balance		
	0.7	- Net Gain on Sale of Property, Plant and Equipment		-
		- Net Gain / (Loss) on the Fair Value of Investment Assets		0.2
	0.7	Subtotal		0.2
		Amounts not in the Income and Expenditure Account but included in the HRA Balance		
9 9 1	-	- HRA Share of Contribution to Pension Reserve	H4	0.2
Ď	14.3	- Amount Set Aside for the Repayment of Debt		14.5
1 66 —	(0.2)	- Transfer to/(from) Earmarked Reserves		-
<u> </u>	14.1	Subtotal		14.7
	14.8	Total		14.9

Note H4 - Contribution to the Pension Reserve

Retirement benefits are offered to employees by the council as part of the terms and conditions of employment. Although these benefits will not actually be payable until employees retire, the council has a commitment to make the payments. This commitment needs to be disclosed at the time that employees earn their future entitlement. The pension reserve reflects the projected shortfall in the amount which may need to be provided in the future to current employees of the Housing Revenue Account. Further details on this may be found in Note 6 to the Core Financial Statements.

Note H5 - Housing Stock

The number of dwellings held or leased by the council on the below dates (excluding properties earmarked for demolition or sale) are shown in the following table.

31 March 2013		31 March 2014
5,032	Low Rise Flats	4,995
3,016	Medium Rise Flats	2,991
2,119	High Rise Flats	2,120
13,127	Houses and Bungalows	12,987
23,294	Total Dwellings Owned by the Council	23,093
14	Homeless Dwellings (Leased)	14
23,308		23,107

ປັ່ນ Note H6 – Housing Revenue Account Property, Plant and Equipment

The following table shows the total Balance Sheet values of the land, houses and other property within the Housing Revenue Account at the end of the year.

31 March 2013 £m		31 March 2014 £m
687.8	- Council Dwellings	702.0
9.5	- Other Land and Buildings	10.0
0.2	- Vehicles, Plant, Furniture and Equipment	0.1
0.7	- Intangible Assets	0.5
698.2	Total Property, Plant and Equipment	712.6

Note H7 - The Vacant Possession Value of Dwellings

The vacant possession value of the stock of dwellings at 31 March 2014 (at 1 April 2010 prices) amounted to £2,064.7 million (31 March 2013: £2,022.9 million). The value of dwellings shown on the Balance Sheet is the existing use value (social housing), which is 34% of the vacant possession value (this ratio is set by the government). The difference between the two values demonstrates the economic cost to government of providing council housing at less than open market rents.

Note H8 - Capital

Capital expenditure on land, houses and other property within the HRA during the year and how it was paid for is shown in the following table.

Р	2012/13 £m		2013/14 £m
ag	<u>د</u> اוו	Sources of Funding	ŽIII
Φ,	(1.4)	- Borrowing	(11.1)
168	(1.3)	- Usable Capital Receipts	(1.8)
ω	(17.5)	- Major Repairs Reserve	(18.3)
	(20.7)	- Government and EU Grants	(11.9)
	(0.5)	- Other Grants and Contributions	-
	(41.4)	Total Capital Expenditure	(43.1)

Capital receipts generated during 2013/14 from the disposal of HRA assets are detailed in the following table.

2012/13 £m		2013/14 £m
(4.7)	Sale of Council Houses (Right-to-Buy)	(8.0)
(0.8)	Sale of Other Land and Buildings	(0.4)
(5.5)	Total Capital Receipts	(8.4)

These receipts were split between the council and the Government, as shown in the table below.

2012/13 £m		2013/14 £m
(1.5)	Paid over to Government	(1.8)
(4.0)	Available to Finance Capital Expenditure	(6.6)
(5.5)	Total Capital Receipts	(8.4)

Note H9 - Rent Arrears

During 2013/14, total rent arrears increased by £300,000 (16.7%). Within total rent arrears, current tenants' arrears as a proportion of net rental income increased from 4.6% to 6.0%. The comparative total figures are shown in the following table.

Page	31 March 2013 £m		31 March 2014 £m
<u> </u>	1.0	Current Tenants	1.2
60	0.8	Former Tenants	0.9
9	1.8	Total Arrears	2.1

An allowance is maintained for these debts. The table below details the movement in the year.

2012/13 £m		2013/14 £m
1.5	Allowance for Bad and Doubtful Debts Brought Forward	1.5
(0.5)	Amounts Written Off during the Year	(0.4)
0.5	Increase in Allowance Charged to the HRA during the Year	0.5
1.5	Allowance for Bad and Doubtful Debts Carried Forward	1.6

Note H10 - Major Repairs Reserve

This is a discretionary reserve to which the council's Major Repairs Allowance (MRA) is transferred, and that is used to finance major repairs to HRA property. The MRA was determined by the Government as part of the final HRA subsidy determination. Where total HRA depreciation charges are greater than the MRA it is a requirement that an amount equal to the difference is transferred to the HRA from the Major Repairs Reserve.

2012/13 £m		2013/14 £m
(5.2)	Balance Brought Forward	(9.6)
(21.9)	Transfer of MRA from the Capital Adjustment Account	(21.9)
17.5	Capital Expenditure on Land and Property in the HRA	18.3
(9.6)	Balance Carried Forward	(13.2)

The Collection Fund Statement

2012/13 £m		Note	2013/14 £m
-	Deficit/(surplus) Brought Forward		0.6
	Income		
(78.1)	Council Tax	C1	(83.4)
(27.1)	Transfers from General Fund: Council Tax Benefits		-
(72.1)	Non Domestic Rates		(75.9)
-	Grant Income		(1.0)
(177.3)	Total Income		(160.3)
	Expenditure		
	Precepts and Demands		
94.3	Wolverhampton City Council		73.3
7.1	West Midlands Police		5.7
3.4	West Midlands Fire and Civil Defence		2.9
104.8			81.9
	Non Domestic Rates		
70.0	Payments to National Pool		-
-	Central Government		35.5
-	West Midlands Fire and Civil Defence		0.7
-	Wolverhampton City Council		35.2
0.4	Cost of Collection Allowance		0.3
70.4			71.7

age 1/1

2012/13 £m		Note	2013/14 £m
	Distribution of Council Tax Surplus/(Payment of Deficit)		
-	Wolverhampton City Council		0.2
-	West Midlands Police		-
-	West Midlands Fire and Civil Defence		-
	Distribution of Business Rates Surplus/(Payment of Deficit)		
-	Wolverhampton City Council		-
-	Central Government		-
-	West Midlands Fire and Civil Defence		-
0			
0.9	Allowance for Bad and Doubtful Debts		
0.9	Council Tax		1.1
1.8	Non Domestic Rates		2.6
2.7	Total Allowance for Bad and Doubtful Debts		3.7
-	Provision for appeals		4.2
	5 year spread adjustment		(2.7)
177.9	Total Expenditure		159.0
0.6	Deficit/(surplus) for the Year		(1.3)
0.6	Deficit/(surplus) Carried Forward		(0.7)

Notes to the Collection Fund Statement

Note C1 - The Council Tax Base

Council tax income derives from charges raised according to the residential properties, which have been classified into eight valuation bands. Individual charges are calculated by estimating the amount of income required to be taken from the collection fund for the forthcoming year and dividing this by the tax base. The Council's tax base is the number of chargeable dwellings in each valuation band (adjusted for dwellings where discounts apply) converted to an equivalent number of band D dwellings. Council tax bills were based on the following proportions for bands A to H.

	Band	Total Number of Chargeable Dwellings after Effect of Discount	Ratio	Band D Equivalent Dwellings	Council Tax (Single Person Household)	Council Tax (Multiple Occupancy)
					£	£
Page	A Disabled	91	5/9	51	614	818
ge	Α	46,797	6/9	31,198	736	981
0 1	В	20,375	7/9	15,847	859	1,145
7	С	14,254	8/9	12,670	981	1,308
ယ	D	5,675	9/9	5,675	1,104	1,472
	E	2,685	11/9	3,282	1,349	1,799
	F	1,531	13/9	2,210	1,595	2,126
	G	787	15/9	1,311	1,840	2,453
	Н	83	18/9	167	2,208	2,944
F		92,278		72,411		
-	Add: Additional Band D Equivalent Dwellings from removing 50% Council Tax Discount for owners of long term void homes Add: Reliefs and deletions			13		
-				113		
_	Less: Allowance for collection difficulties (98.75%)			(906)		
	Total Band D Tax Base)		71,631		

Note C2 - National Non-Domestic Rates

The council collects National Non-Domestic Rates (NNDR) for its area based on local rateable values provided by the Valuation Office Agency (VOA) multiplied by a uniform business rate set by Central Government.

In 2013/14 the administration of NNDR changed following the introduction of business rates retention. Instead of paying NNDR into a central pool, local authorities retain a proportion of the total collectable rates due. In Wolverhampton, the council retain 49%, 50% is paid to central government and the remaining 1% is paid to West Midlands Fire Service.

The total non-domestic rateable value was £190.3 million as at 31 March 2014 (£190.5 million as at 31 March 2013). The national multipliers for 2013/14 were 46.2p for qualifying small businesses, and the standard multiplier was 47.1p for all other businesses (45.0p and 45.8p respectively in 2012/13).

9. WEST MIDLANDS PENSION FUND STATEMENTS

Fund Account

2012/13		Note	2013/14
£m			£m
	Contributions & Benefits		
410.9	Contributions Receivable	P7	419.3
22.6	Transfers In	P8	11.3
16.2	Other Income	P9	16.2
449.7	Total contributions and other income		446.8
445.1	Benefits Payable	P10	479.2
	Payments to and on account of leavers	P11	22.8
0.2	,		0.3
5.3	· ·	P12	5.0
466.4	Total benefits and other expenditure		507.3
(16.7)	Net additions from dealings with members		(60.5)
	Returns on Investments		
136.8	Investment Income	P13	133.4
846.0	Changes in Value of Investments		87.2
97.7	Profit and Losses on Disposal of Investments		103.8
(11.3)	Investment Management Expenses	P12	(9.8)
1,069.2	Net return on investments		314.6
1,052.5	Net increase in the Fund during the year		254.1
8,833.8	Net Assets of the Fund at the beginning of the year		9,886.3
9.886.3	Net Assets of the Fund at the end of the year		10,140.4

'age 175

9. WEST MIDLANDS PENSION FUND STATEMENTS

Net Assets Statement

31 March 2013			31 March 2014
£m		Note	£m
	Investment Assets (at Market Value)	P14	
173.9	Fixed Interest Securities		171.3
943.5	UK Equities		971.3
2,072.5	Overseas Equities		2,310.0
5,729.4	Pooled Investment Vehicles		5,754.1
567.6	Property		629.8
82.9	Foreign Currency Holdings		42.2
241.1	Cash Deposits		221.3
-	Other Investments assets		-
15.4	Outstanding dividend entitlement and recoverable with-holding tax		5.8
9,826.3	Investment Assets		10,105.8
	Investment Liabilities (at Market Value)	P14	
(0.1)	Other Investments liabilities		(1.6)
(0.1)	Investment Liabilities		(1.6)
9,826.2	Net Investment Assets		10,104.2
73.2	Current Assets	P17	67.1
(13.1)	Current Liabilities	P18	(30.8)
9,886.3	Net Assets of the Fund at the end of the year		10,140.5

The accounts summarise the transactions of the scheme and deal with the net assets at the disposal of the Committee. They do not take account of obligations to pay pensions and benefits which fall due after the end of the scheme year. The actuarial position of the scheme, which does take account of such obligations, is dealt with in the actuarial certificate/statement.

The notes form part of these financial statements.

Notes to the Pension Fund Statements

Note P1 - General

The West Midlands Pension Fund is administered by Wolverhampton City Council on behalf of all local authorities in the West Midlands and other employers who have members in the Fund.

The City Council Pensions Committee administers the Pension Fund function. It meets at approximately quarterly intervals, and has members from each of the seven Metropolitan District Councils in the West Midlands Region. An Investment Advisory Sub-Committee and a Joint Consultative Panel \mathbf{v} have been established to deal with the two areas of management and administration of the Fund.

The fund is administered under the rules of the Local Government Pension Scheme as set out in the Local Government Pension Scheme Regulations. This includes:

- (i) the LGPS (Benefits, Membership and Contribution) Regulations 2007 (as amended);
- (ii) the LGPS (Administration) Regulations 2008 (as amended); and
- (iii) the LGPS (Management and Investment of Funds) Regulations 2009.
- (iv) The Local Government Pension Scheme Regulations 2013

Benefits are funded by contributions and investment earnings. Contributions are made by active members of the fund in accordance with the LGPS (Benefits, Membership and Contribution) Regulations 2007 and range from 5.5% and 7.5% of pensionable pay for the financial year ending 31 March 2014. Employee contributions are matched by employers' contributions which are set based on triennial actuarial funding valuations. The last such valuation was at 31 March 2013. Employer contribution rates during 2013/14 ranged from 2.7% to 27.8% of pensionable pay.

The Fund's Statement of Investment Principles (SIP) can be found in the Annual Report and on the fund's website: www.wmpfonline.com.

132

Note P2 - Basis of Preparation

The Statement of Accounts summarises the fund's transactions for the 2013/14 financial year and its position at the year-end as at 31 March 2014. The accounts have been prepared in accordance with the Code of Practice on Local Authority Accounting in the United Kingdom 2013/14 which is based upon International Financial Reporting standards (IFRS), as amended for the UK public sector.

The accounts summarise the transactions of the fund and report on the net assets available to pay pension benefits. The accounts do not take account of obligations to pay pensions and benefits which fall due after the end of the financial year. The actuarial present value of promised retirement benefits, valued on an International Accounting Standard (IAS) 19 basis, is disclosed at Note P5 of these accounts.

Note P3 - Statement of Accounting Policies

Note P3A - Inclusion of Income and Expenditure

1. Membership of the Fund

Membership of the Fund is available for all local government employees including non-teaching staff of schools and further and higher education corporations in the West Midlands Region, together with employees of admitted bodies.

2. Fund Account

U

In the fund account income and expenditure are accounted for in the year in which they arise by the creation of payables and receivables at the year end where necessary. However, provision has not been made where the amount payable or receivable in relation to transfers was not agreed at the year end. (See Note P8).

3. Contribution Income

Contributions receivable have been included in the accounts on the accruals basis at the rates set out in Notes P1 and P5 for basic contributions. Additional contributions as notified by employers for the period have also been included.

Where member employing organisations have not submitted certified returns of contributions payable by the due date for preparation of these accounts, an estimate has been made based on the monthly returns of these bodies.

4. Transfers to and from other schemes

Transfer values represent the amounts received and paid during the year for members who had either joined or left the Scheme as at 31 March 2014, calculated in accordance with the Local Government Pension Scheme Regulations (see Notes P8 and P11). They are accounted for when trustees of the receiving scheme have agreed to accept the transfer.

5. Investment Income

Interest income - is recognised in the fund account as it accrues, using the effective interest rate of the financial instrument as at the date of acquisition or origination.

Dividend income - is recognised on the date the shares are quoted ex-dividend. Any amount not received by the end of the reporting period, where known to be due, have been accrued for in the accounts.

Distributions from pooled funds - are recognised at the date of issue. Any amount not received by the end of the reporting period is disclosed in the net assets statement as a current financial asset.

Property-related income - consists primarily of rental income. Rental income from operating leases on properties owned by the fund is recognised on a straight-line basis over the term of the lease. Any lease incentives granted are recognised as an integral part of the total rental income, over the term of the lease. Contingent rents based on the future amount of a factor that changes other than with the passage of time, such as turnover rents, are only recognised when contractually due

age 17

6. Benefits Payable

Pensions and lump sum benefits payable include all amounts known to be due as at 31 March 2014 relating to the financial year 2013/14.

7. Foreign Currency Transactions

Dividends, interest and purchases and sales of investments have been accounted for at the spot market rates at the date of transaction. End of year spot market exchange rates have been used to value cash balances held in foreign currency bank accounts, market values of overseas investments and purchases and sales outstanding at 31 March 2014.

Note P3B - Valuation of Investments

Page

The market values of investments as shown in the net assets statement have been determined as follows:-

1. Quoted Securities

Securities have been valued at the bid-market price ruling on 31 March 2014 where a quotation was available on a recognised stock exchange or unlisted securities market.

2. Unquoted Securities

The valuation of unquoted securities is based on the latest investor reports and financial statements provided by the fund managers of the underlying funds, adjusted for transactions arising after the date of such reports. A discount may be applied by the fund manager where trading restrictions apply to such securities. Where the first investor report has not been received from the fund manager the security is valued at cost.

3. Pooled Investment Vehicles

Pooled Investment Vehicles are stated at the bid-point of the latest prices quoted or the latest single market prices. In the case of the pooled investment vehicles which are accumulation funds, change in market value also includes income, net of withholding tax, which is reinvested in the fund.

4. Freehold and Leasehold Properties

These have been valued at their open market value. Property is valued by the Fund's Valuers on an annual basis. The market values included in these accounts are contained in a valuation report by Knight Frank LLP, Chartered Surveyors as at 31 March 2014. One third of the commercial property portfolio is valued fully in March each year, with the remaining two thirds being a 'desktop' valuation. The valuation undertaken at 31 March 2014 was therefore one third full valuation, and the remaining two thirds desktop valuations. Agricultural properties were valued by Savills Plc, Agricultural Valuers at the same date.

5. Foreign Currencies

Investments held in foreign currencies have been valued as set out in paragraphs P3B1 to P3B2 above and translated at exchange rates ruling at 31 March 2014.

Any gains or losses arising on translation of investments into sterling are accounted for as a change in market value of investment.

Note P3C - Cash and Cash Equivalents

Cash comprises cash in hand and demand deposits.

Cash equivalents are short-term, highly liquid investments that are readily convertible to known amounts of cash and that are subject to minimal risk of changes in value.

Note P3D - Financial Liabilities

The fund recognises financial liabilities at fair value as at the reporting date. A financial liability is recognised in the net assets statement on the date the fund becomes party to the liability. From this date any gains or losses arising from changes in the fair value of the liability are recognised by the fund.

Note P3E - Investment and Administration Expenses

All investment management expenses are accounted for on an accruals basis.

External investment management and custodian fees are agreed in management or custody agreements governing the administration of the individual mandates. Fees are generally based on the valuation of the underlying investments, either being managed or in safe custody, and as such will fluctuate as the valuations change.

In addition, performance related fees are negotiated with a number of managers and performance related fees totalled £1.3 million in 2013/14 and £2.2 million in 2012/13.

Where a management fee notification has not been received by the 31st March, an estimate is used for inclusion in the fund account.

The cost of using advice from external consultants is included in investment management fees.

The cost of in-house management is charged to the Fund, as is an element of the administering authority's officers time spent on management of the Pension Fund."

Note P3F - Membership

Overall membership of the Fund at the end of the year was as follows:-

31 March 2013		31 March 2014
£m		£m
97.3	Active Members	99.7
77.5	Pensioner Members	80.6
86.5	Deferred Members	90.0

A detailed list of Member bodies is available at Note P24

™ Note P4 - Critical Judgements in Applying Accounting Policies

___ Unquoted private equity investments

The valuation of unquoted securities is based on the latest investor reports and financial statements provided by the fund managers of the underlying funds, adjusted for transactions arising after the date of such reports. A discount may be applied by the fund manager where trading restrictions apply to such securities. Where the first investor valuation report has not been received from the fund manager the security is valued at cost. The value of unquoted private equity at 31 March 2014 was £1,240.5 million (£1,232.0 million at 31 March 2013).

Pension fund liability

The pension fund liability is calculated every three years by the appointed actuary, with annual updates in the intervening years. The methodology used is in line with accepted guidelines and in accordance with IAS 19. Assumptions underpinning the valuations are agreed with the actuary and are summarised in Note P5. This estimate is subject to significant variances based on changes to the underlying assumptions.

Note P4A - Assumptions Made About the Future and Other Major Sources of Estimation Uncertainty

Actuarial present value of promised retirement benefits

Uncertainties

U

Estimation of the net liability to pay pensions depends on a number of complex judgements relating to the discount rate used, the rate at which salaries are projected to increase, changes in retirement ages, mortality rates and expected returns on pension fund assets. Mercer Limited, the Fund's consulting actuaries, are engaged to provide expert advice about the assumptions to be applied.

Effect if actual results differ from assumptions

The effects on the net pension liability of changes in individual assumptions can be measured. For instance, an increase in the discount rate assumption would result in a decrease in the pension liability, however an increase in assumed earnings inflation or assumed life expectancy would significantly increase the pension liability as detailed by the Fund's consulting Actuary below:

Change in assumptions – year ended 31st March 2014	Approx % increase in liabilities	Approx monetary value £m
0.5% p.a. decrease in discount rate	10%	1,449.0
1 year increase in member life expectancy	2%	318.0
0.5% p.a. increase in salary increase rate	2%	315.0
0.5% p.a. increase in pensions increase rate*	9%	1,375.0

Private Equity

Uncertainties

Orivote equity Private equity investments are not publicly listed and as such there is a degree of estimation involved in the valuation.

Effect if actual results differ from assumptions

The total private equity investments in the financial statements are £1,240.5 million. There is a risk that this investment may be under-or overstated in the accounts.

Given a tolerance of say +/-5% around the net asset values on which the valuation is based, this would equate to a tolerance of +/-£62.0 million.

Hedge Funds

Uncertainties

Hedge funds valued at the sum of the fair values provided by the administrators of the underlying funds plus adjustments that the directors or independent administrators judge necessary. Where these investments are not publicly listed there is a degree of estimation involved in the valuation.

Effect if actual results differ from assumptions

The total hedge funds value in the financial statements is £217.5 million. There is a risk that these investments may be under-or overstated in the accounts. Given a tolerance of say +/-5% around the net asset values on which the valuation is based, this would equate to a tolerance of +/-£10.9 million.

age

Note P5 - Actuarial Valuation of the Fund

"A full actuarial valuation of the Fund was made as at 31 March 2013 by the Fund's Actuary, P Middleman of Mercer Human Resource Consulting Limited. The Actuary has determined the contribution rates with effect from 1 April 2014 to 31 March 2017.

On the basis of the assumptions adopted, the valuation revealed that the value of the Fund's assets of £9,886.0 million represented 70% of the Funding Target of £14,091.0 million at the valuation date. The valuation also showed that a common rate of contribution of 13.3% of pensionable pay per annum was required from employers. The common rate is calculated as being sufficient, together with contributions paid by members, to meet all liabilities arising in respect of service after the valuation date. It allows for the new LGPS benefit structure effective from 1 April 2014."

Adopting the same method and assumptions as used for calculating the funding target, the deficit could be eliminated by an average additional contribution rate of 10.3% of pensionable pay for 22 years.

In practice, each individual employer's position is assessed separately and the contributions required are set out in the report dated 31 March 2014. In addition to the certified contribution rates, payments to cover additional liabilities arising from early retirements (other than ill-health retirements) will be made to the Fund by the employers.

The funding plan adopted in assessing the contributions for each individual employer is in accordance with the Funding Strategy Statement (FSS). Different approaches adopted in implementing contribution increases and deficit recovery periods are as determined through the FSS consultation process. For certain employers, in accordance with the FSS, an increased allowance has been made for assumed investment returns on existing assets and future contributions, for the duration of the employer's deficit recovery period.

As a result of the valuation, a revised Rates and Adjustments certificate was prepared for the three years commencing 1 April 2014. For comparison purposes, the figures for the two preceding years are also shown. The rates payable by the Unitary Authorities were certified as follows:

Future Service Rate (% of page 1)	y) plus lump sum (£)				
	2012/13	2013/14	2014/15	2015/16	2016/17
Birmingham City Council	12.1% plus £27,800,000	12.1% plus £29,100,000	12.3% plus £40,113,600	"12.9% plus	12.3% plus £40,113,600
Coventry City Council	12.1% plus £6,600,000	12.1% plus £6,900,000	£41,870,40	13.4% plus	£41,870,400
Dudley MBC	11.8% plus £5,700,000	11.8% plus £6,000,000	£43,724,80		£43,724,800
Sandwell MBC	11.7% plus £7,900,000	11.7% plus £8,300,000	12.2% plus £9,467,000	12.7% plus	12.2% plus £9,467,000
Solihull MBC	11.7% plus £4,300,000	11.7% plus £4,500,000	£12,395,00	13.1% plus	£12,395,000
Walsall MBC	11.7% plus £8,000,000	11.7% plus £8,400,000	£15,518,000		£15,518,000
Wolverhampton City Council	12.2% plus £7,400,000	12.2% plus £7,800,000	12.1% plus £7,418,000	12.7% plus	12.1% plus £7,418,000

The valuation was carried out using the projected unit actuarial method and the main actuarial assumptions used for assessing the Funding Target and the common contribution rate were as follows:

	For past service liabilities	For future service liabilities
Rate of return on investments:	6.5% per annum	6.75% per annum
Rate of pay increases:	4.35% per annum*	4.35% per annum
Rate of increases in pensions in payment (in excess of Guaranteed Minimum Pension):	2.6% per annum	2.6% per annum
Rate of return on investments:	6.5% per annum	6.75% per annum

^{*} allowance was also made for short-term public sector pay restraint over a 3/5 year period depending on the individual employer.

The assets were assessed at market value.

The next triennial actuarial valuation of the Fund is due as at 31 March 2016. Based on the results of this valuation, the contribution rates payable by the individual employers will be revised with effect from 1 April 2017.

Actuarial Present Value of Promised Retirement Benefits for the Purposes of IAS 26

IAS 26 requires the present value of the Fund's promised retirement benefits to be disclosed, and for this purpose the actuarial assumptions and methodology used should be based on IAS 19 rather than the assumptions and methodology used for funding purposes.

To assess the value of the benefits on this basis, the following financial assumptions have been used:

	31 March 2013	31 March 2014
Rate of return on investments (discount rate)	4.2% per annum	4.5% per annum
Rate of pay increases	4.15% per annum	4.15% per annum*
Rate of increases in pensions in payment (in excess of Guaranteed Minimum Pension)	2.4% per annum	2.4% per annum

^{*} includes a corresponding allowance to that made in the actuarial valuation for short-term public sector pay restraint.

The value of the Fund's promised retirement benefits for the purposes of IAS26 as at 31 March 2013 was estimated as £15,611.0 million. The effect of the changes in actuarial assumptions between 31 March 2013 and 31 March 2014 as described above is to decrease the liabilities by c£1,249.0 million. Adding interest over the year increases the liabilities by c£656.0 million, and allowing for net benefits accrued/paid over the period increases the liabilities by another c£63.0 million (including any increase in liabilities arising as a result of early retirements/augmentations). Finally, allowing for actual verses expected membership experience, which emerged at the 2013 valuation, gives a reduction in liabilities of c£401.0 million.

The net effect of all the above is that the estimated total value of the Fund's promised retirement benefits as at 31 March 2014 is therefore £14,680.0 million.

Note P6 - Taxation

1. Value Added Tax

The Fund pays VAT collected on income in excess of VAT payable on expenditure to HMRC. The accounts are shown exclusive of VAT.

2. Taxation of Overseas Investment Income

The Fund receives interest on its overseas bonds gross, but a variety of arrangements apply for the taxation of dividends on overseas equities in the various markets. Where relief is available it may be either in full at source (USA, Belgium, Australia Finland and Norway), or partial relief by claim (Austria, Denmark, France, Germany, Netherlands, Switzerland and Spain).

In some markets (Poland, Canada, Italy, and Sweden) tax is deducted at the treaty rate so that no further adjustment is required, and there are also markets (Malaysia, Hong Kong and Singapore) where no double taxation agreements exist and where the full amount is payable.

Note P7 - Contributions Receivable

Contributions receivable are analysed below:-

	2012/13 £m		2013/14 £m
		From Employers	
Ъ	294.8	Basic Contributions	300.3
a	-	Deficit Funding	-
age	0.5	Augmented Membership	0.3
7	8.3	Additional Cost of Early Retirement	8.8
06	303.6		309.4
		From Employees	
	106.4	Basic Contributions	108.9
	0.9	Additional Contributions	1.0
	107.3		109.9
	410.9	Total Contributions	419.3

The additional contributions above represent the purchase of added membership or additional benefits under the Pension Scheme and are included in the revenue accounts.

One admitted body, Black Business in Birmingham, terminated their agreement in December 2008 and is now in administration with an outstanding liability identified by the actuary of £128,200.

Payments can be analysed by type of Member Body as follows:-

2012/13		2013/14
£m		£m
30.8	Administering Authority	32.9
361.4	Scheduled Employers	367.8
18.7	Admitted Employers	18.6
410.9	Total	419.3

Note P8 - Transfers In

Pac	2012/13 £m		2013/14 £m
ĕ	22.6	Individual transfers in from other schemes	11.3

Note P9 - Other Income

2012/13		2013/14
£m		£m
	Benefits Recharged to Employers	
9.1	Compensatory Added Years	9.0
7.1	Pensions Increases	7.2
-	Magistrates Courts Committee	-
16.2	Total	16.2

Note P10 – Benefits Payable

An analysis of expenditure on benefits by type is given below:-

2012/13 £m		2013/14 £m
	Pensions	
328.8	Retirement Pensions	346.0
25.7	Widows' Pensions	26.5
1.0	Children's' Pensions	0.9
3.1	Widowers' Pensions	3.5
0.1	Ex-Spouse	0.1
0.1	Equivalent Pension Benefits	0.1
-		-
-		-
358 8	Total Pensions	377.1
330.0		
	Lump Sum Benefits	
74.6	Retiring Allowances	90.3
11.7	Death Grants	11.8
86.3	Total Lump Sum Benefits	102.1
445.1	Total Benefits Payable	479.2

The total benefits payable can be analysed by type of Member Body as follows:-

2012/13 £m		2013/14 £m
38.1	Administering Authority	42.2
379.7	Scheduled Employers	406.2
27.3	Admitted Employers	30.8
445.1	Total	479.2

Note P11 - Payments To and On Account of Leavers

	2012/13 £m		2013/14 £m
	12.8	Individual transfers out to other schemes	20.2
Pa	-	Refunds of Contributions	-
Ó	-	State Scheme Premiums	-
Ф —	3.0	Bulk Transfer Pension Increases	2.6
9	15.8	Total	22.8

Note P12 - Investment and Administration Expenses

Costs incurred in the management of the investments of the Fund and the administration of the Fund have been charged to the Fund in accordance with the Local Government Pension Scheme Regulations and can be analysed as follows:

	2012/13 £m		2013/14 £m
		Administration	
	4.8	Pensions Administration	4.2
	0.5	Actuarial fees	0.7
	-	Audit fees	0.1
	5.3	Total Administration	5.0
Page			
ğ		Investments	
υ -	8.8	External management of investments	
94	2.1	In-house management of investments	7.1
44	0.1	Property and legal fees	2.1
	0.3	Safe Custody Expenses	-
	11.3	Total Investments	0.2

Performance related fees are negotiated with a number of managers. Included in external management of investments are performance related fees of £1.3 million in 2013/14 and £2.2 million in 2012/13.

Note P13 - Investment Income

Investment income is analysed below:-

	2012/13 £m		2013/14 £m
	~~~	Dividends and Interest	
		Fixed Interest Securities	
	8.5	UK Private Sector – Quoted	8.8
		Equities	
	33.2	UK	31.9
	43.6	Overseas	43.0
		Pooled Investment Vehicles	
	22.8	UK	19.8
<u>כ</u>	-	UK - Re-invested Income, prior years	-
	1.8	1.8 Overseas Equities	
	0.1	Private Equity	-
	0.9	Interest on Cash Deposits	1.3
ŽĮ.	0.9	Stock lending	0.6
	-	UK Tax, Irrecoverable	(0.1)
	(1.9)	Overseas Taxation	(1.4)
	109.9	Total Dividends and Interest	105.8
	35.7	Property Management Income	37.4
	(8.8)	Property Management Expenses	(9.8)
	26.9	Total Property Management	27.6
	136.8	Total Investment Income	133.4

# Page 196

# 9. WEST MIDLANDS PENSION FUND STATEMENTS

# Stocklending

The stock lending programme provides for direct equity investments to be lent. At the year end the value of quoted equities on loan was £168.1M (2013: £43.9M) in exchange for which the custodian held collateral worth £179.1M (2013: £50.5M). Collateral consists of acceptable securities and government debt.

## **Note P14 - Net Investment Assets**

Further analysis of the market value of investments as set out in the Net Assets Statement is given below:-

Segregated accounts are held separately from the main account by the global custodian and contain assets managed by some of the Fund's external managers.

31 March 2013		31 March 2014
£m	Fixed Interest Securities	£m
470.0		474.0
173.9	UK Companies – Segregated (external)	171.3
173.9		171.3
	UK Equities	
943.5	Quoted	971.3
943.5		971.3
	Overseas Equities	
1,713.4		1,810.6
359.1	Quoted – Segregated (external)	499.4
2,072.5		2,310.0
	Pooled Investment Vehicles	
	Managed Funds	
196.4	UK Quoted, Fixed Interest	192.5
196.4	Other Fixed Interest	600.8
963.2	UK Quoted, Index Linked	948.6
858.8	Overseas Equities	1,105.6
255.5	UK Unquoted Equities	269.6
1,509.7	Overseas Unquoted Equities	1,453.9
593.0	UK Absolute Returns	553.0
105.9	Overseas Absolute Returns	89.6
37.6	UK Property	47.0
225.7	Foreign Property	211.3
	Unit Trusts	
26.6	UK Quoted Equities	66.3
303.6	Overseas Equities	209.4

31 March 2013		31 March 2014
£m		£m
7.0	Overseas Property	6.5
5,729.4		5,754.1
	Property	
532.7		586.8
34.9	UK Leasehold*	43.0
567.6		629.8
	Foreign Currency Holdings	
29.6	United States Dollars	21.3
34.0	Euro	11.3
1.5	Canadian Dollars	1.0
1.1	1.1 Danish Kroner	
0.4	0.4 Hong Kong Dollars	
3.2	Swedish Kroner	0.5
4.3	Swiss Francs	1.4
2.0	Japanese Yen	1.4
1.2	Norwegian Kroner	0.5
-	Malaysian Ringits	-
-	Singapore Dollars	0.6
1.1	Australian Dollars	1.1
-	New Zealand Dollars	0.1
1.2	Hungarian Florints	0.5
0.9	Polish Zloty	0.7
-	Israeli Shekels	-
1.2	Turkish Lira	0.3
1.2	Czech Koruna	0.5
-	Korean Won	-

31 March 2013		31 March 2014
£m		£m
82.9		42.2
	Cash Deposits	
241.1	UK	221.3
	Other Investments	
(0.1)	Broker Balances	(1.6)
15.4	Outstanding Dividend Entitlement and Recoverable with-holding Tax	5.8
9,826.2	Total Net Investment Assets	10,104.2

^{*} All leasehold properties are held on long leases (more than 15 years)

Page 1		·	n 5% of the net assets of the scheme:		
$\overline{\circ}$	31st M	arch 2013		31st March	1 2014
99	Market Value	% of total Market Value		Market Value	% of total Market Value
	£M	%		£M	%
			Security		
	664.0	6.8	Legal & General - All Stocks Index-Linked Gilts Fund	639.1	6.3

The proportion of the market value of investment assets managed in-house and by each external manager at the year-end is set out below.

	31 March 2013			31 March	2014
	Market Value	% of total Market Value		Market Value	% of total Market Value
	£m	%		£m	%
	3497.3	35.6	In-house	3,632.7	36.0
	26.6	0.3	Managers: UK Quoted	33.2	0.3
	131.1	1.3	Managers: US Quoted	146.5	1.4
	193.3	2.0	Managers: European Quoted	236.8	2.3
	37.6	0.4	Managers: Japanese Quoted	36.2	0.4
	107.4	1.1	Managers: Pacific Basin	69.5	0.7
ַ	609.3	6.2	Managers: Emerging Markets	845.9	8.4
5	493.9	5.0	Managers: Global Equities	526.1	5.2
	1,979.9	20.2	Managers: Fixed Interest	1,913.2	18.9
)  - 	270.3	2.8	Managers: Indirect Property	298.0	2.9
≾⊺	-	0.0	Managers: Emerging Market Debt	189.6	1.9
	206.8	2.1	Managers: Commodities	293.4	2.9
	326.5	3.3	Managers: Infrastructure Funds	642.6	6.4
	698.9	7.1	Managers: Absolute Return	1,240.5	12.3
	1,231.9	12.6	Managers: Private Equity	-	-
	9,810.8	100.0		10,104.2	100.0
	15.4		Outstanding Dividend Entitlement and Recoverable with-holding Tax	-	
	9,826.2		Total Investment Assets	10,104.2	

# **Note P15 – Investment Market Value Movements Analysis**

The change in the value of investments during 2013/14 is set out below:-

	Value at 1 April 2013 £m	Purchases at Cost £m	Sales Proceeds £m	Change in Market Value £m	Value at 31 March 2014 £m
Fixed Interest Securities	173.9	-	-	(2.6)	171.3
UK Equities	943.5	28.9	(25.9)	24.8	971.3
Overseas Equities	2,072.5	249.1	(80.7)	69.1	2,310.0
Pooled Investment Vehicles	5,729.4	1,881.1	(1,809.1)	(47.3)	5,754.1
Property	567.6	21.5	(2.5)	43.2	629.8
	9,486.9	2,180.6	(1,918.2)	87.2	9,836.5
Broker Balances	(0.1)				(1.6)
Outstanding dividend entitlement and recoverable Withholding tax	15.4				5.8
Foreign Currency	82.9				42.2
Cash Deposits	241.1				221.3
Total Investments	9,826.2				10,104.2

The change in market value of investments comprises increases and decreases in the market value of investments held at any time during the year. The profit and losses on the sale of investments shown in the Fund Account includes an additional £103.8 million which represents profit realised on sale of the Fund's assets.

Purchases also include transfers in of investments, take-over of shares etc. and invested income. Sales proceeds include all receipts from sales of investments, transfers out of investments, take-over proceeds etc. and reductions in cash deposits including profits or losses realised on the sale.

There were 125 late payments amounting to £2.3 million of contributions during the year which constituted employer-related investments until the amounts were received. Other than this, there were no employer-related investments.

# Page 202

# 9. WEST MIDLANDS PENSION FUND STATEMENTS

Transaction costs are included in the cost of purchases and sale proceeds. Transaction costs include costs charged directly to the scheme such as fees, commissions, stamp duty and other fees. Transaction costs during the year amounted to £0.2 million (2012/13: £0.4 million). In addition to the transaction costs disclosed above, indirect costs are incurred through the bid-offer spread of investments within pooled investment vehicles. The amount of indirect costs is not separately provided to the scheme.

31 March 2013		31 March 2014
£m		£m
0.1	Equities - UK Quoted	0.1
0.3	Equities - Overseas Quoted	0.1
0.4	Total	0.2

The volatility of investment markets is an ever-present and longstanding feature of pension fund management and valuations may vary, either up or down, throughout each day when exchanges are open.

The change in the value of investments during 2012/13 is set out below:-

The change in the value of investments during 2012/13 is set out below:-

	Value at 31 March 2012 £m	Purchases at Cost £m	Sales Proceeds £m	Change in Market Value £m	Value at 31 March 2013 £m
Fixed Interest Securities	158.8	-	-	15.1	173.9
UK Equities	840.5	30.2	(22.8)	95.6	943.5
Overseas Equities	1,779.3	186.8	(154.5)	260.9	2,072.5
Pooled Investment Vehicles	5,213.3	817.6	(800.9)	499.4	5,729.4
Property	615.4	13.4	(36.2)	(25.0)	567.6
	8,607.3	1,048.0	(1,014.4)	846.0	9,486.9
Broker Balances	0.2				(0.1)
Outstanding dividend entitlement and recoverable Withholding tax	11.1				15.4
Foreign Currency	43.7				82.9
Foreign Currency Cash Deposits	139.3				241.1
Total Investments	8,801.6				9,826.2

The change in market value of investments comprises increases and decreases in the market value of investments held at any time during the year. The profit and losses on the sale of investments shown in the Fund Account includes an additional £97.7 million which represents profit realised on sale of the Fund's assets.

Net gains and losses on financial instruments

31 March 2013		31 March 2014
£m		£m
	Financial Assets	
846.0	Fair value through profit and loss	87.2
846.0	Total	87.2

# Page 204

# 9. WEST MIDLANDS PENSION FUND STATEMENTS

Fair value of financial instruments and liabilities

The following table summarises the carrying values of the financial assets and financial liabilities by class of instrument compared with their fair values.

31 March	2013		31 Ma	rch 2014	
Carrying Value Fair Value			Carrying Value	Fair Value	
£m	£m		£m	£m	
		Financial Assets			
9,486.9	9,486.9	Fair value through profit and loss	9,836.5	9,836.5	
339.3	339.3	Loans and receivables	267.7	267.7	
9,826.2	9,826.2	Total	10,104.2	10,104.2	

Valuation of Financial Instruments Carried at Fair Value

The valuation of financial instruments has been classified into three levels, according to the quality and reliability of information used to determine fair values. Criteria utilised in the instrument classifications are detailed below:

## Level 1

Financial instruments at Level 1 are those where the fair values are derived from unadjusted quoted prices in active markets for identical assets or liabilities. Products classified as level 1 comprise quoted equities, quoted fixed securities, quoted index linked securities and unit trusts. Listed investments are shown at bid prices. The bid value of the investment is based on the bid market quotation of the relevant stock exchange.

### Level 2

Financial instruments at Level 2 are those where quoted market prices are not available; for example, where an instrument is traded in a market that is not considered to be active, or where valuation techniques are used to determine fair value and where these techniques use inputs that are based significantly on observable market data. "

## Level 3

Financial instruments at Level 3 are those where at least one input that could have a significant effect on the instrument's valuation is not based on observable market data. Such instruments would include unquoted equity investments and hedge fund of funds, which are valued using various valuation techniques that require significant judgement in determining appropriate assumptions.

The values of the investment in private equity are based on valuations provided by the general partners to the private equity funds in which West Midlands Pension Fund has invested. These valuations are prepared in accordance with the International Private Equity and Venture Capital Valuation Guidelines, which follow the valuation principles of IFRS and US GAAP. The valuation of unquoted securities is based on the latest investor reports and financial statements provided by the fund managers of the underlying funds, adjusted for transactions arising after the date of such reports as appropriate.

The values of the investment in hedge funds are based on the net asset value provided by the fund manager. Assurances over the valuation are gained from the independent audit of the value. The following table provides an analysis of the financial assets and liabilities of the pension fund grouped into Levels 1 to 3, based on the level at which the fair value is observable.

Value at 04 March 0044	Quoted Market Price	Using Observable Inputs	With significant unobservable inputs	
Values at 31 March 2014	Level 1	Level 2	Level 3	Total
	£m	£m	£m	£m
Financial Assets and Liabilities	5,941.9	1,528.6	2,366.0	9,836.5
Financial Assets at fair value through profit and loss	267.7	-	-	267.7
Loans and receivables	6,209.6	1,528.6	2,366.0	10,104.2
Total Financial assets	5,941.9	1,528.6	2,366.0	9,836.5

Values at 31 March 2013	Quoted Market Price	Using Observable Inputs	With significant unobservable inputs	Total
values at 31 march 2013	Level 1	Level 2	Level 3	Total
	£M	£M	£M	£M
Financial Assets and Liabilities				
Financial Assets at fair value through profit and loss	5,538.4	1,484.3	2,464.2	9,486.9
Loans and receivables	339.3	-	-	339.3
Total Financial assets	5,877.7	1,484.3	2,464.2	9,826.2

# **Note P16 – Investment Capital Commitments**

Investment commitments at the end of the financial year in respect of future payments were:-

31 March 2013		31 March 2014
£m		£m
808.7	Non-Equities	920.8
139.0	Property	146.5
947.7	Total	1,067.3

These commitments relate to outstanding commitments due on funds held in the private equity, property and infrastructure portfolios.

### **Note P17- Current Assets**

31 March 2013 £m		31 March 2014 £m
	Receivables and Prepayments	
	Contributions Receivable	
21.6	- Employers	-
8.4	- Employees	-
42.5	Wolverhampton City Council	-
-	Other Receivables	67.2
72.5	Total Receivables and Prepayments	67.2
0.7	Cash	(0.1)
73.2	Total Current Assets	67.1

Note:- Following the bulk transfer of Magistrates Courts Committee staff to the Civil Service Pension Scheme on 31 March 2005, it has now been calculated by Mercer Limited that the Fund is due to receive a total of £27.7 million. This is to be paid in 10 equal and annual instalments commencing on 15 April 2011 and finishing on 15 April 2020 together with interest payments resulting in annual income of £3.3 million. The balance due included in Other Receivables is £19.4 million (2012/13: £19.4 million). During 2013/14 no payments were received and deducted from the debtor as the payment for 2013/14 was made early (in March 2013).

31 March 2013 £m		31 March 2014 £m
	Analysis of Receivables	
-	Central Government Bodies	
51.5	Other Local Authorities	21.5
-	Public Corporations	-
21.0	Other Entities and Individuals	45.7
72.5	Total	67.2

# **Note P18- Current Liabilities**

31 March 2013		31 March 2014
£m		£m
	Payables and Receipts In Advance	
(4.1)	Pensions and Lump Sum Benefits	(7.1)
(8.9)	Other Payables	(23.7)
(0.1)	Trustee Account	-
-	Bulk Transfer Pension increases	-
(13.1)	Total	(30.8)

Pag	31 March 2013 £m		31 March 2014 £m
Эe		Analysis of Payables	
209	(3.2)	Central Government Bodies	(8.2)
	(0.1)	Other Local Authorities	(0.1)
	(9.8)	Other Entities and Individuals	(22.5)
	(13.1)	Total	(30.8)

# **Note P19 - Additional Voluntary Contributions**

As well as joining the fund, scheme members can pay into an additional voluntary contribution (AVC) scheme run by two AVC providers. Contributions are paid directly from scheme members to the AVC providers.

The contributions are not included within the fund accounts, in line with regulation 4 (2) (c) of the Pension Scheme (Management and Investment of Funds) Regulations 2009. The table below shows the activity for each AVC provider in the year.

	31 March 2013			31 March 2014	
	Equitable Life Prudential			Equitable Life	Prudential
_	£m	£m		£m	£m
ag	2.9	27.5	Opening Value of the Fund	2.7	30.2
)e 2	-	7.3	Income	-	7.3
1	(0.4)	(5.0)	Expenditure	(0.4)	(7.8)
9	0.2	0.4	Change in Market Value	0.1	6.1
	2.7	30.2	Closing Value of the Fund	2.4	35.8

### Note P20 - Post Year End Transactions

The responsibility for the pension administration of the Staffordshire and West Midlands Probation Trust will transfer to Greater Manchester Pension Fund on 1 June 2014. The draft Local Government Pension Scheme (Offender Management) Regulations 2014 set out the basis of the bulk transfer calculation, together with specified payment dates. The valuation of the bulk transfer and the associated payment will be determined in 2014/15 between the Actuaries of West Midlands Pension Fund and Greater Manchester Pension Fund. No adjustment is required to be made in the 2013/14 financial statements.

# Page 211

# 9. WEST MIDLANDS PENSION FUND STATEMENTS

# Note P21 - The Nature and Extent of Risks Arising From Financial Instruments

Risk Management

The Fund's activities expose it to a variety of financial risks including:

Investment Risk - the possibility that the Authority will not receive the expected returns.

Credit Risk - the possibility that the other parties might fail to pay amounts due to the Authority.

Liquidity Risk - the possibility that the Authority might not have funds available to meet its commitments to make payments.

Market Risk - possibility that financial loss might arise as a result of stock market movements. Currency risk, other price risk and interest rate risk are types of market risk.

The Fund's overall risk management programme focuses on the unpredictability of financial markets and seeks to minimise potential adverse effects on the resources available to fund services. Policies covering specific areas relating to the Pension Fund are as follows:

## **Investment Risk**

In order to achieve its statutory obligations to pay pensions, the Fund invests its assets, including employer and employee contributions, in a way that allows it to meet its liabilities as they fall due for payment. It does this by investing with regard to liabilities through the triennial actuarial valuation followed by an appropriate asset allocation. During the year, the Fund targeted a 90% exposure to return seeking assets such as equities, property, other alternatives with equity-like returns, including emerging market debt and higher return fixed interest investments. The remaining 10% being allocated to stabilising assets, such as UK Government bonds or gilts, both index linked and conventional.

Risks in return seeking assets include market risk (the greatest risk), issuer risk and volatility, which are partly mitigated by diversification across asset classes, global markets and investments funds. Mitigating interest rate risk and inflation risk points to significant investment in bonds, but doing so at the expense of return seeking assets would increase the costs of funding. Stabilising assets backed by the UK Government are considered low risk. However, corporate bonds carry some additional issuer risk.

# Counterparty Risk

In deciding to effect any transaction for the Fund, considerable steps are taken to ensure that the counterparty is suitable and reliable; that the transaction is in line with the Fund's strategy and that the terms and circumstances of the transaction are the best available in the relevant market at the time. Comprehensive due diligence processes are in place to ensure that any potential counterparty is authorised and regulated, competent to deal in investments of the type and size contemplated and has appropriate administration arrangements with regard to independent auditors, robust administration and accounting, relevant legal structure and experienced staff.

Legal agreements are implemented and continuous monitoring of counterparties is undertaken by Fund officers in relation to suitability and performance, in addition to compliance with regulatory and Fund specific requirements.

Credit Risk

The Fund's deposits with financial institutions as at 31 March 2014 totalled £211.6 million in respect of temporary loans and treasury management instruments. The Fund's surplus cash may be placed with an approved financial institution on a short-term basis and in accordance with the cash management policy and restrictions set out in the Compliance Manual. The policy specifies the cash deposit limit with each approved counterparty, as determined by a comprehensive scoring exercise undertaken by Fund officers using specialist rating and market research data, which is reviewed on a regular basis.

Proposed counterparties are assessed using an amalgamation of credit ratings and market research with the resulting 'score' determining the suitability and individual limit in each case. Due diligence is conducted on potential money market funds with criteria such as AAA rating, same day access and minimum assets under management being prerequisite. A credit rating sensitivity analysis as at 31 March 2014 is shown below:

	Credit Rating Sensitivity Analysis			
	Summary	Rating	Balances as at 31 March 2014	Balances as at 31 March 2013
			£m	£m
	Money Market Funds			
	AIM STIC Global Sterling Portfolio	AAA	79.4	28.3
	HSBC Sterling Liquidity Fund	AAA	81.7	43.3
	Northern Trust Global Sterling Fund	AAA	1.2	-
	Short-term Deposits			
	Nationwide Building Society	A	-	25.0
	Banco Santander	Α	8.0	23.5
	Lloyds Bank PLC	A	-	13.0
	Coventry Building Society	Α	5.0	12.0
	Skipton Building Society	BBB-	-	8.5
Ď	Principality Building Society	BBB+	-	8.0
Page	Newcastle Building Society		5.0	
	Barclays		9.0	-
21	Bank Deposit Accounts			
ယ	Nat West Liquidity Select	Α	50.0	50.0
	Total		239.3	211.6

#### Liquidity Risk

The Fund has a comprehensive daily cash flow management procedure which seeks to ensure that cash is available as needed. Due to the cash flow management procedures and the liquidity of certain asset types held, there is no significant risk that the Fund will be unable to raise cash in order to meet its liabilities. The Fund actually uses this liquidity risk to its benefit, taking advantage of the illiquidity premium found in investments such as private equity.

#### Foreign Exchange Risk

The Fund's exposure to foreign exchange risk is managed through the diversification of portfolios across sectors, countries and geographic regions, along with continuous monitoring and management of holdings. In addition, the Fund's currency exposure is managed in line with the daily cash management policy.

#### Securities Lending

As at 31 March 2014, £171.4 million of stock was on loan to an agreed list of approved borrowers through the Fund's custodian in its capacity as agent lender. The loans were covered by non-cash collateral in the form of equities, gilts, DBVs and G10 sovereign debt, totalling £184.6 million, giving a margin of 7.71%.

Collateral is marked to market, adjusted daily and held by a tri-party agent on behalf of the Fund. Income from stock lending amounted to £0.6 million during the year and is detailed in Note 13 to the accounts. The Fund retains its economic interest in stocks on loan, and therefore the value is included in the Fund valuation. There is, however, an obligation to return collateral to the borrowers; therefore, its value is excluded from the Fund valuation. The securities lending programme is indemnified, giving the Fund further protection against losses.

# Name Page Pa

The Fund's prudent approach to the collective risks listed above and through best practice in corporate governance, ensures that reputational risk is kept to a minimum.

#### Other Price Risk

Other price risk represents the risk that the value of a financial instrument will fluctuate as a result of changes in market prices (other than those arising from interest rate risk or foreign exchange risk), whether those changes are caused by factors specific to the individual instrument or its issuer, or factors affecting all such instruments in the market. The Fund is exposed to share and derivative price risk, which arises from investments held by the fund for which the future price is uncertain. The Fund mitigates price risk through diversification and the selection of securities and other financial instruments is monitored by the council to ensure it is within limits specified in the Fund investment strategy.

Other price risk - sensitivity analysis

Following analysis of historical data and expected investment return movement during the financial year, in consultation with the fund's performance advisors, the Fund has determined that the following movements in market price risk are reasonably possible for the 2013/14 reporting period:

#### Price Risk

	Price Risk				
	Asset Type	Value as at 31 March 2014 £	% Change	Value on Increase £	Value on Decrease £
-	UK equities	1,004.3	16.6%	1,171.0	837.6
	Global equities (ex UK)	3,625.0	19.4%	4,328.3	2,921.8
	Property	927.8	14.7%	1,064.2	791.4
	Corporate bonds (short term)	93.7	7.2%	100.4	87.0
7	Corporate bonds (medium term)*	620.4	9.6%	680.0	560.8
a	Corporate bonds (long term)	168.6	18.0%	198.9	138.3
age	UK fixed gilts (short term)	104.0	3.1%	107.2	100.8
N	UK fixed gilts (medium term)**	117.3	6.9%	125.4	109.2
15	UK fixed gilts (long term)	45.7	12.3%	51.3	40.1
	UK index linked gilts (short term)	70.9	2.3%	72.5	69.3
	UK index linked gilts (medium term)	201.3	5.0%	211.4	191.2
	UK index linked gilts (long term)	366.8	8.5%	398.0	335.6
	Commodities	189.6	13.7%	215.6	163.6
	Cash	173.7	0.6%	174.7	172.7
	Private Equity	1,240.5	28.4%	1,592.8	888.2
	Infrastructure	293.4	15.4%	338.6	248.2
	High Yield Debt***	214.5	13.2%	242.8	186.2
	Absolute Return/Diversified Growth	642.5	11.8%	718.3	566.7
	Total Assets	10,100.0	16.6%	11,791.4	8,408.6

^{*}includes exposure Emerging Market Debt (£263.7 million), Loans (£90.0 million) and the Newton Dynamic Bond Fund (£48.1 million)

^{**}includes exposure to Overseas Bonds (£74.5 million)

#### Currency Risk - Sensitivity Analysis

Currency risk represents the risk that the fair value of future cash flows of a financial instrument will fluctuate because of changes in foreign exchange rates. The Fund is exposed to currency risk on financial instruments that are denominated in any currency other than the functional currency of the Fund (£UK). The Fund holds both monetary and non-monetary assets denominated in currencies other than £UK. The following tables summarise the Fund's currency exposure as at 31 March 2014:

#### Currency Risk (by asset class)

	Asset Type	Value as at 31 March 2014 £	% Change	Value on Increase £	Value on Decrease £
D	Overseas Equities	3,625.0		11.8%	4,052.8
aa	Private Equity	1,055.7		11.8%	1,180.3
Ө	Fixed Interest	263.7		11.8%	294.8
21	Alternatives	574.5		11.8%	642.3
6	Property	217.8		11.8%	243.5
	Liquid Assets	42.2		11.8%	47.2
	Total	5,778.9		11.8%	6,460.8

#### Interest Rate Risk

The fund invests in financial assets for the primary purpose of obtaining a return on investments. These investments are subject to interest rate risks, which represent the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates.

The fund's direct exposure to interest rate movements as at 31 March 2014 is set out below. These disclosures present interest rate risk based on the underlying financial assets at fair value:

^{***}includes Mezzanine debt and Convertibles

Interest rate risk - sensitivity analysis

The council recognises that interest rates can vary and can affect both income to the fund and the value of the net assets available to pay benefits. The fund's consulting actuary has advised that the assumed interest rate volatility is 100 basis points per annum.

The analysis that follows assumes that all other variables, in particular exchange rates, remain constant, and shows the effect in the year on the net assets available to pay benefits of a +/-100 BPS change in interest rates:

	As at 31 March 2013 £m	Asset Type	As at 31 March 2014 £m
	239.3	Cash & Cash Equivalents	211.6
	84.7	Cash Balances	52.1
U V	1,034.3	Fixed Interest Securities	906.0
5	1,358.3	Total	1,169.7

Asset Type	Carrying amount as at 31 March 2014	Change in year in the net bene	
		+180BPS	-180BPS
	£m	£m	£m
Cash & Cash Equivalents	211.6	2.1	(2.1)
Cash Balances	52.1	0.5	(0.5)
Fixed Interest Securities	906.0	9.1	(9.1)
Total change in assets available	1,169.7	11.7	(11.7)

#### Note P22- Impairment for Bad and Doubtful Debts

The following additions and write offs of pension payments were reported in this financial year, in line with the Fund's policy:

Additions Analysis		
Individual Value	Number	Total £
Less than £50	10	183.50
£50 - £100	1	69.30
£100 - £500	1	128.94
Over £500	0	0.00
Total	12	381.74

Page	Write off Analysis		
ğ	Individual Value	Number	Total £
\ \	Less than £50	7	165.58
$\frac{2}{5}$	£50 - £100	28	2,061.82
W	£100 - £500	47	11,164.70
	Over £500	12	16,704.97
	Total	94	30,097.07

#### **Note P23 - Related Parties**

The pensions administration function and the in-house management of investments are performed by Wolverhampton City Council and the costs shown in the table above are recharged to the Pension Fund each year on an estimated basis with an end of year adjustment for actual costs shown as receivable or payable in the accounts. This is a related party transaction as Wolverhampton City Council is also a member body of the Fund.

Key management personnel who are employees of the administering authority and members of the Fund are disclosed in the administering authority's statement of accounts along with details of remuneration and pensions contributions.

There is one member of the pension fund committee who is in receipt of pension benefits from the West Midlands Pension Fund (Councillor Turner). There are six committee members who are active members of the Pension Fund (Councillors S.Eling, M.Evans, S.Evans, L.McGregor, T.Singh and V.Silvester). Each member of the Pension Fund Committee is required to declare their interests at each meeting.

² age	
Ń	Е
19	S
	٧

Ũ	Scheduled Bodies		
DE	District Councils		
Œ			
Ò	Birmingham City Council	Coventry City Council	Dudley Metropolitan Borough Council
19	Sandwell Metropolitan Borough Council	Solihull Metropolitan Borough Council	Walsall Metropolitan Borough Council
	Wolverhampton City Council		
	Major Employers		
	Centro	Staffordshire & West Midlands Probation Trust	West Midlands Fire & Civil Defence Authority
	West Midlands Police Authority		
	Universities (former Polytechnics)		
	Birmingham City University	Coventry University	University of Wolverhampton (The)
	Colleges of Further Education and Higher Education		
	Birmingham Metropolitan College	Bournville College of Further Education	Cadbury Sixth Form College

# age 220

Scheduled Bodies		
City College, Birmingham	City College, Coventry	City of Wolverhampton College
Dudley College of Technology	Halesowen College	Henley College
Hereward College	Joseph Chamberlain College	King Edward VI College
Sandwell College	Sixth Form College Solihull (The)	Solihull College
South and City College Birmingham	Stourbridge College	University College Birmingham
Walsall College	Wolverhampton College	
Schools		
King Edward VI Aston School*	King Edward VI Camp Hill School for Boys*	King Edward VI Camp Hill School for Girls*
King Edward VI Five Ways School*	King Edward VI Handsworth School*	
* Note - became Academies wef 1 August 2011		
J		
Other Bodies		
Alderbrook School	Aldersley High School	Aldridge School - a Science Co
Alexandra Academy	Arden Academy Trust	ARK Academies
Arthur Terry Learning Partners	Aston Manor Academy	Aston University Engineering A
Balsall Common Primary Academy	Balsall Parish Council	Barr Beacon School Trust
Bartley Green School	Billesley Primary School	Birmingham Museums Limited
Bishop Vesey's Grammar School	Black Country University Tech	Blakenhale Infants School
Blakenhale Junior School	Blue Coat Church of England Academy Limited (The)	BOA Birmingham Ormiston Academ
Caludon Castle School	Castle Bromwich Parish Council	Central Learning Partnership Trust (Heath Park Academy)
Charles Coddy Walker Academy -	Chelmsley Wood Town Council	Chilwell Croft Academy - Equit
City of Wolverhampton Academy Trust	Collegiate Academy Trust (The)	Coundon Court
Croft Primary Academy - Elliot	CTC Kingshurst Academy	Deanery Church of England School
Dorrington Academy Trust	E-ACT Heartlands Academy	E-ACT North Birmingham Academy
E-ACT Shenley Academy	E-ACT Willenhall Academy	Earls High School

# ²age 221

Scheduled Bodies		
EBN Free School	Education Central Multi Academy Trust	Erdington Hall Primary Academy
Fairfax School	Finham Park School Academy	Fordbridge Parish Council
Four Dwelling Primary Academy	Four Dwelling Secondary School	George Dixon Academy
Goldsmith Primary Academy	Grace Academy	Great Barr Primary School
Green Meadow Primary School	Greenholm Primary School	Greenwood Academy
Hall Green Secondary School	Handsworth Wood Girls School	Harborne Academy
Heart of England School	High Arcal School Academy	Hillcrest School and Sixth Form Centre
Hockley Heath Academy	Holly Hall Academy (The)	Holte School
Holyhead School	John Henry Newman Catholic College (The)	Joseph Leckie Academy Trust
Jubilee Academy Mossley - ATT	King Edward VI Aston School (A	King Edward VI Camp Hill Schoo
King Edward VI Sheldon Heath Academy	Kings Norton Girl's School and Language College	Kings Rise Community Primary School
Kingshurst Parish Council	Kingswinford School and Sc (The)	Langley School
Lea Forest Primary Academy	Light Hall School	Lode Heath School
Lordswood Boys School	Lordswood Girls School and Sixth Form Centre	Mansfield Green E-Act Academy
Meriden Parish Council	Merritts Brook E-Act Primary Academy	Mesty Croft Academy
Moseley Park School - Central	Nansen Primary School - Park V	Nechells Primary E-ACT Academy
Ninestiles Academy Trust	Oaklands Primary _Ninestiles Academy	Oldbury Academy
Oldknow Academy	Ormiston Academies Trust	Ormiston Forge Academy
Ormiston George Salter Academy	Ormiston Sandwell Community Academy	Park Hall Academy
Park Hall Infant Academy	Park Hall Junior Academy	Park View Educational Trust
Reedswood EACT Primary Academy	Rookery School	Rough Hay Primary
RSA Academy	Ryder Hayes Academy Trust	Sandwell Academy Trust Limited
Sandwell Leisure Trust	Shelfield Community Academy	Shire Oak Academy Trust
Shirestone Community Academy -	Short Heath School	Sidney Stringer Academy Trust
Slade Primary - The Arthur Ter	Smiths Wood Parish Council	Solihull Community Housing Limited
St Clements C of E Academy	St Johns C of E Primary School	St Marys C of E Junior & Infants School
St Michael's C.E. Primary School	St Michaels Junior and Infants	St Patrick's Church of England Primary

Scheduled Bodies		
		Academy
St. Peters Collegiate School	Streetly Academy (The)	Sutton Coldfield Grammar School for Girls Academy Trust
Tile Hill Wood School and Language College	Timberley Academy Trust	Tudor Grange Academy Solihull Trust
Tudor Grange Primary Academy St James	Valuation Tribunal Service (formerly Birmingham Valuation Tribunal)	Victoria Park Primary Academy
Walsall City Academy Trust Limited	Walsall College Academies Trust (The Mirus Academy)	Warren Farm Primary School
West Walsall E-ACT Academy	Westwood Academy (The)	Whitefriars Housing Group
Whitley Academy	Wilson Stuart School	Windsor High School and Sixth Form
Wolverhampton Homes	Wood Green Academy Trust	Woodlands Academy
Woodlands Academy of Learning	WoodView School	

Admitted Bodies		
With Active Members		
4 Towers TMO Limited		
ACUA Limited	Acivico (Building Consultancy)	Acivico (Design Construction a
Age Concern Wolverhampton	Age Concern Birmingham	Age Concern Birmingham (VSOP)
Aston University	Agilisys Services Ltd (OCOS/WO	Aspen Services Ltd (Gosford Pa
Black Country Museum Trust Limited (The)	BID	Black Country Consortium Limited
BME United Limited	Black Country Partnership NHS Foundation Trust	Bloomsbury Local Management Organisation Limited
Chuckery Tenant Management Organisation Limited	Brownhills Community Association Limited	Bushbury Hill Estate Management Board Limited

# Page 22

Admitted Bodies		
Coventry Heritage and Arts Trust	Coventry Solihull and Warwick	Coventry and Solihull Waste Disposal Company Limited (The)
Delves East Estate Management Limited	Coventry Law Centre Ltd	Coventry Sports Trust Limited
Family Care Trust	Dovecotes TMO	Edith Cadbury Nursery School
Home Start Northfield	Friendship Care and Housing Limited (formerly Beechdale)	Heart of England Care
Leamore Residents Association Limited	Home Start Stockland Green/Erdington	Home Start Walsall
Life Education Centres West Midlands	Leisure ^ Community Partnership Limited	Lieutenancy Services (West Midlands) Limited
Marketing Birmingham Limited	Light House Media Centre	Manor Farm Community Association
Murray Hall Community Trust Limited	Midland Heart Ltd	Millennium Point Trust
New Park Village Tenant Management Organisation	Museum of British Road Transport Trust (Coventry) Limited	Mytime Active
Palfrey Community Association	Northern Housing Consortium Limited	Optima Community Association
Priory Family Centre	Penderels Trust Limited (The)	Pool Hayes Community Association
Sandbank Tenant Management Organisation Limited	Riverside Housing Association Limited (formerly Riverside Group Limited)	Sandwell Arts Trust
Sickle Cell and Thalassaemia Support Project (Wolverhampton)	Sandwell Community Caring Trust (The)	Sandwell Inspired Partnership
St Columba's Day Care Centre	Solihull Care Limited	Solihull Care Trust
Voyage Care Limited	Steps to Work (Walsall) Ltd	Titan Partnership
WATMOS Community Homes	University of Warwick	Walsall Housing Group Limited
Wildside Activity Centre	West Midlands Transport Information Services Limited	Whitefriars Housing Group Limited
Wolverhampton Voluntary Sector Council	Wolverhampton Grammar School	Wolverhampton Network Consortium
Without Active Members		
Adoption Support (terminated 31 March 2011)		
Asian Welfare Centre	All Saints Haque Centre	Aquarius Action Projects
Bilston and Ettingshall SureStart	Asian Women's Adhikar Association (AWAAZ)	Belgrade Theatre Trust (Coventry) Limited

# age 224

Admitted Bodies		
Birmingham Heartlands Development Corporation	Birmingham and Solihull Connexions Services	Birmingham and Solihull Learning Exchange (The)
Black Country Museum Development Trust (The)	Black Business in Birmingham	Black Country Connexions
Cerebral Palsy Midlands	BXL (Terminated 10 February 2012)	Cannon Hill Trust (now Midlands Arts Council)
CV One Limited	Community Justice National Training Organisation	Coventry Voluntary Service Council
East Birmingham Family Service Unit	Druids Heath TMO	Dudley Zoo Development Trust
Metropolitan Authorities Recruitment Agency (METRA)	Heath Town Estate Management Board (terminated 4 August 2008)	Job Change Limited
Newman College (terminated 31 March 2008)	Moseley and District Churches Housing Association Limited	National Urban Forestry Unit
Sandwell Regeneration Company Limited. (terminated 31 March 2010)	Relate	RM Education
Springfield/Horseshoe Housing Management Co-operative Ltd	Solihull Community Caring Trust	South Birmingham Family Services Unit
Target Excel plc (Walsall MBC)	St Basil's Centre	Sunderland ARC Limited
University of Birmingham (Westhill)	The Chris Laws Day Care Centre for Older People (terminated 30 November 2010)	TSB Bank plc (formerly Birmingham Municipal Bank)
Wednesbury Action Zone	Walsall Enterprise Agency Limited	Walsall Regeneration Company Limited
West Midlands Examinations Board (The)	West Midlands Councils (formerly West Midlands Leaders Board)	West Midlands (West) Valuation Tribunal
Wolverhampton Development Corporation Limited	West Midlands Local Authorities Employers' Organisation	Wolverhampton Community Safety Partnership (terminated 30 April 2009)
	Wolverhampton Race Equality Council	

With Active Members	Action for Children (West Brom	Agilisys Limited
Action for Children (Smethwick	Alliance in Partnership - Ernesford Grange	Alliance in Partnership - President Kennedy
Alliance in Partnership - Camp Hill	Amey Highways Limited	Amey LG Limited
Alliance in Partnership - Stoke Park	Balfour Beatty Workplace Limited (Birmingham)	Balfour Beatty Workplace Limited (Coventry)
APCOA Parking (UK) Limited	Barnardos (Sandwell)	Bespoke Cleaning Services Limited
BAM Construct UK Limited	Capita IT Services Limited	Creative Support Limited
British Telecom plc	DRB Yew Tree Primary	Enterprise Managed Services Ltd - Solihull
DRB Wychall Primary	Galliford (UK) Limited	Harrison Catering Services Ltd
Enterprise Managed Services (W-ton)	Initial Catering Services Ltd (Rowley)	Initial Catering Services Ltd (Smethwick)
Housing 21	Interserve Facilities Manageme	Interserve Facilities Manageme
Integral UK Limited	KGB Cleaning & Support Services Limited	Lawrence Cleaning Limited
Interserve Facilities Manageme	Mears Group plc	Mears Limited
Leisure Living Limited	Mouchel Limited	New Heritage Regeneration Limited
Mitie PFI Limited	NSL Limited (Solihull)	Pell Frischmann Consultants Limited
NSL Limited	Quadron Services Limited	Redcliffe Catering Limited (Bordesley Green Girls School)
Premier Security Services Ltd	Regent Office Care Limited (Henley College)	Regent Office Care Limited (Hereward College)
Regent Office Care Limited (COWAT)	Serco Limited (Sandwell)	Service Birmingham Limited
Serco Limited	Tarmac Limited	Taylor Shaw (Colton Hills)
Sodexo Limited	Taylor Shaw (Hodge Hill)	Taylor Shaw Limited (COWAT)
Taylor Shaw (Great Barr School	Willmott Dixon Partnership Limited (North Contract)	Willmott Dixon Partnership Limited (South Contract)
Taylor Shaw Limited (St Albans)		
Without Active Members	Alliance in Partnership - Aston	APCOA Parking (UK) Ltd Solihull
Accord Operations (Birmingham)	Burrowes Street Tenant Management Organisations Limited	Central Parking Systems
Birmingham Accord Limited	JDM Accord Limited (Shrewsbury & Atcham)	JDM Accord Limited (Shropshire)

Page 225

Transferee Admission Bodies (Best Value)		
Interserve Project Services Limited (Smethwick Campus)	JDM Accord Limited (Telford & Wrekin)	Liberata UK Limited
JDM Accord Limited (Tamworth)	Mitie Cleaning (Midlands) Limited - Birmingham City Council	Mitie Managed Services (S&SW) Limited (terminated 31 December 2009)
Methodist Homes for the Aged (terminated 31 March 2010)	Mitie Cleaning (Midlands) Limited - Wednesfield	Mitie Property Services (UK) limited
Mitie Managed Services (S&SW) Limited - Coventry (terminated 31 December 2009)	Morrison Facilities Services Limited	MLA West Midlands (terminated 31 March 2009)
MLA West Midlands (terminated 31 March 2009)	Redcliffe Catering Limited (Aston School)	Redcliffe Catering Limited (Camp Hill School)
Morrison Facilities Services Limited	Regent Office Care Limited (City College)	Regent Office Care Limited (Whitefriars)
Regent Office Care Limited	Serco Limited (Stoke)	Strand Limited (terminated 31 July 2009)
Research Machines plc	Target Excel Plc (Magistrates Courts)	Target Excel Plc (Solihull MBC)
Superclean Services	Temple Security Limited	Veolia Environmental Serviced Cleanaway (UK) Limited
Technology Innovation Centre (terminated 31 March 2009)	Wates Construction Limited (Birmingham)	West Midlands E-Learning Company
Vertex Data Science Limited		
Other Major Employers who have Participated in the Fund	Department of Transport	Department of Health and Social Security
Birmingham International Airport Plc	West Midlands Magistrates Courts Committee	
Severn Trent Water Authority		

#### Scope of Responsibility

N

Wolverhampton City Council is responsible for ensuring that its business is conducted in accordance with the law and proper standards, and that public money is safeguarded and properly accounted for, and used economically, efficiently and effectively. The Council also has a duty under the Local Government Act 1999 to make arrangements to secure continuous improvement in the way in which its functions are exercised, having regard to a combination of economy, efficiency and effectiveness.

In discharging this overall responsibility, the Council is responsible for putting in place proper arrangements for the governance of its affairs, facilitating the effective exercise of its functions, which includes arrangements for the management of risk.

The Council has approved and adopted a Local Code of Corporate Governance, which is consistent with the principles of the CIPFA/SOLACE Framework Delivering Good Governance in Local Government. This code is incorporated within the Council's Constitution, which is available for review on the Council's website.

The Council is also responsible for the strategic management and administration of the **West Midlands Pension Fund** with the Council's Chief Executive, Monitoring Officer and Section 151 Officer holding specific responsibilities for supporting the members of the Pensions Committee in their role.

Wolverhampton Homes is the Council's Arm's Length (Housing) Management Organisation (ALMO) and is a company wholly owned by the Council. The control of the ALMO is through the Board which has representatives drawn from 1/3 Council, 1/3 tenants and 1/3 independent. There is a Management Agreement between the Council and Wolverhampton Homes which sets out the contractual and governance arrangements between the parties.

# F

#### **10. ANNUAL GOVERNANCE STATEMENT**

#### The Purpose of the Governance Framework

The governance framework comprises the systems and processes, and culture and values, by which the Council is directed and controlled and its activities through which it accounts to, engages with and leads the community. It enables the Council to monitor the achievements of its strategic objectives and to consider whether those objectives have led to the delivery of appropriate, cost-effective services.

Risk management and internal control are a significant part of the governance framework and are designed to manage risk to a reasonable level. They cannot eliminate all risk of failure to achieve policies, aims and objectives and can therefore only provide reasonable and not absolute assurance of effectiveness. The systems of risk management and internal control are based on an on-going process designed to identify and prioritise the risks to the achievement of the Council's policies, aims and objectives, to evaluate the likelihood of those risks being realised and the impact should they be realised, and to manage them efficiently, effectively and economically.

The governance framework has been in place at the Council for the year ended 31 March 2014 and up to the date of approval of the annual report and statement of accounts.

#### The Governance Framework and Review of Effectiveness throughout 2013/14

The Council has the following Corporate Plan aims and themes: Encouraging Enterprise and Business, Empowering People and Communities, Re-Invigorating the City and Confident, Capable Council, which are underpinned by the governance environment. This environment is consistent with the six core principles of the CIPFA/ SOLACE framework.

The key elements of the systems and processes that comprise the Council's governance framework, and where assurance against these is required, are described below.

	Core principles of the CIPFA/ SOLACE framework	Assurances required	Governance framework providing assurance	Review of Effectiveness	Issues identified
F t t f f a t t F f f a t t F f f t t t t t t t t t t t t t t t		Delivery and communication of an agreed corporate plan     Quality services are delivered efficiently and effectively     Clearly defined roles and functions     Management of risk     Effectiveness of internal controls     Compliance with laws, regulation, internal policies and procedures     Value for money and efficient management of resources     High standards of conduct and behaviour     Public accountability     Published information is accurate and reliable		Statement of Accounts 2013/14     External Audit Report to Those Charged with Governance (ISA 260) Report 2013/14     Annual Internal Audit Report 2013/14     Annual Audit Committee Report 2013/14     Local Government Ombudsman Report 2013/14     Scrutiny reviews     Annual Governance Statement – follow up of 2012/13 issues     Director of Public Health Annual Report 2013/14	Corporate Landlord / FutureSpace Savings Targets PSN Compliance Contract Management and Monitoring Procurement Health and Social Care Reforms FutureWorks Partnership Governance Information Governance Strategic Asset Management Schools Improvement
a	are subject to effective scrutiny and managing risk	Implementation of previous governance issues	<ul> <li>Complaints System</li> <li>Financial Procedures Rules</li> <li>Contracts Procedure Rules</li> <li>Committee Management Information Systems (now modern.gov)</li> </ul>		

# Developing the capacity and capability of members and officers to be effective Engaging with local people and other stakeholders to ensure robust public accountability

# Page 2:

#### **10. ANNUAL GOVERNANCE STATEMENT**

#### **West Midlands Pension Fund**

The West Midlands Pension Fund have completed their own "Assurance Framework – Supporting the Annual Governance Statement" which identified that there had been no adverse matters arising from the work behind their assurance framework.

#### **Wolverhampton Homes**

Wolverhampton Homes have included a Statement of Corporate Governance within the Company's Financial Statements for 2013/14. This states that the control framework has been reviewed by the Company's Audit Committee on behalf of the Board of Wolverhampton Homes and found to be effective. The review included an assurance statement from the Company's internal auditors.

In reviewing the Council's priorities and the implications for its governance arrangements, the council carries out an annual review of the elements that make up the governance framework to ensure it remains effective.

- The transition of Public Health Services to the council from 1 April 2013.
- The establishment of the Health and Well Being Board as a committee of the council which has responsibility for tackling local health inequalities.
- The implementation of and compliance with the Public Sector Internal Audit Standards from 1 April 2013.
- An updated Terms of Reference for the Audit Committee

The Council has responsibility for conducting, at least annually, a review of the effectiveness of its governance framework including the system of internal control. The review of effectiveness is informed by the work of the Members and senior officers within the Council who have responsibility for the development and maintenance of the governance framework, Internal Audit's annual report, the Scrutiny function and also by reports made by the Council's external auditors and other review agencies and inspectorates, as noted above.

Internal Audit has concluded that based on the work undertaken during the year of areas key risk, the implementation by management of the recommendations made and the assurance made available to the council by other providers as well as directly by Internal Audit, it can provide

# age 232

U

#### **10. ANNUAL GOVERNANCE STATEMENT**

reasonable assurance that the Council has adequate and effective governance, risk management and internal control processes. Key areas of concern have been included within the governance issues noted below.

There is a requirement to report in this Statement that the authority is not fully compliant with CIPFA's Statement on the Role of the Section 151 Officer in Local Government (2009) as the Section 151 Officer post is not at the same level in the Authority as members of the Corporate Management Team (known as the Strategic Executive Board) and they do not report directly to the Chief Executive. However, alternative arrangements are in place whereby the Section 151 Officer attends meetings of the Corporate Management Team and has direct access to the Chief Executive when required.

A number of issues were identified in the 2012/13 Annual Governance Statement and an update of the progress made in implementing the actions to improve these areas is included below. Where sufficient progress has not been made, the issues have been included in the 2014/15 issues.

#### Progress on the Governance Issues from 2012/13

The table below describes the governance issues identified during 2012/13 and the progress made against these during 2013/14.

#### 2012/13 - Key areas for Improvement

#### FutureSpaces: Corporate Landlord

The management of and responsibility for the Council's property assets is currently split between two directorates. Several initiatives and proposals for maintenance programmes and better targeted use of properties have been put forward. It is necessary that clarity of ownership and control of decision making is determined to ensure effective progress is made. Also work is on-going to improve the coordination of responsibilities as the Council develops the role of a 'Corporate Landlord' between the Directorates, along with the continued development of a 'One Council' approach to the use of land and assets and the development of options and a strategy to utilise available properties for community use that are not Council owned

# **Update on position and implication for the 2013/14 Annual Governance Statement**

#### The Corporate Landlord

The necessary planning for the introduction of the Corporate Landlord was completed to the set deadlines with the Corporate Landlord subsequently having been implemented incrementally. The key outputs being as follows:

- The approved governance arrangements are now fully operational with the Strategic and Operational Land and Property Boards meeting regularly with integrated work programmes, with shared programme management resources.
- The Strategic Asset Review is scheduled for completion in June 2014 as planned.
- There has been a lifting and shifting of property related service functions from the Community Directorate and the Education and

	property.	Enterprise Directorate to create the Corporate Landlord in the Delivery Directorate. This is being followed by a budget centralisation and review exercise and process re-design.  FutureSpaces  A delivery plan is being developed based on the Council supported by technical advisors developing the detailed design for the refurbishment of the Civic Centre for the open tendering of the refurbishment works through an OJEU Notice. This is essential to ensure a robust Final Business Case for the Cabinet to approve to give a final approval to the programme.  Carried forward to 2013/14
Page 233	Information Governance  Following critical in-year reviews by the Information Commissioners Office in August and December 2012, the Council is putting in place a robust framework and effective working practices, including:  • An established and operational Information Governance Board • Mapped out work programme and resources • A new Information Governance structure • Information Governance policies have been approved • Training programmes are underway	The Information Governance Board continues to meet, and is now supported by an operational group to progress key issues in relation to information governance.  The Council has now centralised all information governance resource into one team which has enabled the development of a single work programme, against which significant progress is being made. The work plan and maturity model were endorsed by the Cabinet in March 2014, and progress has also been scrutinised by the Council's Scrutiny Board.  The mandatory training module has been rolled out primarily by e-learning but supported by a number of training sessions for employees, and 100% compliance has been completed.  The council is both PSN and Public Health Information Governance Toolkit compliant, and is on track to submit the Social Care Toolkit.  Carried forward to 2013/14
	Partnership Governance	Work has begun on a number of the Council's partnerships including:
	Partnerships are increasingly common and increasingly important to the Council, in order to deliver the corporate plan and respond to the	Black Country working: A Black Country Joint Committee and Advisory Board has been established by Wolverhampton City Council and the three

Localism agenda. These partnerships take many forms. For example, formal arrangements such as strategic service delivery partnerships, statutory partnerships and looser, informal relationships with community groups or the 'third sector'. Although each of these partnerships is formed to generate beneficial outcomes they also carry different types of risks and governance can be problematic.

In addition, some of the Council's partnerships have been in place for a number of years and the 'health' and governance arrangements of these partnerships have not been systematically reviewed to ensure they continue to contribute effectively to the corporate priorities. Therefore, the Council is to adopt a revised systematic and consistent approach to identifying its significant partnerships. Once the significant partnerships have been identified, a systematic review of the governance arrangements and the 'health' of each partnership will be carried out to ensure they continue to contribute to the corporate priorities and provide value for money. The findings of the reviews and the risks associated with these partnerships will then be reported to officers and Councillors with portfolio responsibilities.

neighbouring local authorities and the Black Country LEP. This will provide strong joint governance for the Black Country City Deal and Black Country Growth Deal. A framework of how this inter-relates with the council governance has been produced and considered by the Strategic Executive Board.

**City Partnerships:** The Local Strategic Partnership has been replaced by a City Board that will drive forward the City Strategy (the Sustainable Communities strategy for the city). The representatives on the City Board are the Leader and the Chief Executive. The Board is in shadow form, until September 2014.

The governance framework will also continue to be checked and rolled out across the key partnerships.

#### Carried forward to 2013/14

#### **Contract Management and Monitoring**

The Council has historically had an inconsistent approach to its contract monitoring. New processes are being put in place to ensure that contracts can be monitored and reviewed on an on-going basis for value for money in the future.

A draft guidance document for contract management is in the early stages of discussion and it will be necessary to develop this approach widely across the Council, and agree how it can be applied. A simple set of procedures to support the approach will be added to contract procedure rules. It will then be necessary to identify contract managers and provide training. Timing for these activities is currently being worked up.

#### Carried forward to 2013/14

#### **Procurement**

The Interim Head of Procurement had raised concerns over past tendering processes and the failure to follow the Council's Contract The adoption of the Due North e-tendering system and the Agresso ordering system has provided much improved controls and management information and the continued refinement and use of these systems

Procedure Rules. Following an independent review, these concerns were supported by the findings of Internal Audit who identified a number of cases of inconsistencies and ambiguities at various stages of the procurement processes. The recommendations arising from the audit review were agreed and a range of improved working practices are being put in place.

enhances accountability and compliance across the Council.

The Procurement Board have agreed to consider amendments to the contract procedure rules to simplify and rationalise the working practices in procurement. Once the Board has considered the revisions and agreed any changes to be made within the constitution, a structured training programme can be developed.

Carried forward to 2013/14

#### **Savings Targets**

While the Council's current and historical savings targets have been largely delivered, there are still a limited number of such targets that have not yet been and also some, where proposals are yet to be developed. A failure to meet these targets will adversely impact upon the Council's ability to meet its objectives. Close monitoring of the situation continues at both senior officer and Councillor level.

All savings were fully reviewed and re-evaluated as part of the 2014/15 budget setting process that was completed in March 2014. This process identified that a number of savings that were reflected in the 2013/14 budget were to be achieved in an alternative way and not as originally planned. These changes have been reflected in the rebased 2014/15 budget and the original savings removed. In addition no significant adverse overall impact arose during 2013/14 as a result of the savings targets that were included in the budget.

# Resilience Function (Emergency Planning and Business Continuity)

The Council has identified issues in its ability to respond fully to its responsibilities under the Civil Contingencies Act. The Cabinet has approved the creation of a new Resilience Team in 2013/14 to bring together the separate Emergency Planning and Business Continuity functions. This is to underpin the delivery of the new Major Incident Plan and suite of subsidiary plans. The Resilience Team will operate within new governance arrangements and report to a Board. The Board will oversee the delivery of the adopted project plan for Resilience that will be reviewed monthly by the Strategic Executive Board. Annual audits will also be conducted to validate progress against the project plan.

The Resilience Board is now fully operational against the approved governance arrangements. The Board has developed a work programme for 2014/15 that integrates Council and Public Health roles and responsibilities. Good progress is being made against this work programme.

#### **Equalities**

The Council has identified issues in its ability to respond fully to its responsibilities in respect of equalities and consultation. An Equalities Project Board has been formed and approved an equalities work programme. This programme identifies ways of mainstreaming and promoting best equalities practice. Appropriate measures will be implemented during 2013/14.

To further support the roll out of the equalities training a number of employee briefing sessions have been delivered. The Equalities Member Champion continues to chair quarterly the Equalities Advisory Group. An internal officer equalities steering group has also been instigated and meets at least quarterly. Finally, a cumulative equality analysis of the budget was conducted, and lessons learnt from the process have been captured in order to continue to improve the equality analysis toolkit.

#### Action Plan for the Significant Governance Issues identified during 2013/14 which will need addressing in 2014/15

Based on the council's established risk management approach, the following issues have been assessed as being "significant" for the purpose of the 2013/14 annual governance statement. Over the coming year appropriate actions to address these matters and further enhance governance arrangements will be taken. These actions will address the need for improvements that were identified in the review of effectiveness and their implementation will be monitored as part of the next annual review and risk management arrangements in place.

	2013/14 - Key improvement areas and actions for implementation	Responsibility and expected implementation date
	FutureSpaces	Assistant Director, Delivery
Page 2	Delivery arrangements are being developed for the refurbishment of the Civic Centre. The intention is that construction works will begin in early 2015 subject to the final business case evidencing an on-going annual revenue saving of £500,000.	February 2015
SS	Corporate Landlord	
	The adoption of the Corporate Landlord Model is being progressed incrementally against the Strategic Asset Management Review. The adopted work programme for the implementation was approved by both the Strategic Land and Property Board and the Operational Land and Property Board and is now being actively monitored by both Boards. The key deliverables in the programme include:	
	<ul> <li>Centralisation of property related budgets.</li> <li>Creation and delivery of a funded programme of annual condition surveys and statutory testing.</li> <li>Creation and delivery of more robust cyclical maintenance programme based on the annual condition surveys.</li> <li>Development and delivery of a more robust Disposal Programme to achieve the capital receipts from disposals in the MTFS.</li> <li>Agreement of service profiles by building profile for services for Facilities Management to deliver to their clients.</li> <li>Development and delivery of the annual renewables programme.</li> <li>Support to service reviews being conducted with Directorates in respect to providing options appraisals in respect to meeting the property needs for new service operating models.</li> <li>Development of a revised Corporate Asset Management Plan.</li> </ul>	

Covings Towards		Assistant Diverton Finance
Savings Targets  Whilst the Council's current and historical savings targets have to financial environment continues to require substantial year on years and develop further savings will adversely impact upon monitoring of the situation continues at both senior officer and Council	ear savings. The failure to deliver already identified the Council's ability to meet its objectives. Close	Assistant Director, Finance March 2015
PSN Compliance		Head of Service – ICT
Prior to 2013 Wolverhampton City Council obtained GCSx accressinformation with central government and government agencies. during 2013 demanded improved technical security standards at Wolverhampton City Council's compliance with the PSN Code of Cabinet Office on 19 November 2013 following an independent vulnerabilities from external sources of attack and from within the Council's network and security architecture, ICT operational consultants at the Cabinet Office. PSN compliance remains at the of new services. A continual programme of infrastructure upgrade the Council undertaking the annual accreditation process during	The introduction of the Public Services Network (PSN) and more robust Information Governance requirements. If Connection requirements was approved by the health check of the Council's ICT estate, looking for the Council followed by a comprehensive evaluation of practices and information governance policies by CLAS are heart of all ICT decisions regarding the introduction des and refresh ensures compliance is maintained, with	March 2015
Contract Management and Monitoring		Strategic Director, Delivery
Having identified the range of contracts that are in place the mai managed and whether the contracts are meeting their original ex	xpectations. With the wide variety of contracts it will be	Assistant Director, Finance
necessary to develop several different approaches to contract m involvement for the contract managers (at tender stage), regular the options available and general awareness training for nomina	reporting on performance, planned reviews to assess	March 2015
Procurement		Strategic Director, Delivery
The Procurement Board will be instrumental in guiding the devel utilisation of e-procurement systems (Agresso and Due North) wo of this spend can be matched to the contract register. The impro	vill increase the overall visibility of spend and the profile	Assistant Director, Finance
particular categories of expenditure and develop procurement st market sounding, options appraisals and output based specifica	trategies that will extract value for money. The use of	March 2015

ge 2	and increasing the percentage of on contract spend. We will be using collaborative arrangements where these are beneficial to the Council and selecting the appropriate frameworks will be carried out in conjunction with the operational experts. The use of standardised forms and procedures will also aid compliance.  Health and Social Care Reforms  Over the next few years adult social care is required to take a lead role in implementing a service delivery transformation to effectively respond to a number of challenges and opportunities which arise due to significant budget reductions and changes to local and national policy. To be delivered successfully, the service transformation involves putting agreed strategies and plans in place. One of the strategies to be implemented is the Better Care Fund which is an integrated pooled budget to support health and social care to work together in local areas. To achieve the outcomes of the fund will require strong partnership working. Significant planning and investment of resources will also be required to set up primary care, prevention and community services in order to achieve the fund outcomes.  Another strategy the Council will be required to implement will be the Care and Support Bill in April 2015. Mapping, analysis and assessment of the detailed requirements of the Bill to identify the resources required to implement the changes, needs to be undertaken to fully assess the impact of the Bill on the Council. 'The Health and Wellbeing Board will be accountable for the authorisation and delivery of the fund. The Terms of Reference and governance of the Health and Wellbeing Board are being reviewed to take account of these new responsibilities'	Assistant Director, Health, Wellbeing and Disability  March 2015
	FutureWorks  The FutureWorks Programme delivered the new Agresso IT system on 1 April 2014 in line with the contractual timescales. This new system and processes are being used across the council and its partner organisations of Wolverhampton Homes and West Midlands Pension Scheme. This successful delivery has enabled the council to start delivering the agreed year one savings.  The Council is now embarking on delivering phase 2 of the programme in rolling our self-service functionality which will maximise the return on investment as council services are transformed. The Council will continue to manage the risks around the general governance and structure of this programme and through the general programme controls it will bring.	Strategic Director, Delivery Assistant Director, Finance March 2015
	Partnership Governance  While work has begun on a number of the Council's key partnerships, a systematic approach to identifying all of the significant partnerships and in determining the level of review of the governance arrangements alongside the 'health' of	Assistant Director, Partnerships, Economy and Culture supported by the

# Page 241

each partnership, is still being rolled out and will be quite a sizeable task.	Chief Legal Officer March 2015
Information Governance  The Council is building on the rebust framework and effective working practices it has put in place since concensual.	Head of Policy March 2015
The Council is building on the robust framework and effective working practices it has put in place since consensual audits from the Information Commissioner's Office in 2012 and an enforcement notice in 2014, including:	March 2015
<ul> <li>Supporting the Information Governance Board through the development of an operational group to drive progress</li> <li>Mapped out a centralised work programme and resources including a new structure</li> </ul>	
<ul> <li>Review of all Information Governance policies</li> <li>Roll out of a suite of training programmes, including ensuring there is 100% compliance with the mandatory 'Protecting Information' training to all employees</li> </ul>	
Strategic Asset Management	Strategic Director,
The Corporate Landlord model has now been formally adopted by the Council. The implementation of the model and developing a clear understanding of the accountability for activities and financial management will continue.  As part of the Corporate Landlord approach Strategic and Operational Land and Property management is now covered by two governance Boards. The Strategic Land and Property Board is chaired by the Strategic Director Education and Enterprise, and attended by the Corporate Landlord (Strategic Director Delivery). The Operational Land and Property Board is chaired by the Strategic Director Delivery.	Education and Enterprise and Assistant Director, Delivery March 2015
The Strategic Land and Property Board meets monthly and the Operational Land and Property Board meets fortnightly. Both Boards consider land and property matters and consult Members through the existing processes of Cabinet Member Briefing; Property Advisory Group and Executive Team prior to decisions being formally made in line with the Council's Constitution.	
The Strategic Asset Review has now concluded with the agreed categorisation of assets, and clear accountability established for assets including those considered to be of strategic importance (for Economic Growth and Regeneration including Housing).	
Strategic Asset Management will in the future be supported by the Corporate Landlord's Asset Management team acting as a single source of information and knowledge for the Council's assets making sure that the respective governance boards are provided with accurate and relevant information enabling informed decision making and	

U

#### **10. ANNUAL GOVERNANCE STATEMENT**

#### direction.

#### **Schools Improvement**

The Council's vision is to create an education system in Wolverhampton that promotes higher standards for all children and young people and closes the attainment gap. This is a system where the Council celebrates school autonomy and supports school leaders and teachers in leading school improvement and having the highest expectations of every child and young person.

Assistant Director, Education and Enterprise March 2015

The Council's strategic approach to school improvement therefore is to support effective school to school collaboration, strong partnerships and to develop excellent practice in schools whilst demonstrating clear leadership and challenge in the delivery of the its duties in relation to school improvement. This leads to three clear and related targets:

- To ensure that every child in the city has an excellent education.
- To raise standards in schools and academies so that by September 2016 attainment and progress measures at all key stages match or exceed national averages.

To improve the quality of provision in schools and academies so that by September 2016 all schools, settings and academies achieve an Ofsted inspection judgement of good or outstanding.

#### **Future Assurance**

Progress reports on the implementation of the above actions from these key improvement areas will be produced by Audit Services and reported to the Audit Committee during 2014/15.

#### Certification

To the best of our knowledge, the governance arrangements, as outlined above have been effectively operating during the year with the exception of those areas identified as requiring improvement. We propose over the coming year to take steps to address the above matters to further enhance our governance arrangements. We are satisfied that these steps will address the need for improvements that were identified during the review of effectiveness and will monitor their implementation and operation as part of our annual review.



Roger Lawrence, Leader of the Council:

Date:



Simon Warren, Chief Executive:

Date:

#### **Academy**

A school which chooses to opt out of a local authority's control and maintain its own funding.

#### **Accruals (Accrual Accounting)**

Refers to the fundamental accounting principle that income and expenditure are recognised as they are earned or incurred, not as money is received or paid.

See Receivables, Payables

#### **Actuarial / Actuary**

The science and profession of using mathematical techniques to model and quantify the financial effects of uncertain future events. For the council, this is relevant in the context of accounting for the Pension Fund, where future transactions of the fund will occur so far into the future that they cannot yet be known with certainty.

#### Arm's Length Management Organisation

An organisation which is, according to legislation, controlled by (i.e. a subsidiary of) a parent organisation, but whose management structures mean that control is loose and rarely manifests it directly on day-to-day operations of the subsidiary.

#### **Amortisation**

The way in which an asset or liability is accounted for over more than one period (other than property, plant and equipment, for which depreciation applies).

See Depreciation

#### Asset

An item that is owned by and can be used by the Council.

See Non-Current Asset

#### **Bad Debt Provision**

Bad debts are amounts owed to the council which it does not believe will be repaid. The council makes a provision for the amount of bad debt it expects to incur.

#### **Budget**

A budget is a plan of approved spending during a financial year.

#### **Business Rate or National Non-Domestic Rates (NNDR)**

Businesses across the country have to pay business rates. The government decides how much they should pay and Local Authorities collect the money. Local authorities pass the money to the Government who then share the total amount collected nationally between Authorities to help pay for local services.

#### **Capital Adjustment Account**

From 2007/08 onwards, an account whose purpose is to serve as a balancing mechanism between the different rates at which assets are depreciated in line with the SORP, and are financed under the capital controls regime. It is shown in the Balance Sheet as a reserve, although it does not represent funds available for future expenditure.

See Capital Financing Requirement

#### Capital Expenditure

Expenditure on the acquisition of property, plant and equipment, or expenditure which adds to, and not merely maintains, the value of an existing asset.

See Deferred Charge, Property, Plant and Equipment

Property Plant and Equipment

#### **Capital Financing Requirement**

An amount calculated as Non-Current Assets less the balances on the Capital Adjustment Account. See Minimum Revenue Provision

#### **Capital Programme**

The plan of approved spending on fixed assets (which includes assets that do not belong to the council, under certain circumstances.)

#### **Capital Receipt**

Money received from the disposal of land and other assets, and from the repayment of grants and loans made by the council.

#### Chartered Institute of Public Finance and Accountancy (CIPFA)

CIPFA is a UK accountancy body, specialising in the finances of the public sector. CIPFA is responsible for determining the accounting rules and procedures that apply to local authorities.

See Statement of Recommended Practice, Code of Practice

#### **Code of Practice on Local Authority Accounting**

The set of accounting principles and practices developed by the CIPFA/LASAAC Code Board under the oversight of the Financial Reporting Advisory Board. The Code is based on approved accounting standards issued by the International Accounting Standards Board and interpretations of the International Financial Reporting Interpretations Committee, except where these are inconsistent with specific statutory requirements. The Code also draws on approved accounting standards issued by the International Public Sector Accounting Standards Board and the UK Accounting Standards Board where these provide additional guidance.

See International Financial Reporting Standards, Chartered Institute of Public Finance and Accountancy (CIPFA)

#### Collection Fund

A fund administered by the council recording receipts from Council Tax and payments to the General Fund and other public authorities. It also records receipts of National Non Domestic Rates collected and payments to the General Fund and other public bodies.

#### **Community Assets**

Assets that the council intends to hold in perpetuity, that have no determinable useful life, and that may have restrictions on their disposal. Examples of community assets are parks and historic buildings.

#### Consistency

The concept that the accounting treatment of like items within an accounting period and from one period to the next is the same.

#### **Contingent Asset**

A contingent asset is a possible asset arising from past events whose existence will be confirmed only by the occurrence of one or more uncertain future events not wholly within the council's control.

#### **Contingent Liability**

A contingent liability is either:

- a) A possible obligation arising from past events whose existence will be confirmed only by the occurrence of one or more uncertain future events not wholly within the Council's control, or
- b) A present obligation arising from past events where it is not probable that a transfer of economic benefits will be required or the amount of the obligation cannot be measured with sufficient reliability.

#### **Corporate and Democratic Core**

The corporate and democratic core comprises all activities which local authorities engage in specifically because they are elected, multi-purpose authorities. The costs of these activities are over and above those which would be incurred by a series of independent, single purpose, nominated bodies managing the same services. There is therefore no logical basis for apportioning these costs to services.

# **U** Council Tax

A tax paid by residents of the authority to the council, based on the value of their property, to be spent on local services.

#### **Current Asset**

An asset held for a short period of time, for example cash in the bank, stocks and receivables.

#### **Dedicated Schools Grant**

Schools are funded separately from other council services. The council receive a Dedicated Schools Grant (DSG) direct from the Government, which is paid over to schools.

#### **Deficit**

This occurs when spending exceeds income (opposite of surplus).

#### **Defined Benefit Scheme**

A pension or other retirement benefit scheme other than a defined contribution scheme. Usually, the scheme rules define the benefits independently of the contributions payable, and the benefits are not directly related to the investments of the scheme. The scheme may be funded or unfunded (including notionally funded).

#### **Defined Contribution Scheme**

A pension or other retirement benefit scheme into which an employer pays regular contributions fixed as an amount or as a percentage of pay and will have no legal or constructive obligation to pay further contributions if the scheme does not have sufficient assets to pay all employee benefits relating to employee service in the current and prior periods.

#### **De Minimis**

The minimum value below which expenditure and income in respect of assets is not capitalised, but is charged or credited to revenue in full in the period it was incurred or earned.

See Capital Expenditure

#### **Depreciation**

The measure of the wearing out, consumption or other reduction in the useful economic life of property, plant and equipment. See *Impairment* 

#### **Disclosure**

Additional information required by the Code of Practice if a set of conditions are met. If the council judges that the conditions have not been met in its case, they will make no disclosure.

See Code of Practice

#### Discount

A reduction given by a lender in the amount to be repaid on early redemption of a loan. This is generally where the terms of the loan (relative to current market conditions) are favourable to the borrower.

See Premium

#### **Dividend**

A payment made by a company out of profits to its shareholders.

#### **Earmarked Reserve**

A sum set aside for a specific purpose.

See Usable and Unusable Reserves

#### **Events after the Reporting Period**

Those events, both favourable and unfavourable, that occur between the balance sheet date and the date on which the statement of accounts is signed by the responsible officer.

#### **Exceptional Items**

Material Items which derive from events or transactions that fall within the ordinary activities of the council and which need to be disclosed separately by virtue of their size or incidence to give fair presentation of the accounts.

#### **Existing Use Value (Social Housing)**

The value of a dwelling, given that, were it to be sold, the new purchaser must rent out the property, and set rents at social housing (i.e. below open market) levels.

See Vacant Possession Value

#### **Fair Value**

The fair value of an asset is the price at which it could be exchanged in an arm's length transaction less, where applicable, any grants receivable towards the purchase or use of the asset.

# $^{\sim}_{4}$ Fees and Charges

Income arising from the provision of services, for example the use of leisure facilities.

#### **Finance Lease**

A lease that transfers substantially all of the risks and rewards of ownership of property, plant and equipment to the lessee. The payments usually cover the full cost of the asset together with a return for the cost of finance.

See Operating Lease

#### **Financial Instrument**

Any contract that gives to a financial asset of one entity and a financial liability or equity instrument of another entity.

# **Financial Year**

This runs from 1 April to 31 March.

#### **General Fund**

The fund to which the cost of all services of the council (except for Housing Revenue Account services) is charged. The net cost of the General Fund is met by Council Tax, Governments Grants and NNDR.

#### **Going Concern**

The concept that the local authority will remain in operational existence for the foreseeable future, in particular that the revenue accounts and balance sheet assume no intention to curtail significantly the scale of operations.

#### **Government Grants**

Assistance by government and inter-governmental agencies and similar bodies, whether local, national or international, in the form of cash or transfers of assets to an authority, in return for past or future compliance with certain conditions relating to the activities of the council.

# T Heritage Assets

Assets that the council intends to hold for the purpose of informing or educating the public about their heritage, and which are not held for their investment value. Examples include collections of antiques in museums.

# **Housing Revenue Account (HRA)**

A ring-fenced account detailing the expenditure and income arising from the provision of council housing, as required by the Local Government and Housing Act 1989.

#### **Impairment**

N

A diminution in value of a property, plant and equipment resulting from amongst other things, obsolescence or physical damage. To comply with accounting standards the Council undertakes annual reviews of its assets to identify any assets which have been impaired.

See Property, Plant and Equipment

#### **Income and Expenditure Account / Statement**

This describes the expenditure made in a single year by an entity, in accordance with the accounting standards that apply at that time to that body in order to generate a view of its year end position in relation to its profit or usable reserves. The following terms are synonymous: "The Income and Expenditure Account", "Comprehensive Income and Expenditure Statement", "Income and Expenditure Statement".

#### **Infrastructure Assets**

These are inalienable assets, expenditure on which is recoverable only by continued use of the asset created. Examples of such assets are highways and footpaths.

#### **Intangible Assets**

An item which does not have physical substance (for example, software licenses) but can be identified and used by the council over a number of years.

#### **International Accounting Standards (IAS)**

These standards were issued by the International Accounting Standards Committee (IASC) - founded in 1973 as a private enterprise initiated by national accounting companies. This committee issued International Accounting Standards for private companies to follow. These standards have now largely been replaced by International Financial Reporting Standards.

See International Financial Reporting Standards

#### **International Financial Reporting Standards (IFRS)**

These standards are issued by the International Accounting Standards Board (IASB), established on 1 April 2001 with EU support to be the successor to the IASC. The IASB adopted the International Accounting Standards and then began issuing its own International Financial Reporting Standards.

These became mandatory for all private companies quoted on the Stock Exchange in 2004.

#### **Inventories**

Goods owned by the council which have not been used by the end of the financial year.

#### **Investments**

A long-term investment is an investment that is intended to be held for use on a continuing basis in the activities of the council. Investments should be so classified only where an intention to hold the investment for the long term can clearly be demonstrated or where there are restrictions as to the investor's ability to dispose of the investment.

# **Investment Properties**

Interest in land and/or buildings:

- (i) in respect of which construction work and development have been completed.
- (ii) is held for its investment potential, any rental income being negotiated at arm's length.

#### Levy

A payment made by the council to another local service, for example, local transport and the Environment Agency.

#### Liabilities

Amounts due to individuals or organisations which will have to be paid at some time in the future. Current liabilities are usually payable within one year of the balance sheet date.

See Accruals, Payables

## **Major Repairs Reserve**

A reserve to pay for large scale repairs to council houses.

#### **Materiality**

An item is material if its omission, non-disclosure or misstatement in financial statements could be expected to lead to a distortion of the view given by the financial statements.

#### N Provision for the Redemption of Debt (MRP)

A minimum amount, determined according to a formula approved by the council, which must be charged to the revenue account, for debt redemption or for the discharge of other credit liabilities.

See Capital Financing Requirement

#### **National Non-Domestic Rates (NNDR)**

Rates which are levied on business properties and collected by the council and accounted for on an agency basis. These funds are then distributed between the General Fund and other public bodies.

#### **Net Book Value**

The amount at which property plant and equipment are included in the balance sheet, i.e. their historical cost or current value less the cumulative amounts provided for depreciation.

See Property Plant and Equipment

#### **Net Current Replacement Cost**

The cost of replacing or recreating the particular asset in its existing condition and in its existing use, i.e. the cost of its replacement or of the nearest equivalent asset, adjusted to reflect the current condition of the existing asset.

#### **Net Realisable Value**

The open market value of the asset in its existing use (or market value in the case of non-operational assets), less the expenses to be incurred in realising the asset.

#### **Net Worth**

A monetary value, defined as the value of the council's assets less the value of its liabilities. This is the "bottom line" of the Balance Sheet.

#### **Non-Current Asset**

An item, for example land, buildings and vehicles, which yield benefits to the council and the services it provides over a period of more than one year.

# Obsolescence

The term used to describe an asset which no longer has any value to an organisation due to changes in the organisation's operating environment or the emergence of overwhelmingly superior alternatives to that asset.

See Impairment

#### **Operational & Non-Operational Assets**

Operational Assets are those that are used directly in providing council services. Non-operational assets are assets held for any other purpose, for example for investment or where they are no longer used and have been earmarked for disposal.

See Property Plant and Equipment

## **Operating Leases**

Leases other than a finance lease.

See Finance Leases

#### **Payables**

An amount owed by the council for work done, goods received or services rendered, but for which payment has not been made at the end of the year. See Accruals, Receivables

#### **Precept**

The amount levied by the various joint authorities (e.g. police and fire authorities) which is collected by the council on their behalf. A body which can set a precept is called a preceptor.

#### **Premium**

An amount charged by a lender (over and above the outstanding principal) on early redemption of a loan. This is generally where the terms of the loan (relative to current market conditions) are favourable to the lender.

See Discount

#### **Prior Year Adjustments**

Those material adjustments applicable to prior years, arising from changes in accounting policies or from the correction of fundamental errors. They do not include normal recurring corrections or adjustments of accounting estimates made in prior years.

# **Private Finance Initiative (PFI)**

A government initiative which enables authorities to carry out capital projects through partnership with the private sector.

# Property, Plant and Equipment

Tangible assets that yield benefits to the council and the services it provides for a period of more than one year. Examples include land, buildings and vehicles.

See Capital Expenditure

#### **Provisions**

N

Amounts set aside in respect of a liability of uncertain timing or amount, where a reliable estimate of the potential value can be made.

#### **Prudence**

This accounting concept requires that revenue is not anticipated until realisation can be assessed with reasonable certainty. Provision is made for all known liabilities whether the amount is certain or can only be estimated in light of the information available.

#### **Receipts in Advance**

Money received before the end of the financial year, but which relates to the following financial year.

#### Receivables

Sums of money owed to the council but not received at the end of the year.

See Accruals, Payables

#### **Related Party**

There is a detailed definition of related parties in FRS 8. For the council's purposes, related parties are deemed to include:

- (i) the elected members of the council and their partners.
- (ii) the chief officers of the council.
- (iii) the companies in which the council has an interest.
- (iv) central government and preceptors of Wolverhampton's Collection Fund.
- (v) other entities which the council has the ability to control or influence.

#### **Retirement Benefits**

All forms of consideration given by an employer in exchange for services rendered by employees that are payable after the completion of employment.

Retirement benefits do not include termination benefits payable as a result of either;

- (i) an employer's decision to terminate an employee's employment before the normal retirement date; or
- (ii) an employee's decision to accept voluntary redundancy in exchange for those benefits, because these are not given in exchange for services rendered by employees.

# **Revenue Expenditure**

Expenditure on the day-to-day running costs of services e.g. employees, premises, supplies and services.

# **Revenue Expenditure Funded From Capital Under Statute**

Spending on assets that have a lasting value but are not owned by the council, for example, improvement grants.

#### **Revenue Support Grant (RSG)**

Grant from central government towards the cost of providing General Fund services.

#### Ring fenced

Certain accounts, such as the Collection Fund, must be maintained separately outside the General Fund as a statutory requirement.

#### **Service Reporting Code of Practice (SERCOP)**

This guidance is issued by CIPFA and determines the costs which should be shown in the service lines in the Consolidated Income and Expenditure Statement, by determining which types of cost and income should be shown against which service. This promotes comparison between authorities by readers of the accounts.

See Income and Expenditure Account/Statement

#### **Trust Fund**

A fund administered by the council on behalf of others for such purposes as charities and specific projects.

#### **Usable Reserves**

Reserves that can be applied to fund expenditure or reduce local taxation.

# **→** Unusable Reserves

Amounts that have come about purely from accounting adjustments and are not therefore available to spend.

# N Useful life

The period over which the council will derive benefits from the use of an asset.

#### **Vacant Possession Value**

The market value of a property, were it to be sold with no unusual restrictions on the occupation of the property, or the level of any rents or charges made for its use.

See Existing Use Value (Social Housing)

# **Work In Progress**

Expenditure in respect of assets that are not yet ready to be put into use or sold (as appropriate).

Agenda Item No: 9



# **Audit Committee**

14 July 2014

Report title Annual Governance Statement – 2013/14

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s)

Peter Farrow Head of Audit

Tel 01902 554460

Email <u>peter.farrow@wolverhampton.gov.uk</u>

Report to be/has been

considered by

Not applicable

## Recommendation(s) for action or decision:

The Committee is recommended to:

Review and comment upon the contents of the Council's Annual Governance Statement for 2013/14.

# This report is PUBLIC [NOT PROTECTIVELY MARKED]

# 1.0 Purpose

- 1.1 That Members review and comment upon the content of the Annual Governance Statement for the year 2013/2014.
- 1.2 The Council is required under Regulation 4(2) of the Accounts and Audit Regulations 2003, as amended by the Accounts and Audit (Amendment) (England) Regulation 2006 to produce an Annual Governance Statement to be included in the annual statement of accounts, which is signed by the Leader of the Council and the Chief Executive.

# 2.0 Background

2.1 The Annual Governance Statement draws upon the management and internal control framework of the Council, especially the work of internal and external audit and the Council's risk management arrangements. In compiling the Annual Governance Statement assurance is obtained from a range of sources in order that the signatories to the Statement can assure themselves that the statement reflects the governance arrangements for which they are responsible.

## 3.0 Progress, options, discussion

3.1 Progress reports on the implementation of the actions required in the key improvement areas will be produced by Audit Services and reported to the Audit Committee during the year.

#### 4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/27062014/V)

#### 5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report. [AS/07072014/D]

#### 6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

#### 7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

# This report is PUBLIC [NOT PROTECTIVELY MARKED]

# 8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

# 9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations in this report

# 10.0 Schedule of background papers

10.1 Annual Governance Statement - 2013/14





# **Annual Governance Statement 2013/14**

# Scope of Responsibility

N

Wolverhampton City Council is responsible for ensuring that its business is conducted in accordance with the law and proper standards, and that public money is safeguarded and properly accounted for, and used economically, efficiently and effectively. The Council also has a duty under the Local Government Act 1999 to make arrangements to secure continuous improvement in the way in which its functions are exercised, having regard to a combination of economy, efficiency and effectiveness.

Indischarging this overall responsibility, the Council is responsible for putting in place proper arrangements for the governance of its affairs, fathlitating the effective exercise of its functions, which includes arrangements for the management of risk.

The Council has approved and adopted a Local Code of Corporate Governance, which is consistent with the principles of the CIPFA/SOLACE Framework Delivering Good Governance in Local Government. This code is incorporated within the Council's Constitution, which is available for review on the Council's website.

The Council is also responsible for the strategic management and administration of the **West Midlands Pension Fund** with the Council's Chief Executive, Monitoring Officer and Section 151 Officer holding specific responsibilities for supporting the members of the Pensions Committee in their role.

**Wolverhampton Homes** is the Council's Arm's Length (Housing) Management Organisation (ALMO) and is a company wholly owned by the Council. The control of the ALMO is through the Board which has representatives drawn from 1/3 Council, 1/3 tenants and 1/3 independent. There is a Management Agreement between the Council and Wolverhampton Homes which sets out the contractual and governance arrangements between the parties.

# The Purpose of the Governance Framework

The governance framework comprises the systems and processes, and culture and values, by which the Council is directed and controlled and its activities through which it accounts to, engages with and leads the community. It enables the Council to monitor the achievements of its strategic objectives and to consider whether those objectives have led to the delivery of appropriate, cost-effective services.

Risk management and internal control are a significant part of the governance framework and are designed to manage risk to a reasonable level. They cannot eliminate all risk of failure to achieve policies, aims and objectives and can therefore only provide reasonable and not absolute assurance of effectiveness. The systems of risk management and internal control are based on an on-going process designed to identify and prioritise the risks to the achievement of the Council's policies, aims and objectives, to evaluate the likelihood of those risks being realised and the impact should they be realised, and to manage them efficiently, effectively and economically.

The governance framework has been in place at the Council for the year ended 31 March 2014 and up to the date of approval of the annual report at statement of accounts.

# The Governance Framework and Review of Effectiveness throughout 2013/14

The Council has the following Corporate Plan aims and themes: Encouraging Enterprise and Business, Empowering People and Communities, Re-Invigorating the City and Confident, Capable Council, which are underpinned by the governance environment. This environment is consistent with the six core principles of the CIPFA/ SOLACE framework.

The key elements of the systems and processes that comprise the Council's governance framework, and where assurance against these is required, are described below.

Core principles of the CIPFA/ SOLACE framework	Assurances required	Governance framework providing assurance	Review of Effectiveness	Issues identified
Focusing on the purpose of the authority and on outcomes for the community and creating and implementing a vision for the local area  Members and officers working together to achieve a common purpose with clearly defined functions and roles  Promoting values for the authority and demonstrating the values of good governance through upholding high standards of conduct and behaviour  Taking informed and transparent decisions which are subject to effective scrutiny and managing risk  Developing the capacity and capability of members and officers to be effective	Delivery and communication of an agreed corporate plan     Quality services are delivered efficiently and effectively     Clearly defined roles and functions     Management of risk     Effectiveness of internal controls     Compliance with laws, regulation, internal policies and procedures     Value for money and efficient management of resources     High standards of conduct and behaviour     Public accountability     Published information is accurate and reliable     Implementation of	The Constitution (including Head of Paid Service, Chief Financial Officer and Monitoring Officer) Council, Cabinet and Committees Scrutiny function Audit Committee (and Sub-Committee) Standards Committee Internal and External Audit Strategic Executive Board Corporate Development Board Directors Assurance Statements Corporate and business plans Medium Term Financial Strategy Corporate Risk Register Codes of Conduct Business Planning and Performance Management Framework Whistleblowing and other anti-fraud related policies Complaints System	Statement of Accounts 2013/14     External Audit Report to Those Charged with Governance (ISA 260) Report 2013/14     Annual Internal Audit Report 2013/14     Annual Audit Committee Report 2013/14     Local Government Ombudsman Report 2013/14     Scrutiny reviews     Annual Governance Statement – follow up of 2012/13 issues     Director of Public Health Annual Report 2013/14	Corporate Landlord / FutureSpace Savings Targets PSN Compliance Contract Management and Monitoring Procurement Health and Social Care Reforms FutureWorks Partnership Governance Information Governance Strategic Asset Management Schools Improvement
Engaging with local people and other stakeholders to ensure robust public accountability	previous governance issues	<ul> <li>Financial Procedures Rules</li> <li>Contracts Procedure Rules</li> <li>Committee Management Information Systems (now modern.gov)</li> </ul>		

#### West Midlands Pension Fund

The West Midlands Pension Fund have completed their own "Assurance Framework – Supporting the Annual Governance Statement" which identified that there had been no adverse matters arising from the work behind their assurance framework.

## **Wolverhampton Homes**

Wolverhampton Homes have included a Statement of Corporate Governance within the Company's Financial Statements for 2013/14. This states that the control framework has been reviewed by the Company's Audit Committee on behalf of the Board of Wolverhampton Homes and found to be effective. The review included an assurance statement from the Company's internal auditors.

In reviewing the Council's priorities and the implications for its governance arrangements, the council carries out an annual review of the elements that make up the governance framework to ensure it remains effective.

Pa

The key changes to the governance framework during the year include:

- No The transition of Public Health Services to the council from 1 April 2013.
- The establishment of the Health and Well Being Board as a committee of the council which has responsibility for tackling local health inequalities.
- The implementation of and compliance with the Public Sector Internal Audit Standards from 1 April 2013.
- An updated Terms of Reference for the Audit Committee

The Council has responsibility for conducting, at least annually, a review of the effectiveness of its governance framework including the system of internal control. The review of effectiveness is informed by the work of the Members and senior officers within the Council who have responsibility for the development and maintenance of the governance framework, Internal Audit's annual report, the Scrutiny function and also by reports made by the Council's external auditors and other review agencies and inspectorates, as noted above.

Internal Audit has concluded that based on the work undertaken during the year of areas key risk, the implementation by management of the recommendations made and the assurance made available to the council by other providers as well as directly by Internal Audit, it can provide reasonable assurance that the Council has adequate and effective governance, risk management and internal control processes". Key areas of concern have been included within the governance issues noted below.

There is a requirement to report in this Statement that the authority is not fully compliant with CIPFA's Statement on the Role of the Section 151 Officer in Local Government (2009) as the Section 151 Officer post is not at the same level in the Authority as members of the Corporate Management Team (known as the Strategic Executive Board) and they do not report directly to the Chief Executive. However, alternative arrangements are in place whereby the Section 151 Officer attends meetings of the Corporate Management Team and has direct access to the Chief Executive when required.

A number of issues were identified in the 2012/3 Annual Governance Statement and an update of the progress made in implementing the actions to improve these areas is included below. Where sufficient progress has not been made, the issues have been included in the 2014/15 issues.

# Progress on the Governance Issues from 2012/13

The table below describes the governance issues identified during 2012/13 and the progress made against these during 2013/14.

# 2012/13 - Key areas for Improvement

# **FutureSpaces:** Corporate Landlord

assets is currently split between two directorates. Several initiatives and proposals for maintenance programmes and better targeted use of properties have been put forward. It is necessary that clarity of ownership and control of decision making is determined to ensure effective progress is made. Also work is on-going to improve the coordination of responsibilities as the Council develops the role of a 'Corporate Landlord' between the Directorates, along with the continued development of a 'One Council' approach to the use of land and assets and the development of options and a strategy to utilise available properties for community use that are not Council owned property.

# Update on position and implication for the 2013/14 Annual Governance Statement

#### The Corporate Landlord

The necessary planning for the introduction of the Corporate Landlord was completed to the set deadlines with the Corporate Landlord subsequently having been implemented incrementally. The key outputs being as follows:

- The approved governance arrangements are now fully operational with the Strategic and Operational Land and Property Boards meeting regularly with integrated work programmes, with shared programme management resources.
- The Strategic Asset Review is scheduled for completion in June 2014 as planned.
- There has been a lifting and shifting of property related service functions from the Community Directorate and the Education and Enterprise Directorate to create the Corporate Landlord in the Delivery Directorate. This is being followed by a budget centralisation and review exercise and process re-design.

## **FutureSpaces**

A delivery plan is being developed based on the Council supported by technical advisors developing the detailed design for the refurbishment of the Civic Centre for the open tendering of the refurbishment works through an OJEU Notice. This is essential to ensure a robust Final Business Case for the Cabinet to approve to give a final approval to the programme.

Carried forward to 2013/14

#### **Information Governance**

Following critical in-year reviews by the Information Commissioners Office in August and December 2012, the Council is putting in place a roust framework and effective working practices, including:

- An established and operational Information Governance Board
- Mapped out work programme and resources
- A new Information Governance structure
- Information Governance policies have been approved
- Training programmes are underway

The Information Governance Board continues to meet, and is now supported by an operational group to progress key issues in relation to information governance.

The Council has now centralised all information governance resource into one team which has enabled the development of a single work programme, against which significant progress is being made. The work plan and maturity model were endorsed by the Cabinet in March 2014, and progress has also been scrutinised by the Council's Scrutiny Board.

The mandatory training module has been rolled out primarily by elearning but supported by a number of training sessions for employees, and 100% compliance has been completed.

The council is both PSN and Public Health Information Governance Toolkit compliant, and is on track to submit the Social Care Toolkit.

Carried forward to 2013/14

# Partnership Governance

Partnerships are increasingly common and increasingly important to the Council, in order to deliver the corporate plan and respond to the Localism agenda. These partnerships take many forms. For example, formal arrangements such as strategic service delivery partnerships, Work has begun on a number of the Council's partnerships including:

**Black Country working:** A Black Country Joint Committee and Advisory Board has been established by Wolverhampton City Council and the three neighbouring local authorities and the Black Country LEP.

statutory partnerships and looser, informal relationships with community groups or the 'third sector'. Although each of these partnerships is formed to generate beneficial outcomes they also carry different types of risks and governance can be problematic. In addition, some of the Council's partnerships have been in place for a number of years and the 'health' and governance arrangements of these partnerships have not been systematically reviewed to ensure they continue to contribute effectively to the corporate priorities. Therefore, the Council is to adopt a revised systematic and consistent approach to identifying its significant partnerships. Once the significant partnerships have been identified, a systematic review of the governance arrangements and the 'health' of each partnership will be carried out to ensure they continue to contribute to the corporate priorities and provide value for money. The findings of the reviews and the risks associated with these partnerships will then be reported Cofficers and Councillors with portfolio responsibilities.

This will provide strong joint governance for the Black Country City Deal and Black Country Growth Deal. A framework of how this inter-relates with the council governance has been produced and considered by the Strategic Executive Board.

**City Partnerships:** The Local Strategic Partnership has been replaced by a City Board that will drive forward the City Strategy (the Sustainable Communities strategy for the city). The representatives on the City Board are the Leader and the Chief Executive. The Board is in shadow form, until September 2014.

The governance framework will also continue to be checked and rolled out across the key partnerships.

#### Carried forward to 2013/14

# Gentract Management and Monitoring

The Council has historically had an inconsistent approach to its contract monitoring. New processes are being put in place to ensure that contracts can be monitored and reviewed on an on-going basis for value for money in the future.

A draft guidance document for contract management is in the early stages of discussion and it will be necessary to develop this approach widely across the Council, and agree how it can be applied. A simple set of procedures to support the approach will be added to contract procedure rules. It will then be necessary to identify contract managers and provide training. Timing for these activities is currently being worked up.

#### Carried forward to 2013/14

#### **Procurement**

N

The Interim Head of Procurement had raised concerns over past tendering processes and the failure to follow the Council's Contract The adoption of the Due North e-tendering system and the Agresso ordering system has provided much improved controls and management information and the continued refinement and use of these systems

Procedure Rules. Following an independent review, these concerns were supported by the findings of Internal Audit who identified a number of cases of inconsistencies and ambiguities at various stages of the procurement processes. The recommendations arising from the audit review were agreed and a range of improved working practices are being put in place.

enhances accountability and compliance across the Council.

The Procurement Board have agreed to consider amendments to the contract procedure rules to simplify and rationalise the working practices in procurement. Once the Board has considered the revisions and agreed any changes to be made within the constitution, a structured training programme can be developed.

#### Carried forward to 2013/14

# **Savings Targets**

While the Council's current and historical savings targets have been largely delivered, there are still a limited number of such targets that have not yet been and also some, where proposals are yet to be developed. A failure to meet these targets will adversely impact upon the Council's ability to meet its objectives. Close monitoring of the situation continues at both senior officer and Councillor level.

All savings were fully reviewed and re-evaluated as part of the 2014/15 budget setting process that was completed in March 2014. This process identified that a number of savings that were reflected in the 2013/14 budget were to be achieved in an alternative way and not as originally planned. These changes have been reflected in the rebased 2014/15 budget and the original savings removed. In addition no significant adverse overall impact arose during 2013/14 as a result of the savings targets that were included in the budget.

# Resilience Function (Emergency Planning and Business Continuity)

The Council has identified issues in its ability to respond fully to its responsibilities under the Civil Contingencies Act. The Cabinet has approved the creation of a new Resilience Team in 2013/14 to bring together the separate Emergency Planning and Business Continuity functions. This is to underpin the delivery of the new Major Incident Plan and suite of subsidiary plans. The Resilience Team will operate within new governance arrangements and report to a Board. The Board will oversee the delivery of the adopted project plan for Resilience that will be reviewed monthly by the Strategic Executive Board. Annual audits will also be conducted to validate progress against the project plan.

The Resilience Board is now fully operational against the approved governance arrangements. The Board has developed a work programme for 2014/15 that integrates Council and Public Health roles and responsibilities. Good progress is being made against this work programme.

# **Equalities**

The Council has identified issues in its ability to respond fully to its responsibilities in respect of equalities and consultation. An Equalities Project Board has been formed and approved an equalities work programme. This programme identifies ways of mainstreaming and promoting best equalities practice. Appropriate measures will be implemented during 2103/14.

To further support the roll out of the equalities training a number of employee briefing sessions have been delivered. The Equalities Member Champion continues to chair quarterly the Equalities Advisory Group. An internal officer equalities steering group has also been instigated and meets at least quarterly. Finally, a cumulative equality analysis of the budget was conducted, and lessons learnt from the process have been captured in order to continue to improve the equality analysis toolkit.

# Action Plan for the Significant Governance Issues identified during 2013/14 which will need addressing in 2014/15

Based on the council's established risk management approach, the following issues have been assessed as being "significant" for the purpose of the 2013/14 annual governance statement. Over the coming year appropriate actions to address these matters and further enhance governance arrangements will be taken. These actions will address the need for improvements that were identified in the review of effectiveness and their implementation will be monitored as part of the next annual review and risk management arrangements in place

2013/14 - Key improvement areas and actions for implementation	Responsibility and expected implementation date
FutureSpaces	Assistant Director, Delivery
Dervery arrangements are being developed for the refurbishment of the Civic Centre. The intention is that construction works will begin in early 2015 subject to the final business case evidencing an on-going annual revenue saving of £500,000.	February 2015
The adoption of the Corporate Landlord Model is being progressed incrementally against the Strategic Asset Management Review. The adopted work programme for the implementation was approved by both the Strategic Land and Property Board and the Operational Land and Property Board and is now being actively monitored by both Boards. The key deliverables in the programme include:	
Centralisation of property related budgets.	
<ul> <li>Creation and delivery of a funded programme of annual condition surveys and statutory testing.</li> </ul>	
Creation and delivery of more robust cyclical maintenance programme based on the annual condition surveys.	
<ul> <li>Development and delivery of a more robust Disposal Programme to achieve the capital receipts from disposals in the MTFS.</li> </ul>	
Agreement of service profiles by building profile for services for Facilities Management to deliver to their clients.	
Development and delivery of the annual renewables programme.	
Support to service reviews being conducted with Directorates in respect to providing options appraisals in	

respect to meeting the property needs for new service operating models.	
Development of a revised Corporate Asset Management Plan.	
Savings Targets  Whilst the Council's current and historical savings targets have been largely delivered the extremely challenging financial environment continues to require substantial year on year savings. The failure to deliver already identified savings and develop further savings will adversely impact upon the Council's ability to meet its objectives. Close monitoring of the situation continues at both senior officer and Councillor level.	Assistant Director, Finance March 2015
PSN Compliance Prior to 2013 Wolverhampton City Council obtained GCSx accreditation enabling secure access to and exchange of information with central government and government agencies. The introduction of the Public Services Network (PSN) during 2013 demanded improved technical security standards and more robust Information Governance requirements. Wolverhampton City Council's compliance with the PSN Code of Connection requirements was approved by the Cabinet Office on 19 November 2013 following an independent health check of the Council's ICT exacts, looking for vulnerabilities from external sources of attack and from within the Council followed by a comprehensive evaluation of the Council's network and security architecture, ICT operational practices and integration governance policies by CLAS consultants at the Cabinet Office. PSN compliance remains at the heart of all ICT decisions regarding the introduction of new services. A continual programme of infrastructure upgrades and refresh ensures compliance is maintained, with the Council undertaking the annual accreditation process during May and June 2014	Head of Service – ICT March 2015
Contract Management and Monitoring  Having identified the range of contracts that are in place the main task is to establish how the reporting process is being managed and whether the contracts are meeting their original expectations. With the wide variety of contracts it will be necessary to develop several different approaches to contract management however the main principles will be early involvement for the contract managers (at tender stage), regular reporting on performance, planned reviews to assess the options available and general awareness training for nominated contract managers.	Strategic Director, Delivery Assistant Director, Finance March 2015
Procurement The Procurement Board will be instrumental in guiding the development of strategic procurement. The introduction and utilisation of e-procurement systems (Agresso and Due North) will increase the overall visibility of spend and the	Strategic Director, Delivery Assistant Director, Finance March 2015

profile of this spend can be matched to the contract register. The improved management information will be useful to target particular categories of expenditure and develop procurement strategies that will extract value for money. The use of market sounding, options appraisals and output based specifications will also contribute to improving budgetary controls and increasing the percentage of on contract spend. We will be using collaborative arrangements where these are beneficial to the Council and selecting the appropriate frameworks will be carried out in conjunction with the operational experts. The use of standardised forms and procedures will also aid compliance.

#### **Health and Social Care Reforms**

Over the next few years adult social care is required to take a lead role in implementing a service delivery transformation to effectively respond to a number of challenges and opportunities which arise due to significant budget reductions and changes to local and national policy. To be delivered successfully, the service transformation involves putting agreed strategies and plans in place. One of the strategies to be implemented is the Better Care Find which is an integrated pooled budget to support health and social care to work together in local areas. To addieve the outcomes of the fund will require strong partnership working. Significant planning and investment of resources will also be required to set up primary care, prevention and community services in order to achieve the fund outcomes.

Another strategy the Council will be required to implement will be the Care and Support Bill in April 2015. Mapping, analysis and assessment of the detailed requirements of the Bill to identify the resources required to implement the changes, needs to be undertaken to fully assess the impact of the Bill on the Council. 'The Health and Wellbeing Board will be accountable for the authorisation and delivery of the fund. The Terms of Reference and governance of the Health and Wellbeing Board are being reviewed to take account of these new responsibilities'

#### **FutureWorks**

The FutureWorks Programme delivered the new Agresso IT system on 1 April in line with the contractual timescales. This new system and processes are being used across the council and its partner organisations of Wolverhampton Homes and West Midlands Pension Scheme. This successful delivery has enabled the council to start delivering the agreed year one savings.

The Council is now embarking on delivering phase 2 of the programme in rolling our self-service functionality which will maximise the return on investment as council services are transformed. The Council will continue to manage the

Assistant Director, Health, Wellbeing and Disability March 2015

Strategic Director, Delivery Assistant Director, Finance March 2015

risks around the general governance and structure of this programme and through the general programme controls it will bring.	
Partnership Governance  While work has begun on a number of the Council's key partnerships, a systematic approach to identifying all of the significant partnerships and in determining the level of review of the governance arrangements alongside the 'health' of each partnership, is still being rolled out and will be quite a sizeable task.	Assistant Director, Partnerships, Economy and Culture supported by the Chief Legal Officer March 2015
Information Governance The Council is building on the robust framework and effective working practices it has put in place since consensual audits from the Information Commissioner's Office in 2012 and an enforcement notice in 2014, including:  • Supporting the Information Governance Board through the development of an operational group to drive progress  Mapped out a centralised work programme and resources including a new structure Review of all Information Governance policies Roll out of a suite of training programmes, including ensuring there is 100% compliance with the mandatory 'Protecting Information' training to all employees	Head of Policy March 2015
Strategic Asset Management  The Corporate Landlord model has now been formally adopted by the Council. The implementation of the model and developing a clear understanding of the accountability for activities and financial management will continue.  As part of the Corporate Landlord approach Strategic and Operational Land and Property management is now covered by two governance Boards. The Strategic Land and Property Board is chaired by the Strategic Director Education and Enterprise, and attended by the Corporate Landlord (Strategic Director Delivery). The Operational Land and Property Board is chaired by the Strategic Director Delivery.  The Strategic Land and Property Board meets monthly and the Operational Land and Property Board meets fortnightly. Both Boards consider land and property matters and consult Members through the existing processes of Cabinet Member Briefing; Property Advisory Group and Executive Team prior to decisions being formally made in line with the Council's Constitution.  The Strategic Asset Review has now concluded with the agreed categorisation of assets, and clear accountability	Strategic Director, Education and Enterprise and Assistant Director, Delivery March 2015

established for assets including those considered to be of strategic importance (for Economic Growth and Regeneration including Housing).

Strategic Asset Management will in the future be supported by the Corporate Landlord's Asset Management team acting as a single source of information and knowledge for the Council's assets making sure that the respective governance boards are provided with accurate and relevant information enabling informed decision making and direction.

## **Schools Improvement**

The Council's vision is to create an education system in Wolverhampton that promotes higher standards for all children and young people and closes the attainment gap. This is a system where the Council celebrates school autonomy and supports school leaders and teachers in leading school improvement and having the highest expectations of every child and young person.

The Council's strategic approach to school improvement therefore is to support effective school to school collaboration, strong partnerships and to develop excellent practice in schools whilst demonstrating clear leadership and challenge in the delivery of the its duties in relation to school improvement. This leads to three clear and related targets:

- To ensure that every child in the city has an excellent education.
- To raise standards in schools and academies so that by September 2016 attainment and progress measures at all key stages match or exceed national averages.

To improve the quality of provision in schools and academies so that by September 2016 all schools, settings and academies achieve an Ofsted inspection judgement of good or outstanding.

Assistant Director, Education and Enterprise March 2015

#### **Future Assurance**

Progress reports on the implementation of the above actions from these key improvement areas will be produced by Audit Services and reported to the Audit Committee during 2014/15.

#### Certification

To the best of our knowledge, the governance arrangements, as outlined above have been effectively operating during the year with the exception of those areas identified as requiring improvement. We propose over the coming year to take steps to address the above matters to further enhance our governance arrangements. We are satisfied that these steps will address the need for improvements that were identified during the review of effectiveness and will monitor their implementation and operation as part of our annual review.



Roger Lawrence, Leader of the Council:

Date:

Simon Warren, Chief Executive:

Date:

This page is intentionally left blank

Agenda Item No: 10



# **Audit Committee**

14 July 2014

Report title Corporate Risk Register

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s) Peter Farrow Head of Audit

Tel 01902 554460

Email <u>peter.farrow@wolverhampton.gov.uk</u>

Narinder Phagura Strategic Risk Manager

Tel 01902 554580

Email <u>narinder.phagura@wolverhampton.gov.uk</u>

Report has been considered by

Strategic

**Executive Board** 

26 June 2014

# Recommendations for noting:

The Committee is asked to note:

The latest summary of the Council's corporate risk register.

# This report is PUBLIC [NOT PROTECTIVELY MARKED]

## 1.0 Purpose

1.1 To keep members of the Audit Committee aware of the key risks the Council faces, and how it can gain assurance that these risks are being mitigated.

#### 2.0 Background

- 2.1 The Council is no different to any organisation, and will always face risks in achieving its objectives. Sound risk management can be seen as the clear identification and management of such risks to an acceptable level.
- 2.2 Following a joint SEB/CDB risk identification workshop, the results were fed into the first draft of a new corporate risk register which was presented to the Committee in March 2014. Since this time we have met with the risk owners in order to review the risks and develop individual risk management action plans.
- 2.3 The corporate risk register does not include all of the risks that the Council faces. It represents the most significant risks that could potentially impact on the achievement of the corporate priorities. Other risks are captured within operational, programme, project or partnership risk registers in line with the Council's corporate risk management framework and strategy.
- 2.4 A summary of the corporate risk register is included at appendix A of this report which sets out the status of the corporate risks as at June 2014. These risks are reviewed on an ongoing basis and can be influenced by both external and internal factors and as such, may fluctuate over time.

# 3.0 Progress, options, discussion

3.1 The corporate risk register will be updated as required, and presented at approximately quarterly intervals to the Committee. The Committee will also be given the opportunity to 'call in' individual risks for further review.

#### 4.0 Financial implications

4.1 There are no financial implications associated with the recommendation in this report as Councillors are only requested to note the corporate risk register summary. Financial implications may arise from the implementation of strategies employed to mitigate individual corporate risks, but these will be evaluated and reported separately if required. [GE/30062014/G]

#### 5.0 Legal implications

5.1 Although there may be some legal implications arising from the implementation of the strategies employed to mitigate individual corporate risks, there are no direct legal implications arising from this report. (RB/30062014/B)

#### 6.0 Equalities implications

6.1 Although there may be equalities implications arising from the implementation of the strategies employed to mitigate individual corporate risks, there are no direct equalities implications arising from this report.

# This report is PUBLIC [NOT PROTECTIVELY MARKED]

# 7.0 Environmental implications

7.1 Although there may be some environmental implications arising from the implementation of the strategies employed to mitigate individual corporate risks, there are no direct environmental implications arising from this report.

## 8.0 Human resources implications

8.1 Although there may be some human resource implications arising from the implementation of the strategies employed to mitigate individual corporate risks, there are no direct human resource implications arising from this report.

# 9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations made in this report.

# 10.0 Schedule of background papers

10.1 None





# Appendix A

Profile of current corporate risks

Red 1, 2, 3, 4, 7, 8, 10, 12, 14

Amber 5, 6, 9, 11, 15, 16

# Summary Corporate Risk Register @ June 2014

Corporate priorities

 Encouraging new business  Empowering People and Communities • Re-Invigorating the City

Confident, Capable Council

The following are the current and previous **red** corporate risks the Council faces in delivering its corporate priorities

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
1 Page 281	Looked After Children (LAC)  If the number of LAC is not reduced this may result in an increase in costs, budget overspends and an increased demand on children's services.  Risk owner: Sarah Norman Cabinet Member: Cllr Val Gibson	25 Red	20 Red		15 Red April 2015	The risk has been included in the corporate risk register as a result of the increase in the number of LAC over recent years as follows:  • March 2012- 575  • March 2014 – approx. 750  The current LAC number is 783 which has stabilised over the past three months.  When previously reported in February 2014, the measures to mitigate the risk were noted in a LAC transformation plan. Since this time, this plan has been replaced with the Families First Programme which involves a strategic approach to reducing the numbers of children and young people in Wolverhampton currently looked after by the Council. The programme pulls together a number of projects and work streams under a partnership framework involving:  • Targeted Intervention  • Early Help Support  • A Committed Partnership with all agencies.  The key work streams include:-

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
						Thresholds for services
						Early help assessment
						<ul> <li>Multi-agency service teams co-located with social care teams to provide early intervention.</li> </ul>
						<ul> <li>Edge of Care meetings and Admissions to Care Panels to ensure everything is done to keep children at home.</li> </ul>
						<ul> <li>Review of current LAC Panels to identify where children can be moved from external costly placements to in-house provision or returned home/independence.</li> </ul>
						<ul> <li>Exit from care teams and strategy for support when home/independence</li> </ul>
Page						<ul> <li>Placement sufficiency to provide in-house provision and reduce the cost of external and residential provision</li> </ul>
282						Early indicators are showing a gradual reduction in the number of children coming into care in the past three months and an increase in the number of children and young people leaving care, resulting in the numbers of LAC stabilising and a move in the right direction.
						Progress against the programme is monitored by the LAC budget monitoring group every two months which oversees the timeliness of implementing the agreed actions to manage the risk.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
2 01/14	Skills for Work  If the city residents do not have the appropriate skills that employers require then they will be unable to access the jobs and opportunities available resulting in high rates of unemployment and increased demand on Council services.  Risk owner: Tim Johnson Cabinet Member: Cllr Phil Page	20 Red	20 Red		15 Red March 2015	This risk is being managed through the development of a targeted programme plan that considers a whole system approach. The process involves ascertaining the skills gap that needs to be addressed by identifying future job opportunities, sector needs and the specific skills required for these jobs (as well as generic employment skills) and comparing these against the skills currently available within the City.  It is anticipated that the development of the plan will be completed in July 2014. Delivery of the plan will involve local and regional partnership working with schools, the College, University, the City Board and the Black Country LEP. Further details of the mitigating actions will be reported once the plan has been developed.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
3 01/14 Page 284	Information Governance (IG)  If the Council does not put in place appropriate policies, procedures and technologies to ensure:  • that the handling and protection of its data is undertaken in a secure manner and consistent with the provision of the Data Protection Act 1998;  • compliance with the Freedom of Information Act and Environmental Information Regulations  then it may be subject to regulatory action, financial penalties, reputational damage and the loss of confidential information.  Risk owner: Keith Ireland  Cabinet Member: Cllr Paul Sweet	16 Red	16 Red		12 Amber August 2014	This risk continues to be monitored by the Information Governance Board and is managed through the information governance risk register and the associated maturity model which aims to improve information governance across the Council. The model is based on two levels of maturity. Level one (initial stage from January 2013 to August 2014) is focussed on putting structures in place such as key policies, roles and training. Level two (established stage) from September 2014 to December 2015 will be focussed on embedding best practice across the Council.  In terms of Level 1, significant progress has been made including, the establishment of the Board, the appointment of a Senior Information Risk Officer, roll out of e-learning training for all employees, the introduction of an information protective marking scheme, increased response rates for Freedom of Information requests and implementation of an information incident log.  In relation to information incidences, the Council currently has four "live" incidences which have been reported to the Information Commissioners Office (ICO). An enforcement notice is in place, the outcome of which will be determined by the ICO after June 2014. The risk score will also be reassessed following the ICO's review. The target risk reflects the date by when level one is expected to be achieved.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target	Comment
.0.		(1 05 2011)	(Suite 2011)		date	
4 01/14	Medium Term Financial Strategy If the Council is unable to agree and operate within its medium term financial strategy (MTFS) this may result in insufficient reserves to remain solvent, the potential loss of democratic control and the inability of the Council to deliver essential services and discharge its statutory duties.	15 Red	15 Red		10 Amber March 2015	An updated position of the MTFS was presented to Cabinet on 25 June 2014. This continues to show a forecast budget challenge of £123m before savings for the five year period from 2014/15 to 2018/19.  Although the outturn for the 2013/14 year has not yet been finalised, it is understood that the over spend will be lower than that previously estimated and thereby reducing pressures on future years.  In terms of future years, high level proposals have been identified to address a significant proportion of the deficit
	Risk owner: Simon Warren Cabinet Member: Cllr Andrew Johnson					for 2015/16 and further details of these proposals will be considered prior to the next stage of the budget setting process in October, with the aim of accelerating some of these into the current year.
Page 2						An independent high level review and challenge of the MTFS model was commissioned by the Council in May 2014. This report will be shared with Cabinet and the Audit Committee in due course and provides assurance over the reliability of the model and reasonableness of the savings identified.
285						The risks around the MTFS and the measures in place to manage these risks are monitored through the budget risk register which is also reported to Cabinet on a regular basis.
						The current and target risk assessments reflect the strategy to 2016/17. There continues to remain significant financial challenge, uncertainty and risk from 2017/18 onwards over the future direction of the Council.

			INOT PROTECT			
Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
5 01/14 Page 286	FutureWorks  If the Agresso system does not perform as expected then there will be a prolonged need to maintain the old system resulting in inefficiencies in service delivery; unplanned maintenance costs and delays in planned savings materialising.  Risk owner: Keith Ireland Cabinet Member: Cllr Paul Sweet	Red	10 Amber		5 Amber December 2014	The new Agresso system is aimed at changing and improving processes and procedures for HR, finance, payroll and procurement and went live on 1 April 2014, replacing the 30 year old mainframe. The system transition work has been managed as a programme in two phases, with an agreed programme governance, transition plan and risk management arrangements being in place for phase 1, which ensured system and business readiness by 1 April. The programme has an expected completion date for phase 2 of September 2014 after which the mainframe system is planned to be decommissioned in October.  Since 1 April, the system has been used to run the HR, finance and procurement modules. Some problems have been encountered with the payroll module which has meant that the payroll has continued to be run from the old mainframe. Actions are currently being implemented to ensure all anomalies and queries in the payroll module are successfully resolved to enable the July payroll to be run from the Agresso system.  Transitional problems (including new ways of working and the lack of purchase order numbers on supplier invoices) have also been encountered with payments to suppliers which have resulted in a backlog of payments and an increase in associated complaints. Measures have now been put in place to eliminate the backlog and ensure similar issues do not materialise in the future.  The phase 2 project initiation document and project plan is to be presented to the Programme Board in June for approval, followed by the implementation of the communications plan.  Any ongoing significant risks and issues associated with the programme continue to be reported to the Programme Board which is monitoring the timely implementation of mitigating actions on a daily basis.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
6 01/14 Page 287	Compliance with Public Services Network (PSN)  If the Council fails to achieve the required technology controls required for connection to PSN, and other similar information security regimes, then it will experience significant interruption to the delivery of its services.  Risk owner: Keith Ireland Cabinet Member: Cllr Andrew Johnson	15 <b>Red</b>	5 Amber		5 Amber Achieved	When this risk was first identified in January 2014, the council had not received PSN accreditation for 2013/14 and was working its way through an agreed action plan which enabled accreditation to be granted thereafter with a corresponding reduction in the risk assessment.  This accreditation expired in June 2014 and the Council was required to undergo an annual independent health check prior to it being granted accreditation for the forthcoming year (July 2014 to June 2015). The Council is working its way through this and following the review of an independent health check and assessment of the Council's code of connection submission, it has now also been accredited for a further 12 months until June 2015. The Council continues to work through the action plan which recognises additional system vulnerabilities and will be addressing these over the next few months. Progress against the action plan is monitored by the Information Governance Board and members are also updated on progress through the Member Development and IT Steering Group.  The current assessment reflects the accreditation currently in place.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Target score and date	Comment
7 01/14 Page	Safeguarding If the Council's safeguarding procedures and quality assurance processes are not consistently and effectively implemented then it will fail to safeguard children and vulnerable adults and lead to reputational damage.  Risk owner: Sarah Norman Cabinet Member: Cllr Val Gibson	15 <b>Red</b>	15 Red	10 Amber Next Ofsted inspection	This risk relates to safeguarding processes for both vulnerable adults and children which are overseen by the two local safeguarding boards which share an independent Chair.  The last Ofsted inspection in June 2011 assessed the Council's safeguarding services as "Adequate". Since then peer reviews of both the adults and children's safeguarding services have been carried out in 2013. The findings and recommendations from these reviews have been reported to Ofsted and also incorporated into the safeguarding improvement plan which is monitored by the Strategic Director- Community.  The current risk assessment is largely driven by the assessment in relation to children's safeguarding and despite the planned actions the Council is implementing a further Ofsted inspection is awaited before any further any reduction in risk score is considered.
288					

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
8 01/14 Page 28	Business Continuity Management (BCM)  Failure to develop, exercise and review plans and capabilities that seek to maintain the continuity of critical functions in the event of an emergency that disrupts the delivery of Council services.  Risk owner: Ros Jervis Cabinet Member: Cllr Andrew Johnson	15 Red	15 Red		10 Amber Dec 2014	<ul> <li>This risk is managed and monitored by the Wolverhampton Resilience Board which was established in 2013, and is chaired by the Director of Public Health. The risk assessment reflects the key actions currently being implemented to mitigate the risk which include: <ul> <li>Updating the current business continuity policy to reflect changes to the organisational structure and governance arrangements.</li> <li>Finalisation and approval of the high/ priority critical services by the Board.</li> <li>Development and completion of continuity plans to determine the Council's response strategy.</li> <li>Development of a programme for exercising/ testing continuity arrangements.</li> </ul> </li> <li>Although all of the measures noted above will not be completed until 2015, it is anticipated that significant progress in the completion of the plans will have been achieved by December 2014, at which time the risk will be reassessed.</li> </ul>

Risk	Risk title and description	Initial score	Current score	Movement	Target	Comment
ref	Mak title and description	(Feb 2014)	(June 2014)	MOVEMENT	score and date	Comment
10 01/14	Economic Inclusion  If the Council and its partners do not work effectively together to tackle the root causes of poverty in the City then demand for Council services will continue to increase.  Risk owner: Keren Jones Cabinet Member: Cllr Peter Bilson, Councillor Page and Councillor Mattu	12 Amber	16 Red			The increase in this risk reflects the review and assessment by the risk owner since it was first identified at the workshop. The measures in place to manage the risk include:  • The recently established City Board that has responsibility for overseeing the delivery of both economic growth and economic inclusion in the City Strategy. All main public sector partners are represented on the City Board, together with a representative of the City's business community. There will be a supporting Social and Economic Inclusion Board focusing solely on actions to reduce economic exclusion.
Page 290						<ul> <li>A child poverty strategy which is reviewed on a regular basis to ensure it remains appropriate and valid.</li> <li>A new council to tackling economic inclusion is being developed. This includes grass roots support in deprived wards and joining this up with the excellent work already delivered through the Wolverhampton Adult Education service. Funding is currently being sought to create an intermediate labour market that links the AES to real jobs, work experience and training in some of the Council's cultural venues.</li> </ul>

Risk Risk title and description ref	Revised score (Feb 2014)	Current score (June 2014)	Movement	Target score and date	Comment
14 School Improvement  01/14 If the Council does not provide effective support, challenge and appropriate intervention to raise standards in schools, then the Council and these schools are at risk of underperforming, receiving inadequate Ofsted judgements and a potential loss of control and influence.  Risk owner: Tim Johnson Cabinet Member: Cllr Phil Page	15 Red	15 Red		10 Amber Sept 2014	<ul> <li>The increase in this risk reflects the review and assessment of the risk by the risk owner since it was first identified at the risk workshop. The actions to manage the risk include:</li> <li>A service redesign to focus on schools using a risk based approach and an associated service restructure aligned to service priorities with recruitment to management posts.</li> <li>The implementation of an improvement plan for Key Stage 1 that involves the service adopting a role to challenge schools and obtain assurances on performance.</li> </ul>

[NOT PROTECTIVELY MARKED]

The following are the remaining amber corporate risks the Council faces in delivering its corporate priorities. An update on the progress made in managing these risks will be presented to the Audit Committee in the next report.

Risk ref	Risk title and description	Initial score (Feb 2014)	Current score (May 2014)	Movement	Target score and date
9	City Centre Regeneration	12	12		8
01/14	If the city centre regeneration programme is not effectively managed in terms of project timings, costs and scope, then it will be unable to maximise opportunities including:	Amber	Amber		Amber July 2014
	Risk owner: Tim Johnson				
_	Cabinet Member: Cllr Peter Bilson				
<b>P</b> <b>ay</b> <b>G</b> <b>b</b> <b>b</b> <b>b</b>	The Care Act	12	12		4
<b>G</b> 1/14	If the Council does not have robust plans in place to implement the Care Act including:	Amber	Amber		Amber
293	<ul> <li>appropriate governance arrangements,</li> <li>appropriate project management arrangements</li> <li>sufficient financial resources</li> <li>sufficent workforce capability and capacity</li> <li>effective information systems</li> </ul>		tbc		April 2015
	then it will fail to meet its new responsibilities and discharge its statutory obligations.				
	Risk owner: Tony Ivko Cabinet Member: Cllr Steve Evans				
15	Emergency Planning	9	9		6
01/14	Failure to develop, exercise and review plans and capabilities for preventing, reducing, controlling or mitigating the effects of emergencies in both the response and recovery phases of major a incident.	Amber	Amber		Amber April 2015
	Risk owner: Ros Jervis				
	Cabinet Member: Cllr Andrew Johnson				
					Dogo 42 of 4

Risk ref	Risk title and description	Revised score (Feb 2014)	Current score (May 2014)	Movement	Target score and date
16	Equal Pay	12	12		8
01/14	Significant equal pay liabilities have been dealt with over recent years. However, equal pay will remain a potentially significant risk until:  • The second generation claims, from trade union members, have been dealt with.  • Six years after the implementation of single status, until that time "Abdulla" type claims can	Amber	Amber		Amber April 2015
סד	still be brought.  Risk owner: Keith Ireland  Cabinet Member: Cllr Paul Sweet				

lobe following are the green corporate risks which are reviewed on a regular operational basis in order to ensure that the controls in place to nanage the risk continue to be adequate and effective.

- Health and safety
- Fraud
- Single status
- Treasury Management
- Corporate governance
- Equalities
- Community cohesion

Agenda Item No: 11



## **Audit Committee**

14 July 2014

Report title Annual Internal Audit Report 2013/14

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s) Peter Farrow Head of Audit

Tel 01902 554460

Email <u>peter.farrow@wolverhampton.gov.uk</u>

Report to be/has been

considered by

Not applicable

#### Recommendations for noting:

The Committee is asked to note:

The contents of the Annual Internal Audit Report and the overall opinion that "based on the work undertaken during the year, the implementation by management of the recommendations made and the assurance made available to the Council by other providers as well as directly by Internal Audit, Internal Audit can provide reasonable assurance that the Council has adequate and effective governance, risk management and internal control processes"

#### 1.0 Purpose

1.1 The purpose of this report is to provide the Audit Committee with an annual internal audit opinion on the adequacy and effectiveness of the Council's governance, risk management and internal control processes.

#### 2.0 Background

2.1 This report gives a brief description of the role of Internal Audit, the control environment within which it operates, its compliance with the Public Sector Internal Audit Standards and a summary of the work carried out during the year to 31 March 2014.

#### 3.0 Progress, options, discussion

3.1 Regular progress reports on the work of Internal Audit will continue to be presented to the Audit Committee.

#### 4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/27062014/F)

#### 5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report. [AS/07072014/D]

#### 6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

#### 7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

#### 8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

#### 9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations in this report

#### 10.0 Schedule of background papers – None





### **Audit Services**

### **Annual Internal Audit Report 2013/14**



Section	
1	Introduction
2	Internal audit opinion
3	Compliance with the Public Sector Internal Audit Standards
4	Summary of work completed

#### 1. Introduction

1.1 Our internal audit work for the period from 1 April 2013 to 31 March 2014 was carried out in accordance with the Internal Audit Plan. The Plan was constructed in such a way as to allow us to make a statement on the adequacy and effectiveness of the Council's governance, risk management and control processes.

In this way our annual report provides one element of the evidence that underpins the Annual Governance Statement the Council is required to make to accompany its annual financial statements. This is only one aspect of the assurances available to the Council as to the adequacy of governance, risk management and control processes. Other sources of assurance on which the Council may rely, could include:

- The work of the External Auditors (currently PricewaterhouseCoopers PwC)
- The result of any quality accreditation
- The outcome of any visits by Her Majesty's Revenues and Customs (HMRC)
- Other pieces of consultancy or third party work designed to alert the Council to areas of improvement
- Other external review agencies (i.e. Ofsted, the Information Commissioner's Office)

As stated above, the framework of assurance comprises a variety of sources and not only the authority's internal audit service. However, Internal Audit holds a unique role within a local authority as the only independent source of assurance on all internal controls. Internal Audit is therefore central to this framework of assurance and is required to acquire an understanding not only of the authority's risks and its overall whole control environment but also all sources of assurance. In this way, Internal Audit will be able to indicate whether key controls are adequately designed and effectively operated, regardless of the sources of that assurance.

1.2 The definition of internal audit, as described in the Public Sector Internal Audit Standards, is "Internal Auditing is an independent, objective assurance and consulting activity designed to add value and improve an organisation's operations. It helps an organization accomplish its objectives by bringing a systematic, disciplined approach to evaluate and improve the effectiveness of risk management, control and governance processes".

Internal audit activity is organisationally independent and further details behind the framework within which internal audit operates, can be found in the internal audit charter.

#### **Overall Assurance**

1.3 As the providers of internal audit to the Council, we are required to provide the Chief Executive and Section 151 Officer with an opinion on the adequacy and effectiveness of the Council's governance, risk management and control processes. In giving our opinion it should be noted that assurance can never be absolute. The most that internal audit can provide to the Chief Executive and Section 151 Officer is reasonable assurance that there are no major weaknesses in the Council's governance, risk management and control processes. In assessing the level of assurance to be given,

we have taken into account:

- All audits undertaken for the year ended 31 March 2014.
- Any follow-up action taken in respect of audits from previous periods.
- Any key recommendations not accepted by management and the consequent risks.
- Any limitations which may have been placed on the scope of internal audit.
- The extent to which any resource constraints may impinge on the ability to meet the full audit needs of the Council.

#### 2. Internal audit opinion

- 2.1 We have conducted our audits in accordance with the Public Sector Internal Audit Standards. Within the context of the parameters set out in paragraph 1.3 above, our opinion is as follows:
- 2.2 Based on the work undertaken during the year, the implementation by management of the recommendations made and the assurance made available to the Council by other providers as well as directly by Internal Audit, Internal Audit can provide **reasonable assurance** that the Council has adequate and effective governance, risk management and internal control processes.

However, throughout the year we did note a number of key control issues, either through our work or in the preparation of the Annual Governance Statement, and these are listed below:

While not fundamental to the overall control environment, we gave a 'limited' rating in the following areas:

I54 Payment Processes
Diploma Exemplar Building
Windmill Lane Children's Short Break Respite Centre
Compliance with Birmingham City Council Agency Contract
FutureWorks – Data Migration and Reconciliation
Monitoring of School Surplus/Deficit Balances
Information Governance in Schools
Budgetary Control (Children, Young People and Families)
Waste Management Contract – Financial Monitoring

Significant governance issues arising from the Annual Governance Statement:

The Council recognises that the identification, evaluation and monitoring of risks is a

The Council recognises that the identification, evaluation and monitoring of risks is a key aspect in the governance of the organisation. The following matters represent the most significant current governance issues that are subject to attention in order to ensure that lessons are learnt and good practice is embedded:

FutureSpace
Corporate Landlord
Savings Targets
PSN Compliance
Contract Management and Monitoring
Procurement
Health and Social Care Reforms
FutureWorks
Partnership Governance
Information Governance
Strategic Asset Management
Schools Improvement

Further details on each of these can be found in the 2013/14 Annual Governance Statement.

- 2.3 In reaching our opinion, the following factors were taken into particular consideration:
  - The need for management to plan appropriate and timely action to implement our and other assurance providers recommendations.
  - Key areas of significance, identified as a result of our audit work performed in year are detailed in section 4 of this report.

#### 3. Compliance with the Public Sector Internal Audit Standards



During the year we complied with the new Public Sector Internal Audit Standards which came into effect from 1 April 2013.

#### 4. Summary of work completed

A detailed written report and action plan is prepared and issued for every review. The responsible officer will be asked to respond to the report by completing and returning an

action plan. This response must show what actions have been taken or are planned in relation to each recommendation.

#### Year on year comparison

A total of 78 pieces of audit work were completed during the year, where an audit opinion has been provided. A summary of these audit opinions, with a comparison over previous years is given below.

Opinion	2013/14	2012/13	2011/12
Substantial (was Strong)	18	22	36
Satisfactory	51	42	62
Limited (was Weak)	9	6	7

Where appropriate each report we issue during the year is given an overall opinion based on the following criteria:

 a robust framework of controls which ensures that objectives are likely to be **Substantial** achieved and controls are applied continuously or with only minor lapses a sufficient framework of key controls for objectives to be achieved but the control framework could be stronger or the Satisfactory application of controls could be more consistent · a risk of objectives not being achieved due to the absence of key internal Limited controls or a significant breakdown in the application of controls

### The following internal audit reviews were completed during 2013/14

Key: AAN = Assessment of Assurance Need

Auditable Area	AAN Rating		Recor	nmenda	ntions		Level of Assurance
		Red	Amber	Green	Total	Number accepted	Assurance
Previously reported in Q1, Q2 & Q3 reports							
Fallings Park Primary School	Medium	-	2	13	15	15	Satisfactory
Eastfield Nursery School	Medium	-	4	6	10	10	Satisfactory
Grove Primary School	Medium	-	4	6	10	10	Satisfactory
FutureWorks Programme – High Level Design Phase Review	Medium	-	3	-	3	3	Satisfactory
aregate Primary School	Medium	-	-	2	2	2	Substantial
St Mary's Catholic Primary	Medium	-	5	8	13	13	Satisfactory
Green Park Special School	Medium	-	2	8	10	10	Satisfactory
Duke Street Residential Bungalows	Medium	-	-	11	11	11	Satisfactory
Villiers Primary School	Medium	-	1	2	3	3	Substantial
Merry Hill House	Medium	-	-	9	9	9	Satisfactory
St Alban's CE Primary School	Medium	-	5	7	12	12	Satisfactory
Lanesfield Primary School	Medium	-	3	9	12	12	Satisfactory
Goldthorne Park Primary School	Medium	-	1	7	8	8	Satisfactory
St Stephen's CE Primary School	Medium	-	6	6	12	12	Satisfactory
Bilston Nursery School	Medium	-	1	12	13	13	Satisfactory
St Anthony's Catholic Primary School	Medium	-	1	7	8	7 *	Satisfactory

Auditable Area	AAN Rating		Recor	nmenda	ations		Level of Assurance
		Red	Amber	Green	Total	Number accepted	
Ekta Day Centre	Medium	-	-	7	7	7	Satisfactory
D'Eyncourt Primary School	Medium	-	-	6	6	6	Substantial
Black Country Pathways to Enterprise Project	Medium	-	2	-	2	2	Satisfactory
i54 Payment Processes	High	2	1	1	4	4	Limited
Diploma Exemplar Building	High	2	1	1	4	4	Limited
Play Services	Medium	-	-	9	9	9	Satisfactory
Civic Halls Safe Check	Low	-	1	1	2	2	Satisfactory
Carbon Reduction Commitment (CRC) Annual Assurance Review	High	-	2	2	4	4	Substantial
enior Officers Emoluments	High	-	-	-	-	-	N/A **
Senior Officers Salaries >£50K Check	High	-	-	-	-	-	N/A **
Bayroll Contribution Statements for WMPF	High	-	1	-	1	1	N/A **
CRC Assurance Statement	High	-	-	-	-	-	N/A **
Section 106 Agreements – Monitoring Arrangements	Medium	1	3	-	4	4	Satisfactory
Translation and Interpretation Budget	Medium	-	1	2	3	3	Satisfactory
Civic Centre Car Park Stamp Review	Low	-	-	7	7	7	Satisfactory
FutureWorks Detailed Design Phase	High	-	4	-	4	4	Satisfactory
Corpus Christi Catholic Primary School	Medium	-	-	4	4	2 *	Substantial
Vine Island – Contract Monitoring	Medium	-	1	-	1	1	Satisfactory
Strategic Construction Partnership	High	-	-	-	-	-	Substantial

Auditable Area	AAN Rating		Recor	nmenda	itions		Level of Assurance
		Red	Amber	Green	Total	Number accepted	
Springdale Infant School	Medium	-	-	9	9	9	Satisfactory
Church Street Outreach Services	Medium	-	1	6	7	7	Satisfactory
Windmill Lane Children's Short Break Respite Centre	Medium	-	13	4	17	17	Limited
Ashmore Park Nursery School	Medium	-	1	3	4	4	Substantial
West Park Primary School	Medium	-	-	3	3	3	Substantial
Christ Church CE Infant School	Medium	-	2	3	5	5	Satisfactory
Whitgreave Infant School	Medium	-	5	7	12	12	Satisfactory
at Patrick's Catholic Primary School	Medium	-	-	15	15	15	Satisfactory
Sash Receipting and Banking Arrangements (Regeneration)	Medium	-	1	3	4	4	Substantial
Compliance with Birmingham City Council Agency Contract	Medium	2	3	-	5	5	Limited
Street Lighting – Repair and Maintenance	Low	-	2	-	2	2	Satisfactory
Employment Opportunities	Medium	-	3	14	17	17	Satisfactory
City Show Cash Collection Procedures	Low	-	2	3	5	5	Satisfactory
Bert Williams Leisure Centre Catering Income	Low	-	-	4	4	4	Satisfactory
TR17 Pensions Return Certification	High	-	-	-	-	-	Substantial
Capital Expenditure Managed Audit	High	-	1	-	1	1	Satisfactory
Contaminated Land Programme Grant Certification	Medium	-	-	-	-	-	N/A **
FutureWorks - Data Migration and Reconciliation	High	5	-	-	5	5	Limited
Q4 reports not previously reported							

Auditable Area	AAN Rating		Recor	nmenda	ations		Level of Assurance
		Red	Amber	Green	Total	Number accepted	7100u1u1100
Westacre Infant School	Medium	-	3	3	6	6	Satisfactory
Windsor Nursery	Medium	-	1	3	4	4	Substantial
Colton Hills Secondary School	Medium	-	3	7	10	10	Satisfactory
Ernest Bold Resource Centre	Medium	-	-	13	13	13	Satisfactory
Penn Fields Special School	Medium	-	1	5	6	6	Satisfactory
Trinity CE Primary	Medium	-	1	6	7	7	Satisfactory
Civic Hall Box Office	Medium	-	2	4	6	6	Satisfactory
Monitoring of Surplus / Deficit Balances (Schools)	Medium	-	4	-	4	4	Limited
hformation Governance (Schools)	Medium	NA	NA	NA	NA	NA	Limited***
Electronic Social Care records	Medium	-	-	-	-	-	Substantial
Gamilies in Focus	Medium	-	1	2	3	3	Substantial
All Age Disability Transition Arrangements	Medium	-	4	3	7	7	Satisfactory
Budgetary Control (Children, Young People and Families)	High	-	3	-	3	3	Limited
Recovery of Costs for Third Party Collisions	Medium	-	3	1	4	4	Satisfactory
IT Provision in Schools	Medium	-	1	4	5	5	Satisfactory
Housing Benefits – Local Discretionary Grant Scheme	Medium	-	5	-	5	5	Satisfactory
Review of Procurement Procedures (E&E)	Medium	-	2	2	4	4	Satisfactory
Waste Management Contract – Financial Monitoring	Medium	3	3	-	6	6	Limited
Pensions Gratuities	Medium	-	4	1	5	5	Satisfactory

Auditable Area	AAN Rating		Recoi	mmenda	ntions		Level of Assurance
		Red	Amber	Green	Total	Number accepted	
Complaints Process (Adults and Community)	Medium	-	-	2	2	2	Substantial
Payroll Managed Audit	High	-	-	-	-	-	Satisfactory
Local Taxes Managed Audit	High	-	-	-	-	-	Substantial
Housing Benefit Managed Audit	High	-	-	-	-	-	Substantial
Housing Rents Managed Audit	High	-	-	-	-	-	Substantial
Accounts Receivable and Billing Managed Audit	High	-	-	-	-	-	Substantial
Accounts Payable Managed Audit	High	-	1	1	2	2	Satisfactory
Hixed Assets Managed Audit	High	-	1	6	7	6****	Satisfactory
Annual Pay Award 2013/14	Medium	-	-	-	-	-	Substantial
Agreements  Agreements	High	-	3	2	5	5	Satisfactory
FutureWorks Data Migration and Reconciliation – First Cutover Dress Rehearsal	High	-	4	1	5	5	Satisfactory

#### Notes

- * Relatively minor recommendations not immediately agreed by schools but work on-going to identify acceptable solutions where necessary. This did not impact upon the overall levels of assurance at the schools concerned.
- ** Certification only therefore, no audit opinion required.
- Due to the nature of this audit (covering a sample of schools with varying procedures) recommendations were not formally categorised in terms of being red, amber or green. Instead, a generic action plan was agreed with key officers.
- **** Relatively minor recommendation considered by management to be resource intensive. Alternative solutions to be considered.

#### Key issues arising during the year

Issues that arose during Q1, Q2 and Q3 have already been flagged to the Audit Committee during the year.

#### In our quarter 1 progress report we provided further details on:

FutureWorks Programme High Level Design Phase Review

#### In our quarter 2 progress report we provided further details on:

- i54 Payment Processes
- Diploma Exemplar Building Contract
- FutureWorks Detailed Design Phase Review

#### In our quarter 3 progress report we provided further details on:

- FutureWorks Data Migration and Reconciliation
- Compliance with the Birmingham City Council Agency Contract
- Windmill Lane Children's Short Break Respite Centre

#### In our quarter 4 progress report we are reporting for the first time on:

#### Monitoring of Surplus / Deficit Balances (schools)

An audit of the arrangements for the monitoring of schools' surplus / deficit balances was undertaken, and we raised the following issues:

- Lack of a strategic overview of individual school balances.
- No formal processes in place when challenging schools' financial performance.
- Only limited use made of information provided by schools regarding excessive surplus balances.
- An insufficient challenge made to schools in deficit.

All recommendations were agreed by management for implementation by September 2014 at the latest.

#### Information Governance (Schools)

Our exercise provided a benchmark to demonstrate that the schools who responded to a questionnaire we issued, had a general knowledge and understanding of information governance. However, only limited assurance could be provided as only 50% of the sample responded. It was also apparent that the majority of schools require further guidance in order to ensure they are fully aware of the risks around information governance, as at the moment there is a risk that schools may struggle in the event of a data breach, and the Council may suffer reputational damage as a consequence.

An action plan was agreed as a result of our work in this area and we will review progress with implementation in the coming months.

#### **Budgetary Control (Children, Young People and Families)**

Our audit of the budgetary control procedures operating within areas of Children, Young People and Families identified the following:

- Not always undertaking budget manager responsibilities in accordance with Financial Procedure Rules or making effective use of available budgetary control information.
- Lack of robust budgetary reporting arrangements.
- Limited involvement of budget managers in the budget setting process.

The recommendations were agreed with management for full implementation.

#### **Waste Management Contract – Financial Monitoring**

The Council entered into a partnership agreement for the management of the city's waste in 2006. The service area, has for some time raised a number of issues surrounding the contract and have been taking a series of actions to try and resolve these. The Head of Commercial Services requested our review in order to obtain an independent view on the quality of information that was being provided by the contractor. Our audit supported this view, including:

- The lack of sharing of information by the contractor on an open book basis.
- The accepted rate per tonne historically applied to calculate incinerator costs incurred by the contractor was resulting in an under recovery of Council costs.
- Contrary to partnership agreement conditions, costs associated with the trade waste services were being included as part of the waste management agreement.
- The need to strengthen investment / risk fund partnership agreement clauses, covering the operation and request for information by the Council to the contractor regarding the fund(s)
- The lack of information from the contractor on actual costs means that the Council could not identify whether the contractor has made a trading surplus which should be paid into the investment fund.
- Contradictory clauses within the partnership agreement regarding the provision of the contractor's base profit and overheads.

Negotiations with the contractor are taking place and are being escalated within the respective organisations in order to reach a resolution. However, we will continue to offer our support where we can.

#### **Managed Audits**

Managed Audits are the work we do on the Council's key financial systems and incorporate the requirements of the External Auditors, in order that they can place reliance on our work and thereby reduce their own year-end testing accordingly. The programme of Managed Audit testing undertaken during 2013/14 has been completed. Of the eight Managed Audits undertaken and finalised, we concluded that there was substantial assurance in four and satisfactory assurance in the other four. Two Managed Audits, General Ledger and Budgetary Control, have yet to be finalised, but our level of assurance is likely to be at least satisfactory in both instances.

#### **School Audits**

During the year we maintained a strong audit presence in the City's schools. Our annual school audit review programme focuses upon the adequacy and effectiveness of LA maintained schools' governance, risk management and control processes. Schools completed during the year were assessed as having either satisfactory (19) or substantial (7) levels of assurance. Over the year we found the following recurring issues:

- Schools were failing to obtain declarations of business interests from all staff which is now a requirement within the Scheme for Financing Schools.
- Schools had not established a register of gifts and hospitality which is also now a requirement within the Scheme for Financing Schools.
- Purchase orders continue to be raised after the receipt of goods and services.
- Schools were failing to review Charging and Remissions Policies and annually review and approve all charges levied.

- Governing Bodies were not always approving expenditure in excess of Headteachers' delegated limits and demonstrating compliance with Contracts Procedure Rules for Maintained Schools with Delegated Budgets.
- Schools were not always undertaking checks on vehicle details. Further, there was no
  evidence that staff had the required business use insurance when performing official
  school duties.
- Income arrears were not being monitored by Governing Bodies and schools had not established suitable arrears policies.

#### Savings/reducing potential fraud and error – our contribution

We review the order processing and creditor payments system for potential duplicate payments on an on-going basis. Results of this work during 2013/14 were as follows:

- 50 duplicate payments were identified and stopped prior to payment, with a value of £276,500.
- 31 payments to the value of £68,100 have been or are in the process of being recovered from suppliers.

#### **Benefit Fraud Investigations 2013/14**

The Benefit Fraud Team moved within Audit Services during the year. The table below identifies the value and number of benefit fraud overpayments resulting from investigations during 2013/14. A total of 527 investigations were completed, which identified 177 overpayments. Sanctions resulting from these overpayments were as follows:

Sanction	Value of overpayment £000	Number of cases
Non Sanction Over Payment	265	100
Prosecution	301	53
Administrative Penalty	18	18
Formal Caution	2	6
Total	586	177

#### **National Fraud Initiative**

The table below identifies frauds and errors, as at March 2014, from the Audit Commission's National Fraud Initiative (NFI) data matching exercises. The Housing Benefit figures are also included in the figures reported above.

Description	Number of frauds / errors	Current value (£)
Housing benefit claimants to student loans	11	39,820
Housing benefit claimants to WCC payroll	2	1,450
Housing benefits claimants to WCC pensions	6	18,053
Housing benefit claimants to external payrolls	2	4,724
Housing benefits claimants to external pensions	5	41,654
Housing benefits claims to external housing benefits claims	1	1,770
Housing benefits claims to external housing tenants	1	360
Pension gratuity to DWP deceased records	6	16,005
Overpaid VAT	21	4,474
Duplicate invoice records (different creditors)	2	5,246
Single person discount	139	95,063
*Rising 18's	51	12,879
Total		241,498

Action is being taken to recover the value of the fraud and error wherever possible.

#### Audit and assurance effectiveness measures

Our performance against the following Audit and Assurance effectiveness measures, that were prepared around the successful delivery of the audit service, is as follows:

Audit Plan measures	
Audit reports identifying suggested areas for action, issued to auditees within two weeks of completion of fieldwork.	Approximately 60% of audit reports were issued within two weeks of the completion of audit fieldwork.
Number of audits where time taken to complete assignment is more than 10% longer than planned.	Approximately 40% of reviews took 10% longer than anticipated, with the other 60% completed either on target or under. In the majority of instances, reasons for audit work exceeding budget is that unforeseen issues arise which take time to resolve.
Delivery of at least 80% of the audit plan, and an opinion which provides suitable assurance on the overall governance, risk management and control environment.	The audit plan was subject to significant revision during the course of the year in order to take account of emerging issues and a changing risk profile, during which has been a challenging year for the Council. However, all key risk areas identified in the plan have been completed.

Risk Based Audit Plan produced and	Yes, the Audit Plan was approved before the year
available to the Council in advance of the year to which it relates.	commenced.

Recommendations measures	
90% of recommendations accepted by Council management.	Over 90% of our recommendations made in year were accepted by Council management.
Number of key recommendations followed up, implemented by the Council by the target date.	Approximately 90% of previous key recommendations followed up had been implemented within the agreed date. Where they had not, this was often as a result of a change in circumstance and where appropriate the date was extended and the recommendation re-iterated.

Relationships measure	
•	Yes, the vast majority of feedback was of a positive nature, and is available for review if required.

External Audit measure	
Full reliance placed on internal audit work by External Audit.	Yes, the External Auditors continue to comment favourably on work completed by Internal Audit in support of the Managed Audit arrangement.

#### Quality assurance and improvement programme

Internal audit has a quality assurance and improvement programme. During the year, the internal audit activity has followed this programme and there have been no significant arras of non-conformance or deviations from the standards as set out in the Public Sector Internal Audit Standards.

#### Counter fraud and fraud investigations

We have continued to investigate all allegations of suspected fraudulent activity, and where appropriate whistleblowing allegations, throughout the year. Details of these are monitored through the work of the Audit Committee's Investigations Sub-Committee.

We have also undertaken a wide range of counter fraud initiatives during the year, including a Council newsletter 'The FRAUDitor', Fraud seminars, fraud surgeries, a refreshed website and various self-assessments against recognised counter fraud best practice. Again, further details of these have been reported through the Audit Committee's Investigations Sub-Committee.

#### FutureWorks programme – the assurance framework and our role

We have agreed, and put in place, an internal audit assurance framework for the current FutureWorks programme as shown below:

Project Governance

Risk management

Project Management

Project Management

Implementation review once a preferred supplier has been appointed

Underpinned by Internal Audit representation at Programme Board level throughout

On-going assessment against the recommendations made by Patricia Hughes

A programme of further and on-going reviews will be required at key stages of the project to provide assurance in respect of compliance with this framework.

We continue to work closely with the external auditors (PwC) in relation to this Programme so that our work is co-ordinated and assurance can be provided efficiently. This has included participating in a workshop with PwC to develop an Assurance Framework for the procurement phase of the programme and this has now been completed. As at 31 March 2014, the following combined assurances from internal and external audit had been given against the programme:

Programme area	External and independent assurance obtained to date (the third line of defence)
Reported previously:	
Project Governance	Internal audit position statement – September 2012
The governance and reporting structure and the decision making process.	gave substantial assurance on the project's governance arrangements regarding an appropriate structure being in place to approve key decisions and which addressed the recommendations laid out in the Patricia Hughes report.
Project Management	PwC health check – September 2012 concluded that the programme has robust project and risk management arrangements in place.
The key project management controls in place i.e. project plan with clearly	
defined milestones, resourcing plans, and identification and management of key project risks.	Internal audit position statement – September 2012 gave substantial assurance on the project management arrangements.

#### **Procurement methodology**

The procurement methodology followed and compliance with the prescribed approach.

Internal audit position statement – September 2012 gave substantial assurance on the sign-off arrangements in respect of documentation completed prior to publication of the OJEU notice.

## Tender evaluation and appointment process

The evaluation approach/framework to be adopted for the review and evaluation of tender submissions to ensure there is a transparent process in place in the event of the process being challenged.

Internal audit was able to observe the appropriate conduct of the procurement process up to the appointment of the successful bidder. This included access to dialogue sessions with both shortlisted contractors. Observations and recommendations were fed back to key staff and the board as appropriate.

#### **Risk Management**

How the risks associated with the project are being managed and the role of the Audit Committee in reviewing these.

### In September 2012 the Audit Committee received the following reports:

PwC health check - concluding that the programme has robust project and risk management arrangements in place.

Internal audit update report – the first 'position statement' giving substantial assurance around both the governance and project management framework put in place to oversee the future delivery of the project.

### In July 2013 the Audit Committee received the following report:

PwC external audit update – who stated that they had carried out an audit 'health check' of the programme in March and April 2013. The scope of this review was to consider the Council's programme management controls as at 30 March 2013. Their work specifically focused on the overall programme structure, the Council's assurance framework and controls over the procurement phase of the programme. They reviewed key project documentation and discussed progress with the programme team. They concluded that as at April 2013 the Council had good programme and risk management arrangements in place for the procurement phase of the programme but that more needed to be done to clarify and formalise its future assurance needs for the delivery phase.

#### Implementation Review

How the programme is progressing.

### In September 2013 the Audit Committee received details of the following:

Internal audit high level design review – June 2013 gave satisfactory assurance on the high level design documents for key work streams.

	In December 2013 the Audit Committee received details of the following:
	Internal audit detailed design phase review – November 2013 provided satisfactory assurance on the development of the detailed design documents.
	In March 2014 the Audit Committee received details of the following:
	PwC 'FutureWorks Programme stage gate review – Testing phase' in January 2014 - this concluded that: "The programme has a strong and committed team focussed on delivering the system by 1 April 2014. As with many large programmes of this nature, the team faces a number of challenges most of which have mitigations in place. We have not raised any high priority areas." An agreed action plan has been produced in response to the PwC report and arrangements are in place for internal audit to monitor progress against this, in conjunction with programme management and Agilisys.
	An Internal Audit review focussing on the discreet area of data migration and reconciliation procedures associated with the January data upload identified a number of issues and therefore could provide only limited assurance regarding this particular process.
Reported for the first time:	
	Further to the above, we have recently revisited these issues and noted that while risk levels had reduced from the time of our original report, some concerns remained.

As the project has entered the delivery phase, we have continued to work on data migration and reconciliation procedures relating to the Payroll and HR elements of the system which are yet to fully go live. This work will continue until all the significant issues have been resolved. We have also been supporting the Hub staff in helping to resolve operational issues relating to the implementation of Agresso. We will endeavour to continue to provide this support while backlogs remain, this will impact on our ability to undertake some areas within the existing audit plan. When new systems have settled we will undertake full 'end to end' audit reviews of all of the key systems.

#### Single Status Programme – the assurance framework and our role

Similar to the FutureWorks programme above, we have agreed, and put in place, an internal audit assurance framework for the single status programme as shown below:



Underpinned by Internal Audit representation at Programme Board level throughout

On-going assessment against the recommendations made by Patricia Hughes and the followup of recommendations made in previous audit reports

Availability of internal audit staff at all stages for additional 'deep dive' testing

As at 31 December 2013 we had completed extensive assurance work in respect of payroll implementation which was the final phase of the single status project. Following the implementation of single status, on-going assurance work has taken place in respect of the new transitional arrangements in order to monitor compliance with the collective agreement.

We have been actively involved in the Single Status appeals process and currently have representation on the Equality and Governance panel. Throughout the Single Status process we have been embedded in the programme and have had a presence on the Single Status Board and Pay Strategy Board. The purpose of these groups is to ensure on-going compliance with the collective agreement and to prevent any potential pay inequalities arising from the appeals process.

The Single Status Board was resurrected with effect from 15 January 2014 and meets fortnightly to explore ways of assisting employees with the loss of allowances from 1 April 2014. In addition to the above areas we have also been actively involved in both the pensions autoenrolment and real time information (RTI) projects, which were subsumed under the single status programme. We have been embedded into the project team and have audited the successful implementation of both projects. There are currently no outstanding red or amber issues relating to our work on the Single Status project.

#### **Equal Pay**

We have played an active part in this project and we have previously undertaken assurance reviews around the Council's settlement strategy, the accuracy of proposed settlement offers, and the accuracy of payments. Further assurance work has been undertaken based on the recent outcome of the Birmingham City Council v Abdulla case. We continue to have an ongoing role in terms of providing assurance around future settlements and have presence at equal pay project team meetings.

#### Introduction of New Procurement Procedures

A review was undertaken of the draft Procurement Code in September 2013 which has now been ratified by the Council and implemented. The review identified that the Procurement Code largely addresses compliance with relevant legislation, regulation, EU directives and the Public Service (Social Value) Act, and clearly reflects proposed governance and monitoring arrangements via the Procurement Board. We will continue to monitor the implementation of the Procurement Code and operation of the Procurement Board via Audit representation on the Board. The first meeting of the Board was held on 30 April 2014.

In addition, an audit of the property services procurement processes prior to responsibility for this area of work transferring to corporate procurement has been undertaken, and an update will be provided to the next Audit Committee meeting.

Agenda Item No: 12



### **Audit Committee**

14 July 2014

Report title CIPFA Audit Committee Update – Issue 13

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s)

Peter Farrow Head of Audit

Tel 01902 554460

Email <u>peter.farrow@wolverhampton.gov.uk</u>

Report to be/has been

considered by

Not applicable

#### **Recommendations for noting:**

The Committee is asked to note:

The contents of the latest CIPFA Audit Committee Update, Issue 13 – helping audit committees to be effective.

#### 1.0 Purpose

1.1 CIPFA issue regular briefings for audit committee members in public sector bodies. Their aim is to provide members of audit committees with direct access to relevant and topical information that will support them in their role.

#### 2.0 Background

2.1 The latest edition of these briefings covers, amongst other topics, reviewing the audit plan and an update on the Local Audit and Accountability Act.

#### 3.0 Progress, options, discussion

3.1 Progress against the audit plan, and updates on the Local Audit and Accountability Act will continue to be brought before the Audit Committee.

#### 4.0 Financial implications

4.1 There are no financial implications arising from the recommendations in this report. (GE/23062014/W)

#### 5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report. (AS/20062014/W)

#### 6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

#### 7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

#### 8.0 Human resources implications

8.1 There are no human resources implications arising from the recommendations in this report.

#### 9.0 Corporate landlord implications

9.1 There are no corporate landlord implications arising from the recommendations in this report

#### 10.0 Schedule of background papers

10.1 CIPFA Audit Committee Update – Issue 13

**CIPFA Better Governance Forum** 

# Audit Committee Update -helping audit committees to be effective

#### Issue 13

- Reviewing the Audit Plan
- Update on the Local Audit and Accountability Act
- Briefing on Topical Governance Issues

**March 2014** 

### Introduction

#### **Dear Audit Committee Member,**

Welcome to Issue 13 of our briefings for audit committee members in public sector bodies.

It has been produced by the CIPFA Better Governance Forum and is free to our subscribing organisations. Its aim is to provide members of audit committees with direct access to relevant and topical information that will support them in their role.

This issue includes an article on reviewing the audit plan. This was a topic addressed in our very first issue but with the implementation of the Public Sector Internal Audit Standards (PSIAS) in April 2013 we felt it would be timely to revisit the topic.

We have also included an update on the Local Audit and Accountability Act which has just received Royal Assent. It is likely that further consultation on regulations will be issued in 2014 and audit committees will want to monitor the likely impact on their authority.

I hope you will find this issue helpful. If you have missed earlier issues they are available directly from our website. See the full list overleaf.

We welcome feedback on these briefings and suggestions for future topics. Please let us know if we are getting them right.

Best wishes

Diana Melville Governance Advisor CIPFA Better Governance Forum Diana.Melville@cipfa.org

### Previous Issues of Audit Committee Update

You can download all the previous issues from the CIPFA Better Governance Forum website. The earlier issues are on the archive site. Click on the links below to find what you need.

Issue	Principal Content	Link
1	Reviewing the Audit Plan – Please note that Issue 13 provides an updated review of this topic.	Issue 1
2	Reviewing the Annual Governance Statement	Issue 2
3	Countering Fraud	Issue 3
4	Strategic Risk Management, Governance Risks in 2011, Role of the Head of Internal Audit	Issue 4
5	Understanding the Impact of IFRS on the Accounts, Key Findings from CIPFA's Survey of Audit Committees in Local Government	<u>Issue 5</u>
6	Partnerships from the Audit Committee Perspective	<u>Issue 6</u>
7	Assurance Planning, Risk Outlook for 2012, Government Response to the Future of Local Audit Consultation	Issue 7
8	Commissioning, Procurement and Contracting Risks	Issue 8
9	Reviewing Assurance over Value for Money	Issue 9
10	Public Sector Internal Audit Standards and Updates to Guidance on Annual Governance Statements	Issue 10
11	Local Audit and Accountability Bill, the Implications for Audit Committees	Issue 11
	Update of CIPFA's Guidance on Audit Committees	
12	Reviewing Internal Audit Quality, New CIPFA Publication, Audit Committees Practical Guidance for Local Authorities and Police, Regular Briefing on Current Issues	Issue 12

### Receive our Briefings Directly:

This briefing will be sent to all key contacts of organisations that subscribe to the CIPFA Better Governance Forum with a request that it be forwarded to all audit committee members.

If you have an organisational email address (for example jsmith@mycouncil.gov.uk) then you will also be able to register on our website. This will give you access to governance material, guidance documents and you can receive these briefings directly.

Visit our website <a href="http://www.cipfa.org/Services/Networks/Better-Governance-Forum">http://www.cipfa.org/Services/Networks/Better-Governance-Forum</a> or <a href="register">register</a> today.

# Workshops and Training for Audit Committee Members in 2014

#### **Developing your Audit Committee**

A practical workshop designed to improve the focus and impact of audit committees and extend the knowledge and skills of audit committee members.

11 June Edinburgh, 9 July York, 10 July Birmingham, 4 December London

#### **Advanced Audit Committees**

This workshop examines the audit committee role in strategic risk management, value for money, counter fraud and treasury management.

26 June London, 20 November London

#### **Essential Skills for Board Members**

The role of a board member in a public sector body, featuring sessions on corporate governance, decision making, accountability and evaluating board performance.

8 May Glasgow, 17 June London, 8 July Glasgow

#### **CIPFA Annual Audit Conference**

#### 14 and 15 May, Norton Park Hotel, Winchester

Strategic issues and professional briefings affecting audit in the public services.

#### Keynote session: The UK: Where to from Here?

Political journalist Bill Jamieson will outline the outlook for the UK economy, implications for the public services and the intensifying battle over independence.

#### Other major sessions include:

- The National Audit Office's new role in Local Government and Health Mike Suffield, NAO
- Understanding Cyber Risks Alan Calder, IT Governance
- Internal Audit's Role in Improving the Public Sector's Procurement and Management of Contracts Jay Hussain, Moore Stephens
- The Future of Public Sector External Audit Steve Warren, Jersey Audit Office

## Early booking offers are available until 4 April. Better Governance Forum prepaid places can be used.

View the website for the full programme and booking details

#### In-house training

In house training for any of the events shown above or tailored to your needs is available. For further details contact <a href="mailto:chris.o'neill@cipfa.org">cipfa.org</a>

or email Diana. Melville@cipfa.org

### Reviewing the Internal Audit Plan

#### Introduction

The review of the internal audit plan is a key task for the audit committee. For the audit committee the plan provides the opportunity to review the content of the plan, the risks and priorities that the head of internal audit has identified and also the resourcing. Under the Public Sector Internal Audit Standards (PSIAS)¹ approval of the risk based internal audit plan by the committee is one of the functional reporting requirements that help to safeguard the independence of internal audit.

The aim of this article is to support audit committee members to provide an effective review of the audit plan and also to consider the professional standards requirements of the PSIAS in relation to the plan.

#### Purpose of the Plan

The internal audit plan has a number of purposes:

- It represents the head of internal audit's view of the risks facing the organisation.
- It provides an opportunity for the head of internal audit to discuss and agree priorities for assurance with senior managers and the audit committee.
- The audits included will provide the basis for the annual internal audit opinion.
- It provides the opportunity for a review of the internal audit resources that are necessary to deliver the plan and annual opinion.
- It provides a work plan for the internal audit team.
- It is a prompt for considering how other sources of assurance can be used by internal audit.

The plan is clearly very important to the internal audit team but it is also at the heart of much of what the audit committee will be doing over the coming year. The audits contained in the plan will feature as risk and assurance issues on the audit committee agenda. The annual opinion is a key source of evidence for the annual governance statement. Oversight of the performance of internal audit and conformance with the professional standards are also key parts of the audit committee's terms of reference.

#### **Professional Standards Requirements**

The PSIAS set out a number of requirements for planning and there are also a number of public sector requirements. In summary these are:

- The head of internal audit must establish risk-based plans to determine the priorities of the internal audit activity, consistent with the organisation's goals. (PSIAS 2010)
- The risk-based plan must take into account the requirement to produce an annual internal audit opinion and the assurance framework. (PSIAS 2010 public sector requirement)
- The plan must incorporate or be linked to a high-level statement of how the internal audit service will be delivered and developed in accordance with the internal audit charter and how it links to the organisational objectives and priorities. (PSIAS 2010 public sector requirement)
- The risk-based plan must explain how internal audit's resource requirements have been assessed. (PSIAS 2030 public sector requirement)
- The plan must be based on a documented risk assessment, undertaken at least annually and include input of senior management and the board. (PSIAS 2010.A1)

Public Sector Internal Audit Standards, 2013

#### The Content of the Plan

As a risk-based plan, aligned to the organisation's own objectives and priorities, each one will be different, even for similar organisations. It is likely however that there will be some areas common to many plans. The PSIAS do identify some key areas for assurance so it is likely that these will be identified in the plan. These areas include:

- governance and ethics
- organisational performance management and accountability
- information technology governance
- the effectiveness of risk management
- risk exposures and controls affecting the following:
  - o achievement of strategic goals
  - o the integrity of financial data
  - effectiveness and efficiency
  - o safeguarding of assets and
  - o compliance with laws and regulations
- the potential for the occurrence of fraud and how the organisation manages fraud risk.

The head of internal audit will have discussed risks and priorities with the senior managers to make sure that the final plan is risk based and adding most value to the organisation. It is also likely that there will have been discussion with the external auditor so that account may be taken of the external auditor's scope to place reliance on the work of internal audit. This will be of particular relevance for financial control systems that underpin the financial statements but it may also include other areas of risk that are of concern to the external auditor.

#### What Makes a Good Plan?

In addition to a plan that meets the professional standards, a good plan should achieve a balance between clearly setting out the planned audit work and retaining flexibility to respond to changing risks and priorities during the year.

Traditionally heads of internal audit have produced quite detailed plans identifying all the individual audits planned for the year. This does have the advantage of providing a clear route map to the end of year opinion. However, as the year progresses it is likely that risks and organisational priorities will change, resulting in changes to the plan. This is a particular problem for heads of internal audit in many organisations at the moment because of the pace of change and a high level of uncertainty affecting the risk environment. As a result some heads of internal audit are keeping their audit plans more open, the plan reflecting themes and types of work rather than individual audits. Detailed plans are then produced on a quarterly basis. This can result in more realistic and flexible plans.

A good plan should also include a good balance between operational controls and strategic risks. The audit committee needs to look at what assurance internal audit is able to offer in relation to your organisation's major projects, programmes, new initiatives, partnerships and strategies. In 2010 CIPFA issued its *Statement on the Role of the Head of Internal Audit* which set out the core roles of the head of internal audit. The first of these included: objectively assessing the adequacy of governance and management of existing risks, commenting on responses to emerging risks and proposed developments. The plan provides an excellent opportunity for the head of internal audit to consider emerging risks and provide assurance on new developments.

² CIPFA Statement on the Role of the Head of Internal Audit in Public Service Organisations, CIPFA 2010

#### The Question of Audit Resources

In most public sector organisations the resourcing of internal audit is under pressure.³ As organisations seek to reduce costs many audit teams, whether in-house, bought in or delivered through a shared service, have had to reduce their budgets. Heads of internal audit have a professional responsibility to advise the board if the level of agreed resources will impact adversely on the provision of the annual internal audit opinion ⁴ When reviewing the audit plan the audit committee should bear in mind whether the audit plan will be adequate to support the annual opinion and that the level of resource is adequate to deliver the plan. Resources include the knowledge, skills and experience of the audit team as well as the head count. Pressures may be experienced in areas where extended knowledge or experience is necessary, for example audits of information systems or procurement and contracts. The impact may be that there is reduced coverage of the control environment: either the depth of coverage is reduced or that lesser risks go unaudited.

#### Monitoring the Audit Plan

Once the plan is approved the audit committee will be kept up to date with the audits completed and key issues and recommendations arising. These regular updates are helpful to the committee, keeping them up to date with the robustness of controls, any new concerns and action that is being taken by management. The head of internal audit will also report on changes to the audit plan or any issues interfering with its delivery.

#### How the Audit Committee can Add Value:

- 1. Understand what the professional standards require your internal auditors to do and support the work they do to meet those standards.
- 2. Have an understanding of what assurances your committee needs to fulfil its terms of reference and support the annual governance statement. This will help you identify the assurance that will come from internal audit and what other assurances may be provided.
- 3. Be clear on what the role of the audit committee is in relation to the plan. You may have delegated authority to approve the plan or you may be reviewing the plan for comment and recommendation.
- 4. Respect the head of audit's professional judgement and recommendations. Do ask questions about what has been included or excluded and why, but don't try to re-write the plan at the meeting. It is probably best for the audit committee to feed in suggestions for the plan at an earlier stage.
- 5. Encourage an open discussion on audit resources, respecting both the professional obligations on the head of internal audit and the organisation's wider financial plans.

**Diana Melville Governance Advisor, CIPFA Better Governance Forum** 

³ 59% of heads of audit expected further reductions in their audit team size in CIPFA and Grant Thornton's survey in 2012, <u>The Developing Internal Audit Agenda</u>.

⁴ Public sector requirement, PSIAS 2030,2013

### Local Audit and Accountability Act

#### Introduction

The Local Audit and Accountability Act received Royal Assent on 30 January 2014. The Act concerns the abolition of the Audit Commission and arrangements for establishing a new regime of local audit for local authorities, police bodies, NHS trusts and clinical commissioning groups in England. The Act also includes provisions relating to council tax referendum and the code of practice on local authority publicity. This article will outline the key points that will be of interest to audit committee members:

- Overview of the local audit arrangements.
- Role of auditor panels.
- Impact on the role of the audit committee.
- Future developments.

#### Overview of the New Local Audit Arrangements

The Audit Commission will be closed on 31 March 2015. The monitoring of existing audit contracts that are set to run until 2017 (but may be extended until 2020) will be passed to a transitional body from 1 April 2015.

From 2017 (unless existing audit contracts are extended) organisations will be able to appoint their own external audit provider. There are provisions in the Act about who can be an external audit provider and the regulation of audit providers. Further regulations will be issued on these. These will not be the direct concern of the audit committee so I will not go into further detail here.

All local audit providers will have to work to the Code of Audit Practice which will now be written by the National Audit Office (NAO). The code will set out the scope of the audit and the work and reports that the audit providers will need to provide. The NAO is currently working with the Audit Commission on a plan for taking on the Code of Audit Practice. The current code was approved in March 2010.

The procurement process provides options for different approaches to procurement: individual organisation procurement and joint procurement (for example a regional group of authorities). In addition the Act gives powers to the secretary of state to make regulations concerning collective sector-led procurement. This is one of the late changes to the Act. Smaller authorities (typically town and parish Councils) are likely to use a collective procurement option as it would not be cost effective to operate separate procurements. It is also possible that an offer of a collective procurement service will be offered by the Local Government Association (LGA), or similar, which other authorities can choose to buy into.

#### Role of Auditor Panels

Unless a local authority is using a collective procurement service, it will be required to establish an auditor panel. The role of the panel is to safeguard independence in the appointment of the external auditor. Health and police bodies will use their existing audit committees to fulfil this role as the committees have a majority of independent members. The same option is also open to local authorities, so long as the audit committee fulfils the criteria for independence:

- The committee should be chaired independently of local authority councillors or officers
- It must have a majority of independent members.
- Panel members should meet the requirements of independence, as set out in the Act or other regulations.

The panel can be completely separate from the audit committee or it could include some joint membership. For example it could be a sub-committee of the audit committee which only includes the independent members. It is also possible to set up a joint panel to facilitate a joint procurement exercise. For example, four neighbouring authorities could set up a joint panel, each providing a representative. Again this may or may not involve joint membership of each council's audit committee. Another option would be for the auditor panel of one authority to operate as the panel on behalf of others.

The primary responsibility of the panel (or audit committee if it is acting as the panel) will be to advise the authority on the selection and appointment of a local auditor to audit its accounts. The final choice and power of appointment will not lie with the panel but with the authority, police or health body as appropriate.

Another important role for the panel is to advise the organisation on the maintenance of an independent relationship with their external auditor. Ensuring independence is an important principle in external audit. It is built into the ethics followed by auditors and in the private sector it is monitored and reported on by the audit committee. Usually this role includes policies on non-audit or consultancy work being awarded to the external auditor.

The panel will also have a role in the event of the resignation or removal of the external auditor and this will be set out in regulations from the secretary of state.

CIPFA is planning to issue guidance on auditor panels later in 2015, assuming a significant lead-in to bodies making their first local appointment by December 2016. Discussions are ongoing with Department for Communities and Local Government (CLG) as to the nature of any proposed guidance.

#### Impact on the Audit Committee

A key consideration here is whether the audit committee will be taking on the panel role directly or whether it will be working alongside a panel. If the committee is taking on new roles and responsibilities then it will need to amend its terms of reference accordingly and is likely to require additional training and guidance in the role. If the current make-up of the audit committee does not fit the requirements of the panel then changes will need to be made before the committee can operate as a panel. Forthcoming regulations and guidance will be important for informing these developments.

Where the committee is working alongside the panel then consideration needs to be given to how the two will interact. The existing audit committee will retain its current role of liaison with the external auditor, receiving the results and recommendations from the external audit work, overseeing the relationship between external auditor and the finance team and the internal audit service. The audit committee will have a view therefore of how successfully the audit contract is being fulfilled. This should be communicated to the panel. The panel's guidance on maintaining an independent relationship may need to be monitored by the audit committee.

#### **Future Developments**

The Act will be supported by regulations issued by CLG. These provide more detail on the operation of the Act in practice. The first set of regulations was issued for consultation in November 2013 and further consultation on regulations is expected during 2014. The NAO will also be consulting on the Code of Practice. Further detail will also follow on the monitoring arrangements of existing contracts.

The full text of the Act is available on the Parliament website: <a href="http://services.parliament.uk/bills/2013-14/localauditandaccountability/documents.html">http://services.parliament.uk/bills/2013-14/localauditandaccountability/documents.html</a>

**Diana Melville Governance Advisor, CIPFA Better Governance Forum** 

# Recent Developments You May Need to Know About:

### Legislation and Regulations

#### **Future of Local Audit**

#### Consultation on Secondary Legislation: Summary of Responses

The government has issued a summary of responses it received on its consultation on draft regulations on the conduct of local audit under the Local Audit and Accountability Act. The consultation included questions about smaller authorities, auditor panels and the removal and resignation of auditors, plus others. In response to the comments received, the government has recognised the contribution that audit committees can make, even where they do not act as the auditor panel:

The Government will consider with the sector the development of guidance on how best to utilise the perspective and expertise of audit committees. However, we believe that it is right to have independent oversight in this area, which the audit committee, as a committee of the authority, cannot provide. The Act already allows an audit committee to act as an auditor panel if it has an independent chair and majority for these purposes. An auditor panel may also draw on an audit committee's advice in carrying out its functions, as of course may the authority.

The government has said that further consultation will be issued in May on other draft regulations.

Summary of responses to the consultation

### Reports, Recommendations and Guidance

#### **Audit Committees, Practical Guidance for Local Authorities and Police**

CIPFA's new publication includes its new Position Statement plus extended guidance and resources to support audit committees. The last issue of Audit Committee Update included a summary of the changes and additions from the previous edition.

The new Position Statement is available also as a PDF download from the CIPFA website. This can be shared on public agendas such as audit committee papers.

<u>CIPFA Publications</u> New Position Statement

#### Financial Statements, A Good Practice Guide for Local Authorities

CIPFA has produced a publication to support those producing the financial statements to present them in a way that helps the public understand what is being shown. It is aimed at finance practitioners but may be of interest to audit committee members keen to improve the readability of their accounts.

**CIPFA Publications** 

#### **Local Government Governance Review 2014**

This is Grant Thornton's third annual review of governance in local government. As well as reviewing the effectiveness of annual governance statements and the explanatory foreword this report looks in depth at the issues of risk leadership and governance in partnerships and alternative delivery models.

**Grant Thornton** 

#### **NHS Governance Review 2014**

This third annual review, aims to assist CCG governing bodies and trust boards assess their governance arrangements. It includes coverage of leadership and the governance of commissioning as well financial governance and accountability. **Grant Thornton** 

#### **Cyber Risk Management**

The Institute of Risk Management (IRM) has produced a guidance document on cyber risks, including risks in the supply chain and from social media. The document also highlights key questions for boards to consider about the organisation's approach to managing the risks. The full document is available to download from the Better Governance Forum website. Cyber Risk Management

#### **Protecting the Public Purse and Fraud Briefings**

This report presents the results from the Audit Commission's most recent fraud survey among local authorities in England, which indicates that £178m of fraud was detected by local government in the last year. Just over three quarters of that total was detected by one quarter of councils. The report includes analysis of the main types of fraud risk experienced and examines the trends in detection.

The Audit Commission is also making available individual fraud briefings for all councils in England, via the external auditor. The intention is for the briefings to be presented to the audit committee. The report includes details of what will be provided and questions audit committee members can use to review and understand the data. Protecting the Public Purse

#### **Police Accountability: Landscape Review**

A report from the National Audit Office (NAO) looking at accountability under the new police and crime commissioner arrangements. The NAO finds that gaps in this framework, such as the limited effectiveness of police and crime panels, and HM Inspectorate of Constabulary's lack of authority to carry out routine inspections of commissioners or their offices, could limit the degree of assurance the Home Office can take from the new accountability structure. National Audit Office report

#### **Government Whistleblowing Policies**

The NAO has examined the procedures in place for whistleblowing within government departments against a set of standard criteria. The report shows the breakdown for each department

National Audit Office Report

### Look Out For

#### CIPFA Code of Practice on Managing the Risk of Fraud and Corruption

CIPFA will be issuing a new code for consultation in May. The code will set out the core principles of good counter fraud and corruption practice. It builds on the "Red Book" Managing the Risk of Fraud, Actions to Counter Fraud and Corruption first published in 2006 and updated 2008.

The code will completely update the Red Book principles and provides an opportunity to raise the importance of having good counter fraud practice in all public sector organisations. With the number of significant changes affecting the delivery of counter fraud, particularly in local government, the publication of common standards is an important step forward. The code can be applied to all organisations operating in the public sector, whatever their size.

The code will be supported by additional, sector-specific guidance and an assessment tool. These will help organisations to evaluate their arrangements and plan improvement. The review of an organisation's arrangements against the code will be of interest to audit committees and will provide evidence to support the annual governance statement.

 $\begin{array}{c} \textbf{Better Governance Forum} \ \, \text{www.cipfa.org/Services/Networks/Better-Governance-Forum} \quad 11 \\ \textbf{Page 329} \\ \end{array}$ 

### Sharing this Document

Audit Committee Update is provided to subscribers of the Better Governance Forum for use within their organisation. Please feel free to circulate it widely to your organisation's audit committee members and colleagues. It can also be placed on an intranet. It should not be shared with audit committee members of organisations that do not subscribe to the Better Governance Forum or disseminated more widely without CIPFA's permission.

Audit Committee Update is covered by CIPFA's copyright and so should not be published on the internet without CIPFA's permission. This includes the public agendas of audit committees.

Published by:

#### CIPFA \ THE CHARTERED INSTITUTE OF PUBLIC FINANCE AND ACCOUNTANCY

3 Robert Street, London WC2N 6RL

www.cipfa.org

©2014 CIPFA

No responsibility for loss occasioned to any person acting or refraining from action as a result of any material in this publication can be accepted by the authors or publisher.

While every care has been taken in the preparation of this publication, it may contain errors for which the publisher and authors cannot be held responsible.



Diana Melville Governance Advisor Better Governance Forum **CIPFA** 3 Robert Street London, WC2N 6RL Tel: 01722 349398 Diana.Melville@cipfa.org www.cipfa.org/services/networks/bett er-governance-forum



Certificate No.



Agenda Item No: 13



# **Audit Committee**

14 July 2014

Report title Payment Transparency

Cabinet member with lead responsibility

Councillor Andrew Johnson

Resources

Accountable director Keith Ireland, Delivery

Originating service Audit

Accountable employee(s) Pete

Peter Farrow Head of Audit Tel 01902 554460

Email peter.farrow@wolverhampton.gov.uk

Report to be/has been

considered by

Not applicable

#### Recommendations for noting:

The Committee is asked to note:

The Council's current position with regards to the publication of all its expenditure.

## This report is PUBLIC [NOT PROTECTIVELY MARKED]

#### 1.0 Purpose

1.1 This report is to update the Committee on the Council's current position with regards to the publication of all its expenditure.

#### 2.0 Background

- 2.1 The latest position on the Council's payment transparency activity is as follows:
  - Following the introduction of Agresso, the Council has reviewed how it will
    publish spend data and will no longer use a third party. Instead it will start to
    publish its own data. The new format for the publishing of spend data is
    currently being finalised, with the intention that April, May and June 2014
    data will be available on the Council's internet site by the end of July 2014.
  - Since last reported to the Audit Committee in March 2014, there have been no requests for information from the public (as an 'armchair auditor') via the mailbox available on the Council's website.

#### 3.0 Progress, options, discussion

3.1 We will report back to the next Audit Committee on if the intended date for publishing was achieved, and will continue to provide quarterly updates on the details of any 'armchair auditor' requests the Council receives.

#### 4.0 Financial implications

4.1 The payment publication service was provided by a third party in 2013/14 at a cost of £13,000. This expenditure will no longer be incurred when in-house resources are used to prepare and publish data. [GE/30062014/J]

#### 5.0 Legal implications

5.1 There are no legal implications arising from the recommendations in this report.

#### 6.0 Equalities implications

6.1 There are no equalities implications arising from the recommendations in this report.

#### 7.0 Environmental implications

7.1 There are no environmental implications arising from the recommendations in this report.

# This report is PUBLIC [NOT PROTECTIVELY MARKED]

#### 8.0 Human resources implications

- 8.1 There are no human resources implications arising from the recommendations in this report.
- 9.0 Corporate landlord implications
- 9.1 There are no corporate landlord implications arising from the implications in this report.
- 10.0 Schedule of background papers None

